Holocene fire-climate-vegetation dynamics in Central Canadian Shield boreal forests of Northwestern Ontario, Canada

by Dominic Lafontaine-Senici

A thesis submitted to the Faculty of Graduate and Postdoctoral Affairs in partial fulfillment of the requirements for the degree of

> Doctor of Philosophy In Forest Sciences

Faculty of Natural Resources Management Lakehead University Thunder Bay, Ontario

> © 2014 Dominic Lafontaine-Senici

ProQuest Number: 10611980

All rights reserved

INFORMATION TO ALL USERS The quality of this reproduction is dependent upon the quality of the copy submitted.

In the unlikely event that the author did not send a complete manuscript and there are missing pages, these will be noted. Also, if material had to be removed, a note will indicate the deletion.



ProQuest 10611980

Published by ProQuest LLC (2017). Copyright of the Dissertation is held by the Author.

All rights reserved. This work is protected against unauthorized copying under Title 17, United States Code Microform Edition © ProQuest LLC.

> ProQuest LLC. 789 East Eisenhower Parkway P.O. Box 1346 Ann Arbor, MI 48106 - 1346

Abstract

The boreal forest is the largest terrestrial ecosystem in North America, one of the least disturbed by humans, and most disturbed by fire. This combination makes it an ideal system to explore the environmental controls on fire frequency, the relative importance of climatic and non-climatic factors, and the importance of fire in governing ecosystem processes at a large scale. Multiple vegetation and forest fire history records were established by analyzing Holocene-aged lake sediment stratigraphies collected in the central boreal shield region of northwestern Ontario, Canada. The stratigraphies were dated using ¹⁴C AMS. In the first research chapter macroscopic sedimentary charcoal and plant macroremains were analysed in order to investigate local fire frequency and tree abundance. The results show contrasting fire regime dynamics between a xeric and mesic landscape and highlight the influence of local factors as important drivers of fire frequency at centennial to millennial scales. The second research chapter examines spatial and temporal variations in fire frequency to evaluate the influence of some local non-climatic factors on the regional fire regime. The results show that sites within landscapes fragmented by high open-water lakes and wetland cover tended to have lower fire frequencies, and that this trend was stronger in the past. Results clarify some regional fire regime controls and imply climatevegetation interactions will significantly affect future boreal fire regime dynamics. The third research chapter examines regional fire-vegetation interactions. The results show that millennialscale species community diversity may not be driven by fire frequency variations. This study contributes to our knowledge of the varying spatial controls forest fire regimes in the central boreal forest region.

CONTENTS

Abstracti	i
Tablesv	i
Figures vi	i
Equationsvii	i
Prefaceix	ζ
Acknowledgements	ζ
Acronyms and Definitionsx	i
CHAPTER ONE General Introduction	l
Research context & background	l
Research objectives4	ļ
General philosophy	5
Fire regime	5
Vegetation succession	7
Methodological approach	3
Paleolimnology	3
Age-depth modeling	3
General methods for reconstructing forest fire history)
General methods for reconstructing vegetation history10)
Dissertation organization11	l
CHAPTER TWO Multi-millennial fire frequency and tree abundance differ between xeric and mesic boreal forests in central Canada	2
Abstract/Summary12	2
Introduction13	3
Materials and methods15	5
Study area15	5
Sampling17	7
Charcoal analysis18	3
Macroremains analysis18	3
Age-depth model19)
Reconstruction of fire history)
Reconstruction of vegetation history	
Comparison between sites	2
Fire and vegetation analysis24	ŀ

Results	24
Age-depth model	24
Fire history reconstruction	25
Between-site analysis of fire records	27
Local vegetation history	28
Fire frequency and local tree abundance	29
Discussion	32
Fire history	32
Local tree abundance	34
Conclusions	36
Acknowledgements	37
CHAPTER THREE The effects of forest fuel connectivity and drainage on spatiotemporal dynamics of Holocene fire regimes in the central boreal forest of North America	38
Abstract	38
Introduction	39
Materials and methods	42
Study area	42
Sampling	45
Dating and age-depth models	45
Charcoal analysis and fire history reconstruction	45
Landscape analysis	47
Statistical analysis	48
Results	50
Dating and age-depth models	50
Charcoal analysis and fire history reconstruction	51
Landscape analysis	55
Temporal and spatial effects on fire interval	56
Discussion	58
Conclusion	63
Acknowledgements	63
CHAPTER FOUR Fire-vegetation interactions in mixedwood central boreal shield forests of Northwestern Ontario, Canada	65
Abstract	65
Introduction	66
Methods	68

Study Area	68
Regional modern vegetation	69
Lake sediments	70
Chronologies	70
Charcoal analysis and fire reconstructions	70
Pollen analysis and vegetation reconstruction	72
Rate of change and diversity and richness	73
Fire and vegetation interactions	75
Results	76
Chronologies	76
Fire reconstruction	77
Pollen influx and diversity	79
Vegetation dynamics	79
Fire and vegetation interactions	85
Discussion	
Holocene fire dynamics	88
Holocene fire dynamics Early post-glacial vegetation history	88
Holocene fire dynamics Early post-glacial vegetation history Transition and long-term vegetation dynamics	
Holocene fire dynamics Early post-glacial vegetation history Transition and long-term vegetation dynamics Fire-climate-vegetation interactions	
Holocene fire dynamics Early post-glacial vegetation history Transition and long-term vegetation dynamics Fire-climate-vegetation interactions Conclusions	
Holocene fire dynamics Early post-glacial vegetation history Transition and long-term vegetation dynamics Fire-climate-vegetation interactions Conclusions Acknowledgements	
Holocene fire dynamics Early post-glacial vegetation history Transition and long-term vegetation dynamics Fire-climate-vegetation interactions Conclusions Acknowledgements CHAPTER FIVE Future Work and Conclusion	
Holocene fire dynamics Early post-glacial vegetation history Transition and long-term vegetation dynamics Fire-climate-vegetation interactions Conclusions Acknowledgements CHAPTER FIVE Future Work and Conclusion General Conclusion.	
Holocene fire dynamics Early post-glacial vegetation history Transition and long-term vegetation dynamics Fire-climate-vegetation interactions Conclusions Acknowledgements CHAPTER FIVE Future Work and Conclusion General Conclusion	
Holocene fire dynamics Early post-glacial vegetation history Transition and long-term vegetation dynamics Fire-climate-vegetation interactions. Conclusions Acknowledgements CHAPTER FIVE Future Work and Conclusion General Conclusion Literature Cited Appendices	
 Holocene fire dynamics Early post-glacial vegetation history Transition and long-term vegetation dynamics Fire-climate-vegetation interactions Conclusions Acknowledgements CHAPTER FIVE Future Work and Conclusion General Conclusion Literature Cited Appendices SI 2.1: Figure S1 	
 Holocene fire dynamics Early post-glacial vegetation history Transition and long-term vegetation dynamics Fire-climate-vegetation interactions Conclusions Acknowledgements CHAPTER FIVE Future Work and Conclusion General Conclusion Literature Cited Appendices SI 2.1: Figure S1 SI 2.2: Figure S2 	
 Holocene fire dynamics	
 Holocene fire dynamics	

Tables

Table 2.1 Characteristics of the study lakes	Error! Bookmark not defined.
Table 2.2 Accelerator mass spectrometry ¹⁴ C dating	Error! Bookmark not defined.
Table 2.3 Comparisons of median fire-return intervals	Error! Bookmark not defined.
Table 3.1 Characteristics of the study lakes	Error! Bookmark not defined.
Table 3.2 Median fire intervals (mFIs)	Error! Bookmark not defined.
Table 3.3 Generalized linear mixed models	Error! Bookmark not defined.
Table 4.1 Characteristics of the study lakes	Error! Bookmark not defined.
Table 4.2 Variance (%) and eigenvalues of the two PCAs	Error! Bookmark not defined.
Table 4.3 Median fire return intervals (mFIs)	Error! Bookmark not defined.
Table 4.4 Correlation analyses (Spearman's rho)	Error! Bookmark not defined.

Figures

Figure 2.1 Map of the study area	Error! Bookmark not defined.
Figure 2.2 Age-depth models	Error! Bookmark not defined.
Figure 2.3 Fire events and Fire Return Intervals (FRIs)	Error! Bookmark not defined.
Figure 2.4 Bivariate L-function analyses	Error! Bookmark not defined.
Figure 2.5 Macroremains influx and fire return interval (FI	RI)Error! Bookmark not defined.
Figure 3.1 Map of the study area	Error! Bookmark not defined.
Figure 3.2 Age-depth models	Error! Bookmark not defined.
Figure 3.3 Sediment-charcoal records	Error! Bookmark not defined.
Figure 3.4 Bivariate L-function analyses	Error! Bookmark not defined.
Figure 3.5 Scatterplots and fitted fire intervals (FI)	Error! Bookmark not defined.
Figure 4.1 Map of the study area	Error! Bookmark not defined.
Figure 4.2 Age-depth models	Error! Bookmark not defined.
Figure 4.3 Sedimentary charcoal and pollen records	Error! Bookmark not defined.
Figure 4.4 Simplified diagrams of pollen percentages	Error! Bookmark not defined.
Figure 4.5 Principal Component Analyses (PCA)	Error! Bookmark not defined.

Equations

(3-1)	48
(3-2)	49
(4-1)	74
(4-2)	74
(4-3)	74
(4-4)	75

Preface

This integrated article format thesis dissertation is comprised of a chapter detailing the methodological context of the work and three original research chapters that have been/will be submitted to peer-reviewed scientific journals. Each original research chapter has been edited and modified from the submitted versions for formatting and figure and table numbering.

A version of Chapter 2 was published in the Journal of Ecology.

Senici D, Lucas A, Chen HYH, Bergeron Y, Larouche A, Brossier B, Blarquez O, Ali AA. (2013) Multi-millennial fire frequency and tree abundance differ between xeric and mesic boreal forests in central Canada. Journal of Ecology, 101, 356-367.

Where D.S., H.Y.H.C., Y.B., and A.A.A., conceived the study; D.S. and A.A.A. led the field work; D.S. performed lab work on charcoal; A.Lu. performed lab work on macrofossils; D.S. led the data collection and data analysis; D.S., and A.A.A. led the writing; and all co-authors contributed in reviewing the manuscript.

A version of Chapter 3 was submitted to Journal of Quaternary Science.

Senici D, Chen HYH, Bergeron Y, Ali AA. The effects of forest fuel connectivity and drainage on spatiotemporal dynamics of Holocene fire regimes in the central boreal forest of North America.

Where D.S., H.Y.H.C., Y.B. and A.A.A. conceived the study. D.S. and A.A.A. led the field work; D.S. led the data collection and data analysis; D.S., H.Y.H.C., and A.A.A. led the writing; and all co-authors contributed in reviewing the manuscript.

A version of Chapter 4 is being prepared for submission to Global Change Biology.

Senici D, Chen HYH, Bergeron Y, Ali AA, Lavoie M. Fire-vegetation interactions in mixedwood central boreal shield forests of northwestern Ontario.

Acknowledgements

Detailed acknowledgements are included in each manuscript chapter. This work would not have been possible without the efforts of other researchers and those who helped me in the laboratory. I am indebted to by primary supervisor and co-author of each thesis chapter; Dr. Han Chen, my co-supervisor Dr. Yves Bergeron and to the co-primary investigator of the original project proposal and outline Dr. Adam Ali.

Thanks also to the support and discourse held with my fellow labmates and colleagues; notably, Yong Luo, Brian Brassard, Samuel Bartels, Yu Zhang, and Anthony Taylor. Thanks to the team of researchers who helped extract the sediment sequences: Loic Birker, Benoit Brossier, Etienne DuBerger, and Raynald Juliens. Additional thanks go to Aurelie Genries who taught me the basics of pollen analysis. Lastly I would like to thank Élisabeth Robert and Martin Lavoie at Université Laval for their help and advice with regard to pollen analysis.

I would also like to thank family and friends for their support throughout the years taken to complete these studies.

As with any endeavour; I would like to acknowledge and thank all of those whose work these contributions were borne and built from. And last but not least, I thank the editors and reviewers of the journals where these papers have been submitted to – these efforts have always benefitted myself and produced improved scientific documents.

This dissertation was funded in part by a Natural Sciences and Engineering Research Council of Canada (NSERC) Discovery grant, Ontario Early Researcher Award, and Northern Ontario Heritage Foundation Corporation from H.Y.H.C, NSERC UQAT/UQAM Industrial Chair in sustainable forest management from Y.B. and the CNRS (BOREOFIRE, PALEO2, INSU) by A.A.A. Additional funding was provided by an NSERC Industrial Postgraduate Scholarship, with Forest Ecosystem Science Co-operative Inc, and Resolute Forest Products and an Ontario Graduate Scholarship (OGS) from Lakehead University.

Dominic G. Lafontaine-Senici September 1st, 2014, Thunder Bay, ON

Acronyms and Definitions

a (unit)	annus (year)
AMS	accelerator mass spectrometry
BP	Before Present (CE 1950)
Cal a BP, cal yr BP	calibrated years Before Present
CE	Common Era
CHAR	charcoal accumulation rate
Charcoal	charred particles
CONISS	constrained incremental sums of squares cluster analysis
FI	fire interval
Fire regime	long-term description of fire activity in a region; multiple components.
FRI	fire return interval
LIA	Little Ice Age (1350 - 1850 CE)
LOESS/LOWESS	locally weighted (regression) and scatterplot smoothing
mFRI	median fire return interval
mFI	median fire interval
PCA	Principal Components Analysis
ROC	Rate of Change
SNI	signal-to-noise index

CHAPTER ONE General Introduction

Research context & background

Paleoecology refers to the study of past organisms in relation to their past environment. Paleoecological and paleoenvironmental research use physical, chemical, geological and biological evidence from fossil deposits to investigate the past occurrence, distribution and abundance of different ecological units (species, populations, communities), events (natural disturbances) and environmental states (climate) on a variety of timescales (Birks and John 2008). While it encompasses any time interval in which life existed on Earth, most paleoecologists are concerned with investigating past ecosystem dynamics in the Quaternary period (the last 2.58 million years), with the bulk of research concentrated in the last 20,000 years. The field is multidisciplinary in nature with researchers often combining techniques from paleontology, sedimentology, and geochemistry in order to reconstruct past environmental dynamics and investigate biotic responses to environmental change.

Layers of sediments that accumulate on the bottoms of lakes are an archive of environmental variability that can be used to understand how local terrestrial ecosystems have responded to natural and anthropogenic influences at multi-millennial scales (Last and Smol 2001). The accumulated sedimentary material can be quantified through various physical and chemical processes to reconstruct previous environmental conditions via preserved fossil proxies; charred particles (hereafter charcoal), plant macroremains and pollen can be used to reconstruct high-resolution records of wildfire and vegetation dynamics at local and regional scales (Faegri and Iversen 1989; Whitlock and Larsen 2001). Paleoecological research continues to enjoy tremendous interest and progress in the scientific community as it can be used to identify ecological thresholds for historical "natural" rates of change and can therefore be of use in environmental and socioeconomic planning for conservation, resource development and model building (Birks and John 2008; Brubaker *et al.* 2009).

The North American boreal forest constitutes the largest biome in most Canadian provinces and territories and in Alaska. Boreal regions have a history of repeated glaciation and glacial displacement and the last 10,000 years of post-glacial recovery have generated a forested landscape developed on a mosaic of bedrock outcrops, glacial till, lacustrine deposits and organic soils in poorly drained depressions with the entirety interspersed with lakes, wetlands, fens and bogs. In the boreal forest, fires are a major driver of the carbon cycles and is a significant contributor to ecosystem functioning and biodiversity (Rowe and Scotter 1973; Harden et al. 2000; Bowman et al. 2009). The combination of heterogeneous mosaic of terrain differences and active fire regime has generated a diversity of ecosystem compositions, structures and productivities throughout the boreal biome (Johnson et al. 1998; Weir et al. 2000; Mansuy et al. 2010; Barrett et al. 2013). Examining the natural variability and complex interactions occurring at various spatiotemporal scales of fire activity is crucial in order to understand how future global climate changes will affect fire regimes (Hessl 2011; Girardin et al. 2013b). Empirical studies of natural fire regimes at millennial-scales form our understanding of how the current landscape evolved, how major fire regime drivers and environmental changes interacted, and how boreal forest ecosystems responded to recurrent natural disturbances. In the North American boreal forest, where observational records of historical fire and vegetation are short (< 100 years), fossils preserved in lacustrine sediments potentially contain the most complete postglacial histories of naturally occurring wildfire and vegetation dynamics. Quantitative studies that seek to improve our understanding of large-scale controls of fire regimes are highly relevant

to creating fire policy and sustainable forest management strategies in the context of anticipated future changes in climate and fuels (Gavin *et al.* 2007; Cyr *et al.* 2009; de Groot *et al.* 2013b).

Individual fire events rely upon four variables: an ignition source, available fuels, oxygen and a chemical chain reaction to sustain the burn (Johnson and Gutsell 1994; Halsey et al. 2000). Fossil charcoal indicates that wildfires began soon after the appearance of terrestrial plants in the Silurian Period (420 million years ago (Ma)), but was not widespread until the Early Carboniferous Period (350 - 300 Ma) (McAndrews et al. 1973; Bowman et al. 2009). Fire has continued to play a central role in structuring the natural vegetation of many terrestrial ecosystems not only leading to convergent evolutionary adaptations in widely separated species (Kerby et al. 2007), but influencing vegetation community structure, nutrient cycling, slope stability, soil formation and atmospheric composition. Globally, the frequencies of fire ignitions, fuel abundance and distribution, local geographic conditions and atmospheric oxygen have varied throughout geologic time and have altered fire regimes. Additionally, the relative importance of various fire regime controls have not remained constant through time (Marlon et al. 2008; Power et al. 2008; Krawchuk et al. 2009; Oris et al. 2013) necessitating the construction and interpretation of multiple paleoenvironmental records to assess the natural variability and understand the spatial controls of fire regimes (Heyerdahl et al. 2008; Blarquez et al. 2013).

Complex interactions between the controls of fire operate at multiple temporal and spatial scales (Kashian *et al.* 2006). Broadly, these controls can be characterized as top-down and bottom-up controls. Top-down controls typically refers to climate or biome transformations that can homogenize fire regimes over large regions by overriding the importance of local factors. Climatic variability over short (annual) to long (millennial) time scales drives fire regimes

directly by modifying fuel moisture and indirectly through vegetation assemblages, abundance and density (Flannigan and Van Wagner 1991; Turner and Romme 1994). Bottom-up controls of the fire regime include vegetation types and density, fuel connectivity, surficial deposit drainage, local physiographic features (aspect, slope, relief) and other local site factors (Cyr *et al.* 2007; Mansuy *et al.* 2010; Senici *et al.* 2010; Barrett *et al.* 2013). The relative importance of these controls varies depending on the spatial scale examined and through time in response to climatic and natural ecosystem changes (Gavin *et al.* 2003; Gedalof 2011; Barrett *et al.* 2013).

In the boreal forest of North America there exists temporal limitations on dendrochronological approaches to fire history reconstructions primarily due to the short life span of many boreal tree species and limited historical records of fire activity. In contrast, paleoecological techniques can be used to examine fire regime temporal and spatial variability throughout the Holocene (*c*. 11,700 cal. yrs BP to present) (Whitlock and Larsen 2001; Blarquez *et al.* 2013). When compared to eastern and western North American boreal forests, the central boreal region is only beginning to become explored palynologically to describe fire-climate-vegetation dynamics. Due in part to its intermediate geographical position, the central boreal forest region is potentially different from eastern and western boreal regions because of variability among several factors linked to initiating changes in fire activity including climate (Girardin *et al.* 2006; Ali *et al.* 2012) and physical site characteristics (Cyr *et al.* 2007; Genries *et al.* 2012).

Research objectives

The objective of this dissertation is to improve our understanding of long-term fireclimate-vegetation dynamics in the central boreal forest and determine how fire and vegetation dynamics may have been influenced by climate and local environmental characteristics. To further this goal, this study aims to:

- Develop multiple robust Holocene-scale proxy records of wildfire (charcoal) and vegetation (plant macroremains and pollen) preserved in lake sediments in the central boreal forest.
- Analyse multiple sites across environmental gradients to examine the relative importance and spatiotemporal variability of multiple environmental influences on long-term fire regime by making use of the reconstructed paleoecological records.

General philosophy

Fire regime

Fire has been the dominant natural disturbance regime in the North American boreal forest since the last Ice Age, and is process important for several boreal species and is responsible for shaping landscape diversity and influencing biogeochemical cycles (Stocks *et al.* 2002). The basic theoretical concepts of fire history were laid out by Johnson and Van Wagner (1985) and subsequently refined by Johnson and Gutsell (1994). Fire regimes are a critical foundation for understanding and describing effects of climate on fire patterns and characterizing their combined effects on vegetation and the carbon cycle (Bond-Lamberty *et al.* 2007; Burton *et al.* 2008; Pechony and Shindell 2010).

In general, a fire regime characterizes the spatial and temporal patterns and ecosystem impacts of fire on a landscape. Fire frequency, size, seasonality, intensity, severity, and type are among the most common descriptors of a fire regime (Malanson 1987; Whelan 1995; Weber and Flannigan 1997; Stocks *et al.* 2002). Fire frequency refers to the number of fires of a given size occurring per unit time in a given area. Size, or total area burned, largely determines the forest

mosaic and sizes of different aged patches within a given landscape. Size also influences the dispersal of propagules from the edges of a fire's boundaries thereby indirectly influencing postfire regeneration. The season (timing) of a fire not only influences successional trajectory by moderating vegetative or seed reproductive responses based on the phonological state of the prefire vegetation composition (Weber and Flannigan 1997), but affects fire intensity through seasonal climate effects on fuel moisture. Fire intensity measures the energy output from a fire and is strongly correlated with factors like pre-fire species composition, stand age, topography, substrate, fuel loading, weather, and regional climate. Fire severity describes how fire intensity has affected an ecosystem and is a measure of the magnitude of impact including combined above and below ground biomass losses and mortality (Keeley 2009). Lastly, type describes fire class, whether ground, surface, or crown, based on the fuel layer(s) consumed in the combustion process.

The boreal fire regime is characterized by infrequently occurring, high intensity standreplacing crown fires that occur throughout the principal fire season (April to mid-October) that can incinerate well over a million of hectares of standing vegetation (Rowe and Scotter 1973; Bonan and Shugart 1989; Stocks *et al.* 2001). The large wildfires that account for the majority of area burned are usually caused by lightning (Nash and Johnson 1996; Wotton and Martell 2005; Krawchuk *et al.* 2006), and occur during persistent high-pressure systems that dry forest fuels (Skinner *et al.* 1999; Skinner *et al.* 2002). Smaller fires usually occur under marginal burning conditions, and while numerous, only account for a small percentage of area burned (Stocks *et al.* 2002).

Vegetation succession

In the field of ecology, the study of succession has a long history, being discussed unofficially as early as the nineteenth century. Succession remains a core tenet within ecology, although it is sometimes incorporated into the broader discussion of vegetation dynamics. In this dissertation ecological succession refers to the observed directional non-seasonal cumulative change in the types of plant species that occupy a given area through time.

Two major views of this process were taught by Clements (1916) and Gleason (1926). In Clement's model, vegetation changes from pioneer species through a series of predictable communities which replace each other until a final stable or climax community dominates. This model is said to be linear (always follows the same order) and deterministic (predictable), where the present pattern is governed by the past pattern. Gleason's model suggests that vegetation change following disturbance is a function of the kinds of plants involved and their characteristics relative to the disturbance. Gleason's individualistic concept sees the relationship between coexisting species as the result of similarities in their requirements and tolerances, and partly due to chance. In the modern era, ecological research has focused on understanding and predicting forest succession using more rigorous and data-driven testing of the underlying mechanisms driving community changes (Connell and Slatyer 1977; Heinselman 1981; Tilman 1982; Pickett et al. 1987; Taylor and Chen 2011). Debates continue as to the general predictability of successional dynamics and the relative importance of various mechanisms, consequently understanding plant succession is a on-going key challenge for ecologists and resource managers (Taylor et al. 2009).

Methodological approach

Paleolimnology

Paleolimnology can be broadly defined as the study of the physical, chemical and biological information stored in lake and river sediments (Last and Smol 2001). The overall paleolimnological approach is relatively straightforward. The raw materials used are lake sediments which, under ideal conditions, accumulate at the bottom of a basin in an continuous and undisturbed manner. In a typical study, sediment cores are extracted from the bottom of a lake, after which the sediment profiles are sectioned into appropriate time slices, and the age-depth profile can be established using geochronological techniques (Last and Smol 2001). In this study we collected complete lacustrine sediment sequences at eight different lakes from atop the frozen surface early spring 2010. Within these sediments are various indicators and other proxy data that can be interpreted in a statistically rigorous manner that can useful to other scientists and the general public. Chronologies for six of the eight sampled lakes were established; all six final sites were analysed for macroscopic charcoal and two lakes were analysed for plant macroremains and pollen.

Age-depth modeling

An accurate and precise chronology is an essential pre-requisite for any paleoecological investigation. Chronologies give time scales for events, and hence rates of patterns and processes, and make it possible to compare and correlate events across space in different stratigraphical sequences. The chronologies in this dissertation are based on radiocarbon dating by ¹⁴C accelerated mass spectrometry (AMS) from two laboratories: Poznan Radiocarbon Laboratory, Poznan, Poland (Poz-) and Beta Analytic Inc., Miami, Florida (Beta-). Chronologies of deposits are obtained by drawing a curve through a number of dated depths. Raw radiocarbon

ages are expressed as "¹⁴C BP" or "years BP", where "before present" is defined as 1950 CE (Willard Libby obtained first ¹⁴C dates 1949). Calibrated ¹⁴C ages are given in "cal yr. BP" or "cal a BP" (or variations thereof) dependent on specific journal author guidelines.

General methods for reconstructing forest fire history

Burning, or combustion, of biomass is a complex process that rapidly oxidizes fuel, releasing energy in the form of heat and greenhouse gases. During a fire, organic matter may not be equally exposed to the combustion process, resulting in the production of charred particles that can be quantified and analysed to produce a fire history record. Through a variety of transport mechanisms and taphonomic processes including convective dispersal as well as postfire wind transport, overland flow, and sediment focusing (Clark 1988; Higuera et al. 2007), charcoal particles may enter nearby depositional environments such as lakes or wetlands where they are preserved. Over time, charcoal from subsequent fires accumulates within these environments, eventually forming a long-term record of local disturbance. Sedimentary charcoal records have proven useful for identifying long-term trends in fire activity at local to global scales, studying interactions with vegetation (Blarquez and Carcaillet 2010; Girardin et al. 2013a), and understanding the linkages with climatic changes (Marlon et al. 2008; Ali et al. 2012). Interpreting fire history from sediment charcoal records depends upon understanding the processes controlling charcoal accumulation and the use of analytical methods that appropriately reflect these processes (Higuera et al. 2007; Peters and Higuera 2007). Macroscopic charcoal, usually fragments > 150 μ m in diameter are predominantly derived from local (~ 3 km) fire events (Lynch et al. 2004a; Higuera et al. 2007). Once charcoal has been tallied, the record is typically decomposed into separate time series that describe different aspects of fire history (Clark et al. 1996; Long et al. 1998). Most time series of charcoal accumulation rates (CHAR)

contain a slowly varying component, or background CHAR, and a rapidly changing component called peak CHAR. Where the background component may represents changes in fuel accumulation (Millspaugh *et al.* 2000), burning rates, and local erosion within the basin catchment (Higuera *et al.* 2007), the peak component represents past fire 'episodes' or 'events' (the term 'fire event' is used exclusively throughout the dissertation). This approach to decomposing fire records is facilitated by the CharAnalysis software (Higuera 2009); detailed accounts of the statistical parameters selected and how this software was used to reconstruct fire histories are presented in each chapter.

General methods for reconstructing vegetation history

Past vegetation assemblages can be inferred through the analysis of subfossil plant remains preserved in lake sediments. In this dissertation we used two different proxies, macroremains (Birks and Birks 2000) and pollen (Faegri and Iversen 1989) to reconstruct vegetation history at the local and regional scale respectively. Although each proxy has particular strengths and weaknesses, there are several reasons why macroremains and pollen are valuable as indicators of past conditions. First, they are more easily preserved in sediments due in part to their structural chemistry; under conditions of low oxygen and microbial activity i.e. the bottom of a lake, they are decomposed very slowly. Second, due to their small size they can be carried some distance from their source. While macroremains e.g. fruits, seeds, cones, twigs, etc. are mainly derived from the local flora, pollen grains can be carried some distance from their source, suspended in turbulent air masses, and therefore represent a flora from a wider area of surrounding landscape (Moore *et al.* 1991). Third, pollen grains and spores have unique structure and sculpturing that make them highly recognizable. Often pollen identification can be taken to the species level, though sometimes it is possible to only deduce the genus or the family (McAndrews *et al.* 1973;

Moore *et al.* 1991). Some types of macroremains can be identified with greater taxonomic precision and can often be frequent of taxa that produce very little pollen and are therefore valuable as a fossil material (Birks and Birks 2000). The fourth and final reason is the abundance with which these fossils occur in sediment. Macroremains are produced in small numbers when compared to pollen, and large samples are needed for fossil analysis; in contrast, pollen production is much greater and smaller subsamples can be used in analysis. Complimentary macroremains and pollen analysis of the same sediments can provide local species compositional patterns (macroremains) and regional species abundances and regional vegetation patterns (pollen)

Dissertation organization

The format of this dissertation is as follows: Chapter 1 describes the overall approach and methodological context for the use of lake sediment stratigraphies to examine and interpret fire and vegetation changes throughout the Holocene. Due to differences in acquisition dates of ¹⁴C AMS ages and pollen data related to laboratory access, combined with the impetus to publish original research, original research chapters 2 and 3 reflect the data available when they were written and submitted for publication, and have not been altered to reflect data (pollen vegetation records) obtained at later dates. Chapter 2, examines how local tree abundance responded to fire frequency throughout the Holocene. Chapter 3, examines the influence of local non-climatic factors on local-to-regional fire frequency. Chapter 4, examines regional vegetation compositions and species specific responses to fire. Chapter 5 describes potential future work, some limitations related to this work and the general conclusions.

CHAPTER TWO Multi-millennial fire frequency and tree abundance differ

between xeric and mesic boreal forests in central Canada.

Abstract/Summary

1. Macroscopic sedimentary charcoal and plant macroremains from two lakes (Ben and Small), 50 km apart, in north-western Ontario, Canada, were analysed to investigate fire frequency and tree abundance in the central boreal forest. These records were used to examine the controls over the long-term fire regime, and vegetative dynamics associated with fire return intervals (FRIs).

2. There were 52 fire events at Lake Ben (surrounded by a xeric landscape) between 10 174 calibrated years before present (cal. year BP) and the present with an average FRI of 186 years with individual FRIs oscillating between 40 and 820 years. Forty-three fire events were recorded at Lake Small (surrounded by a mesic landscape) between 9972 cal. year BP and the present with an average FRI of 229 years and a range of 60–660 years. FRIs at Lake Small decreased significantly after c. 4500 cal. year BP, whereas at Lake Ben FRIs remained similar throughout the Holocene. Different FRI distributions and independence in the occurrence of fire events were detected between 10 000 and 4500 cal. year BP for the two sites. Between 4500 cal. year BP and the present, similar FRIs were observed, but fires continued to occur independently.

3. Longer FRIs resulted in declining abundance of *Larix laricina* in both landscapes. Longer FRIs resulted in a decline in the abundance of *Picea mariana* in the xeric landscape, but a marginal increase in the mesic landscape. Abundances of *Pinus banksiana*, *Pinus strobus* and *Betula papyrifera* were unrelated to FRI, underlying that these species maintain their local abundance irrespective of fire frequency.

4. Synthesis. Our results show contrasting fire regime dynamics between a xeric and mesic landscape in central boreal forests, Canada. These results highlight the influence of local factors as important drivers of fire frequency at centennial to millennial scales. Local site factors, especially soil moisture, need to be incorporated into predictive models of vegetation response to climate change.

Key-words: charcoal, climate, fire history, Holocene, macroremains, Neoglaciation, palaeoecology, land-use history, Ripley's K-function, Spearman's ρ, vegetation abundance

Introduction

Fire is a critical driver for many forest ecosystems (Bowman *et al.* 2009). In boreal forests of North America, changes in the frequency and extent of wildfires affect vegetation composition and structure (Bergeron 2000; Bouchard *et al.* 2008; Fauria and Johnson 2008; Chen and Taylor 2012). Fires disrupt the dominant vegetation by consuming large quantities of biomass, thereby providing space and nutrients for colonizing vegetation. Ignition frequency and dry forest fuels are recognised as major contributors in creating the large stand-replacing fires that characterize most boreal fire regimes (Wotton and Martell 2005; Krawchuk *et al.* 2006); under natural settings these factors are primarily controlled by large-scale climatic patterns (Fauria and Johnson 2006; Girardin and Wotton 2009). The area burned by fires in the Canadian boreal forest has increased in the last four decades as temperatures in the summer have increased (Gillett *et al.* 2004) and it is anticipated that further warming and drying throughout the 21st century will increase fire frequency (Wotton *et al.* 2010) and extent (Flannigan *et al.* 2005), with severe environmental consequences (Flannigan *et al.* 2009).

Predicting the responses of boreal fire regimes to future climate scenarios is not straightforward. Fire-climate relationships can be counter-intuitive and can vary depending on location; changes in fire frequency have been linked not only to climate but to several regional and local factors including geomorphological context, forest type and fuel characteristics (Girardin *et al.* 2006; Ali *et al.* 2009a). The relative importance of these biological and physical controls on fire varies across a range of temporal and spatial scales. For instance, Neoglaciation (*c.* 4500 calibrated years before present, hereafter cal. yr BP), a documented climatic trend during the Holocene affecting the onset of cooler and wetter conditions, decreased fire frequency in coniferous boreal forests (Ali *et al.* 2009a) caused by a reduction in the length of the fire season (Hely *et al.* 2010). However, fire frequency in some forests are unchanged or have increased during the late Holocene under colder/wetter climatic conditions (Lynch *et al.* 2004b; Briles *et al.* 2008; Higuera *et al.* 2009; Blarquez *et al.* 2010). Differences in fire regimes among landscapes may also be correlated with relative proportion of area under different fire hazards, i.e. relative amount of xeric and mesic sites (Heinselman 1981; Romme 1982), or forest type (Hely *et al.* 2000; Cumming 2001), with the interaction between fuel characteristics and topography generating different fire frequencies (Hellberg *et al.* 2004; Wallenius *et al.* 2004). In the boreal zone, regional soil characteristics and superficial deposits have also been shown to influence fire regimes (Mansuy *et al.* 2010; Senici *et al.* 2010) and sites with rapid drainage are more prone to fire ignitions (Mansuy *et al.* 2010).

Several paleoecological records (e.g., Lynch *et al.* 2004b; Ali *et al.* 2008) suggest North American boreal forest vegetation compositions have shifted into new arrangements at centennial to millennial time-scales. These shifts are often accompanied by changes in the fire regime, indicating that changes in fire frequency and forest composition coincide over long timescales (Higuera *et al.* 2009; Johnstone *et al.* 2010). However, throughout the Holocene, the relative influences of climate and non-climatic factors on fire frequency and biomass burning are not fully understood (Gavin *et al.* 2006; Marlon *et al.* 2006). Although fire frequencies in the current climate determine the relative abundance of trees (Bergeron 2000; Bouchard *et al.* 2008; Chen and Taylor 2012), the centennial and millennial-scale relationship between tree abundance and fire frequency is unclear.

In this study, we used sedimentary macroscopic charcoal and plant macroremains from two lakes in a boreal forest in central Canada to examine how forest vegetation responded to varying levels of fire frequency throughout the Holocene. The two lakes are on different soil types and surficial deposits, and the amount of waterbodies surrounding each site also vary, all of which could affect local fire ignition and propagation and forest vegetation. Our specific objectives were: (i) to examine the temporal variation of fire frequency, expressed in this study as fire return interval (FRI), and fire synchronicity of the surrounding landscapes of the two lakes; and (ii) determine how local species dynamics might be related to FRI through the Holocene.

We expected that the two lakes would display similar fire histories throughout the Holocene based on the relative primacy of climate *versus* local factors in determining fire regimes (Bessie and Johnson 1995; Fauria and Johnson 2008). We also expected increasing FRIs after the onset of Neoglaciation (*c.* 4500 cal. yr BP) (Viau *et al.* 2006b) due to its inhibition effect of cooler and moister climate conditions on fire ignition and spread. We expected that local woody and herbaceous species abundance would vary depending on FRI; it was expected that macroremains influx of pioneer species would decrease, whereas late-successional species would increase as FRI increases.

Materials and methods

Study area

The two kettle lakes, Lake Ben (49°21'25" N; 89°46'10" W; 480 m above sea level [a.s.l]) and Lake Small (49°34'52" N; 90°23'08" W; 478 m a.s.l), are situated in boreal mixedwood forest in northwestern Ontario, Canada (Fig. 2.1). The lakes (Table 2.1) are located approximately 50 km apart within the Moist Mid-Boreal (MBx) ecoclimatic region (Ecoregions Working Group 1989), which is characterized by a strong continental climate with long cold winters and short warm summers. The mean summer temperature is 14°C and the mean winter temperature is -13°C. Mean annual precipitation ranges between 700-800 mm. Regional soils are generally thin and coarse-textured over granitic Precambrian bedrock (Rowe 1972). Locally, the regional soils surrounding the Lake Ben site are of the dystric brunisol and humo-ferric podzol soil orders on glaciofluvial deposits with rapid drainage (xeric landscape), whereas the Lake Small site is surrounded by humo-ferric podzol on morainal deposits with moderate drainage (mesic landscape). Local waterbodies make up approximately 10% and 27% of the terrestrial area within a 2 km radius from Lake Ben and Lake Small, respectively.



Figure 2.1 Location of the study sites in the boreal mixedwood forest.

Table 2.1The main characteristics of Lake Ben and Lake Small. Species are listed in order of dominance at each site, with *Pinus banksiana* dominating the current local vegetation composition at Lake Ben and *Picea mariana* dominating the local vegetation at Lake Small

Name (unofficial)	Ben	Small
Latitude (N)	49°21'25.02"	49°34'52.19"
Longitude (W)	89°46'10.36"	90°23'08.34"
Elevation (m [a.s.l.])	480	478
Local woody vegetation	Pinus banksiana, Picea mariana, Larix laricina, Betula papyrifera, Abies balsamea,	Picea mariana, Abies balsamea, Pinus banksiana
Hillslopes	Moderate	Moderate
Lake surface (ha)	1.63	0.508
Water depth (m)	5.0	3.9
Core length (cm)	660	390
Sedimentation rate/mean accumulation (cm yr ⁻¹)	0.0866	0.0626
Median time-resolution (yr per sample ⁻¹)	16	26

Fires are frequent, with an average regional fire cycle of 150 (126-188) years since 1820 (Senici *et al.* 2010). Regional post-fire regeneration typically includes pioneer species such as *Pinus banksiana* Lamb., *Populus tremuloides* Michx., *Betula papyrifera* Marsh., *Pinus strobus* L., and *Larix laricina* (Du Roi) K. Koch, with slow-growing shade-tolerant *Picea mariana* (Mill.) Britton, *Picea glauca* (Moench) Voss and *Abies balsamea* (L) Mill. replacing the pioneer species over time (Chen and Popadiouk 2002; Brassard *et al.* 2008). The current vegetation composition and structure at each lake is a result of recent fire histories and local land use, including timber harvesting, over the past 80 years.

Sampling

Sediment cores were extracted from the deepest point in each lake from atop the frozen lake surface early spring 2010. A Kajak-Brinkhurst (KB) gravity corer was used to collect the most recently deposited material at the water-sediment interface and was extruded on site in 1 cm sections. A single complete sequence of sediment was collected in the form of 1-m overlapping cores using a Livingston piston corer (Wright *et al.* 1984). Sediment was wrapped in polyurethane and aluminum foil for preservation and transported to the laboratory. Cores were sliced into disks at contiguous 1-cm intervals.

Charcoal analysis

Subsamples of 1 cm³ were taken from each disk for charcoal analysis. Each subsample was deflocculated in hot 10% KOH solution, bleached in 6% sodium hypochlorite (NaClO) solution, then wet-sieved through a 160 μ m mesh. Deflocculation and bleaching was performed to distinguish charcoal from other biological materials present in the sediment. Charcoal fragments larger than 160 μ m are produced from fire events occurring 0 – 3.0 km from the lakeshores (Lynch *et al.* 2004a; Higuera *et al.* 2007), thus this procedure allowed us to reconstruct fire events at the local scale. Charcoal particles were identified under a × 20 stereo microscope and measured for surface area using a digital camera connected to WinSeedle (Regent Instruments Incorporated, Quebec). Charcoal measurements (cumulative area per cm³) are multiplied by the estimated sedimentation rate based on the numerical age-depth model to obtain the charcoal accumulation rate (CHAR, mm².cm².yr⁻¹).

Macroremains analysis

Quantification of plant macroremains was performed at 2 cm intervals on sediment sequences from each lake. Samples were chemically treated in hot 5% KOH solution to separate and distinguish biological material. The samples were washed with water and wet-sieved through a 250 µm mesh to extract the macroremains from the sediment. Plant remains were identified under a binocular microscope. Identification of macroremains was based on Cappers *et al.*

18

(2006) and a reference collection of boreal species at the Herbarium at Université Montpellier 2 (UM2).

Age-depth model

The chronology is based on radiocarbon dating by ¹⁴C Accelerated Mass Spectrometry (AMS) measurements. Samples were processed by AMS in ¹⁴C laboratories in Poznan, Poland (Poz-) and Miami, Florida (Beta-). The CALIB program (Reimer et al. 2004) was used to calibrate the ¹⁴C ages to dendrochronological years using the IntCal09 (Reimer 2009) calibration curve. Ten and seven ages were obtained for Lake Ben and Lake Small respectively (Table 2.2), based on terrestrial plant macroremains and charcoal fragments. Bulk sediment was used in a single case (Beta-293636; Table 2.2) when plant macroremains and charcoal quantities were insufficient for ¹⁴C dating. The age-depth models were constructed using the MCAgeDepth program (Higuera et al. 2009) in MATLAB (MathWorks 2005) using a weighted cubic smoothing spline. The number of age estimates in each chronology determined the smoothing parameter used for each spline. The probability distribution of each calibrated age was used to weight the influence of each estimate in the age-depth model (Telford *et al.* 2004). Confidence intervals for each age-depth model were derived from 100 Monte Carlo simulations; for each simulation, age estimates used to develop the chronology were randomly selected based on the probability distribution of the calibrated ¹⁴C age. The final chronology represents the median age at each depth from the Monte Carlo simulations.

Site and	¹⁴ C year BP	cal yr. BP (2σ)	Materials	Lab. Code
depth (cm)	2	•		
Lake Ben				
30-35	320 ± 30	387 (309-473)	Plant macroremains	Poz-39543
58-60	605 ± 30	602 (545-653)	Plant macroremains	Poz-39542
120-122	1670 ± 30	1573 (1512-1685)	Plant macroremains	Beta-293602
184-185	2720 ± 30	2814 (2764-2874)	Plant macroremains	Beta-293601
259-260	3780 ± 40	4155 (4008-4302)	Plant macroremains	Beta-293600
359-360	3930 ± 40	4364 (4247-4504)	Plant macroremains	Beta-293599
459-460	4940 ± 40	5663 (5603-5831)	Plant macroremains	Beta-293598
606-610	7310 ± 40	8107 (8025-8190)	Plant macroremains and charcoal	Poz-39541
631-635	8 300 ± 50	9316 (9134-9440)	Plant macroremains and charcoal	Beta-293597
651-661	8 990 ± 50	10174 (9934-10232)	Plant macroremains	Poz-39891
Lake Small				
47-50	1190 ± 30	1115 (1014-1221)	Plant macroremains and charcoal	Beta-293608
76-80	2120 ± 30	2093 (2006-2285)	Plant macroremains and charcoal	Beta-293607
145-150	3490 ± 40	3764 (3652-3865)	Bulk sediment	Beta-293606
216-220	3750 ± 40	4110 (3989-4233)	Plant macroremains and charcoal	Beta-293605
287-288	6410 ± 40	7349 (7349-7419-	Plant macroremains	Beta-293609
356-360	7860 ± 40	8644 (8563-8917)	Plant macroremains	Beta-293604
380-390	8850 ± 40	9972 (9972-10150)	Plant macroremains	Beta-293603

Table 2.2 Accelerator mass spectrometry 14 C dating of Lake Ben and Lake Small (mean ± 1 SEM)

Radiocarbon dates have been calibrated using the CALIB 5.0.1 program (Stuiver and Reimer 1993)

Reconstruction of fire history

CHAR series (Supplementary Information 1 hereafter SI1) were decomposed into background (*Cbackground*), and peak (*Cpeak*) components. A locally defined threshold based on universally applied criteria was used to identify charcoal peaks likely related to the occurrence of one or more local fires (i.e. "fire events"). These methods are described in detail by Higuera *et al.* (2008; 2009) and summarized below. Prior to decomposition, charcoal data were interpolated to

constant 20-yr time steps, corresponding approximately to the average of median resolution-time per sample of the 2 lakes.

Low-frequency variations in a charcoal record (Cbackground) which represent changes in charcoal production, sedimentation, mixing and sampling, were subtracted to obtain a residual series, Cpeak (i.e., Cpeak = Cinterpolated – Cbackground). Consistent with theoretical evidence (Higuera et al., 2007) and previous work (e.g., Gavin et al., 2006; Higuera et al. 2008, 2009), we assume that Cpeak is composed of two subpopulations, Cnoise, representing variability in sediment mixing, sampling and analytical and naturally occurring noise, and Ctire, representing charcoal input from local fires. For each sample, we used a Gaussian mixture model to identify the *Cnoise* distribution. We considered the 99th percentile of the *Cnoise* distribution as a threshold to separate samples into "fire" and "non-fire" events. We estimated *Chackground* with a locally-weighted regression using a 900-yr window at Lake Ben and a 700-yr window at Lake Small. For each record, we chose the window width that maximized a signal-to-noise index (SNI > 3) and the goodness-of-fit between the empirical and modelled Cnoise distributions (KS-test, P-value > 0.05) (Higuera et al., 2009). We did not screen peaks based on the original charcoal counts of each peak as in Higuera et al. (2008; 2009), because this procedure is specific to charcoal count data only (Ali et al2009b). All statistical treatments for charcoal analysis were performed using the program CharAnalysis, written by Philip E. Higuera and freely available at http://charanalysis.googlepages.com.

Reconstruction of vegetation history

The plant macroremains counts (Supplementary Information 2; SI2) were transformed into influx values (#. cm². yr⁻¹) for each species or genus based on the numerical age-depth model at each site. Influx transformations minimize the impact of changes in the sedimentation rate. Knowing that the production of macroremains differs among species, those species that dominated the

reconstructed influxes were rescaled from 0 to 1 by dividing by the maximum influx value in each species. This procedure allowed comparison between species long-term dynamics to be made efficiently. Selected taxa for analysis include: *Betula papyrifera, Pinus banksiana, Pinus strobus, Picea mariana, Larix laricina* and a grouping of herbaceous species that includes *Potentilla* spp., *Cyperaceae* spp., *Carex* spp., *Poa* spp. and *Juncus* spp. For a more comprehensive understanding of the temporal dynamics of each species the influxes were smoothed using a locally weighted regression (LOESS) to highlight the main trends in macroremains dynamics over time.

Comparison between sites

To examine if fire histories differ between the sites we examined the median fire return intervals (mFRI) using the nonparametric two-sample Mann-Whitney test (MW-test) and the overall FRI distributions using the nonparametric two-sample Kolmogorov-Smirnov test (KS-test) (Clark 1989; Ali *et al.* 2009a). The complete series, 10,000 - 0 cal. yr BP (BP = 1950 CE), and defined temporal periods of 10,000 - 4500 cal. yr BP and 4500 - 0 cal. yr BP were examined and analysed. These divisions were defined in regard to the Neoglacial climate cooling trend in central Canada (*c.* 4500 - 1500 cal. yr BP) (Viau *et al.* 2006b) and allowed us to examine if this climatic change had affected the local fire frequency at each site. For the purposes of this study, time periods will hereafter be referred to as complete Holocene (10,000 - 0 cal. yr BP), early Holocene (10,000 - 4500 cal. yr BP) and late Holocene (4500 cal. yr BP to present).

To examine whether the recorded fire events occurred simultaneously at each lake over centennial to millennial time-scales, we assessed their synchrony through the bivariate Ripley's K-function (Ripley 1977) modified to a single dimension (time) (Doss 1989). The modified bivariate K-function allowed us to compare the synchrony of fire events between the two sites

22

within a defined temporal window (\pm 100 years). This analysis was performed for each temporal division, as well as the complete time series. It is important to note that the analysis of Ripley's K function is more robust for shorter sequences, which allow a better characterization of fire event synchrony (Gavin *et al.* 2006); consequently results for the temporal partitioning of the sequence (early and late Holocene) are likely more reliable indicators of fire event synchrony. To facilitate interpretation of the results, the K-function was transformed into an *L*-function (Gavin *et al.* 2006), which stabilizes the means and variances of the K-function outputs. Confidence intervals (95% and 99%) for the *L(t)* values were determined by randomization of 1000 fire events. *L(t)* values > 0 indicate synchrony between fire events, while values near 0 indicate independence and values < 0 indicate asynchrony. The modified bivariate Ripley's K function and its transformation into *L(t)* were performed with the K1D program (Gavin 2010).

Several scenarios may be relevant when interpreting long-term fire history with respect to the relationships between climatic (large-scale) and non-climatic (local-scale) factors as driving forces. These scenarios relate to patterns of FRIs as described by the MW- and KS-test results and fire event synchronicity as described by the modified bivariate Ripley's K function. In *Scenario 1*, similar FRIs and synchronous occurrence of fire events are observed between sites. These results indicate that large-scale processes, *i.e.* climate or biome transformations are the main factors determining fire frequency. In *Scenario 2*, the distributions of FRIs are similar, but fire events is similar between two sites, but local non-climatic factors (landscape connectivity, local weather during fire ignition, topography, and watershed size) override the influence of climate at one or both sites. In *Scenario 3*, the distributions of FRIs are different, but fire events occur synchronously. This scenario *3*, the distributions of FRIs are different, but fire events

fires events at one site with regional fire occurrence strongly influenced by large-scale processes like climate. In *Scenario* 4, FRI distributions are different with asynchronous or independent fire events between sites. This scenario indicates that local non-climatic factors strongly influence the regional fire regime.

Fire and vegetation analysis

To assess the relationship between local vegetation composition and fire, Spearman's rank correlation coefficient (Spearman's ρ), a non-parametric measure of statistical dependence by assessing the monotonic relationship between two variables was calculated. Spearman's ρ was calculated for FRI and macroremains influx values of the dominant taxa at Lake Ben and Lake Small throughout the 10,000 – 0 cal. yr BP time period. Macroremains influx values were interpolated to the time-resolution per sample used in the charcoal analysis (20 years), thereby allowing comparison with the FRI output data.

Results

Age-depth model

The age-depth models represent 10,374 and 9975 years of sedimentation at the Lake Ben and Lake Small sites, respectively. The two age-depth models (Fig. 2.2a,b) are comparable in sedimentation rate with Lake Ben having a higher mean sedimentation rate at 0.0866 ± 0.0534 cm yr⁻¹ to Lake Small's 0.0626 ± 0.0495 cm yr⁻¹ (Table 2.1). Both models exhibit acceleration in the sedimentation rate between *c*. 5000 - 3500 cal. yr BP, with the maximum deposition occurring at 4310 cal. yr BP (0.0291 cm yr⁻¹) in Lake Ben and 3785 cal. yr BP (0.0333 cm yr⁻¹) in Lake Small.


Figure 2.2 Age-depth models, sedimentation rate, and sampling resolution for Lake Ben and Lake Small. Heavy grey lines correspond to 95% CI.

Fire history reconstruction

In total, fifty two (52) fire events were identified at Lake Ben between 10,174 cal. yr BP and the present, with the first fire recorded *c*. 9660 cal. yr BP. The signal-to-noise index (median SNI= 6.09) ensured a clear separation between the charcoal background and fire event signals allowing accurate reconstruction of local fire history. FRIs range from 40 to 820 years, with an average FRI of 186 ± 23 years (mean \pm SE) and a median FRI (mFRI) of 140 years. FRIs were shortest

between 5000 and 4000 cal. yr BP (Fig. 2.3a). FRI length was relatively stable throughout the Holocene with no major changes recorded between time periods (Fig. 2.3a and Table 2.3).

Forty three (43) fire events were identified at Lake Small between 9972 cal. yr BP and the present, with the first fire occurring *c*. 9700 (median SNI = 8.69). FRIs at this site range from 60 to 660 years, with an average FRI of 229 ± 26 years and an mFRI of 170 years. FRIs at the Lake Small site are longer than those at the Lake Ben site throughout the early Holocene (Fig. 2.3a,b). FRIs decrease markedly *c*. 4500 – 4000 cal. yr BP (Fig. 2.3b); the shortest FRIs at the Lake Small site occurred between 4000 – 2000 cal. yr BP. FRIs increased between 2000 – 1000 cal. yr BP but have been decreasing from 1000 cal. yr BP to the present.

Table 2.3 Two-sample MW-test comparisons of median fire-return intervals (mFRI, mean ± 1 SEM) and Two-sample KS-test comparisons of FRI distributions for Lake Ben and Lake Small. The sites display different mFRI (*P*-value < 0.05), and have different FRI distributions only in the 10,000 – 4500 cal. yr BP time period (in bold)

Time period (cal. yr BP)	Lake Ben		Lake Small		KS-test	MW-test
	Average FRI	mFRI	Average FRI	mFRI		
10,000 - 0	186 ± 23	140	229 ± 26	170	0.600	0.095
10,000 - 4500	190 ± 35	120	316 ± 44	260	0.029	0.012
4500 - 0	183 ± 36	140	166 ± 24	120	0.999	0.783



Figure 2.3 Fire events and Fire Return Intervals (FRIs) at (a) Lake Ben and (b) Lake Small. Macroremains influxes smoothed by LOESS (sampling proportion of 0.2) of dominant woody plant and herbaceous species at Lake Ben and Lake Small respectively in the time period 10,000 -0 cal. yr BP. The vertical dotted lines indicate the onset of Neoglacial climate cooling in Central Canada (*c*. 4500 cal. yr BP).

Between-site analysis of fire records

Over the past 10,000 years the mFRIs were similar at Lake Ben and Lake Small (MW-test, P <

0.05) as were the overall FRI distributions (KS-test, P < 0.05) (Table 2.3). The bivariate Ripley's

K-function analysis for the same time period suggests that fire events were predominantly

independent with limited synchrony for a temporal window of 2500 - 2800 years (P < 0.05, Fig.

2.4a). These data indicated that regional forest fire history during the last 10,000 years has been

controlled by a combination of climate and local factors (Scenario 2).

In the early Holocene the median and average FRI distributions between Lake Ben and Lake Small were significantly different (MW-test, KS-test, P < 0.05) (Table 2.3). The analysis of the bivariate Ripley's K-function (Fig. 2.4b) revealed independent fire events at centennial (P < 0.05) and millennial (P < 0.01) time-scales between these sites. The data underlined that localscale factors likely influenced the fire regime during this time period (*Scenario 4*). In the late Holocene, both mFRIs and average FRI distributions were similar between sites (Table 2.3), with fires events occurring independently (Fig. 2.4c). The data indicates that local factors likely overrode climatic controls of the fire regime at one (or both) of the sites in these time periods (*Scenario 2*).



Figure 2.4 Bivariate L-function analyses performed on the fire events detected at Lake Ben and Lake Small. Analysis was performed for 3 time periods (10,000 - 0, 10,000 - 4500, and 4500 - 0). Heavy grey lines correspond to 95% CI and dashed grey lines correspond to 99% CI based on 1000 randomizations of shifting records relative to one another.

Local vegetation history

In total, 18 taxa were identified in the plant macroremains assemblages at each lake (SI2). The most abundant taxa at Lake Ben were *L. laricina* (needles and seeds), *P. banksiana* (needles, brachyblasts and seeds) and *P. mariana* (needles and seeds). The most abundant taxa at Lake Small were *L. laricina* and *P. mariana*. The main tree assemblages included *B. papyrifera*, *P. banksiana*, *P. strobus*, *P. mariana* and *L. laricina*. Identified shrubs include *Rubus* spp. and

Vaccinium spp, however there were too few macroremains for detailed analysis. Herbaceous species are represented by *Potentilla* spp., Cyperaceae spp., *Carex* spp., *Poa* spp. and *Juncus* spp.

At Lake Ben, the oldest period 10,000 – 9000 cal. yr BP was dominated by *P. mariana* transitioning to dominance of *P. banksiana* approx. 8000 – 6500 cal. yr BP (Fig. 2.3a, SI2). The period 6000 – 4000 cal. yr BP was characterized by the maximum influx of macroremains and throughout concomitantly the presence of *P. banksiana*, *P. strobus*, *P. mariana*, *L. laricina* and *B. papyrifera* was recorded. From 2300 cal. yr BP to the present, Lake Ben was characterized by a low influx of all macroremains taxa except *L. laricina*.

Lake Small had relatively low macroremains influx for all taxa in the early Holocene (Fig. 2.3b). *L. laricina* dominates the record between 10,000 – 9000. Influx of woody species increased considerably at *c*. 4000 cal. yr BP; this trend was not recorded in the herbaceous grouping. The rapid and simultaneous increase in woody macroremains coincided with acceleration in the sedimentation rate recorded *c*. 4000, suggesting a potential artefact; however, the use of influx transformations to analyse macroremains data had likely minimized the impact of changes in the sedimentation rate. Around 3200 cal. yr BP, influxes decreased slightly for all taxa.

Fire frequency and local tree abundance

Significant correlations were detected between macroremains influxes of both *P. mariana* and *L. laricina* and FRI at each site. Significant correlations were of low to moderate ($\rho < \pm 0.4$) strength (Fig. 2.5). At Lake Ben, *P. mariana* (Fig. 2.5g) was positively ($\rho = 0.1$) correlated with FRI, indicating that this species increased with FRI. At Lake Small, a negative correlation ($\rho = -0.1$) for *P. mariana* (Fig. 2.5h) indicates that the species was most abundant at this site when

FRIs were short (100 - 150 years), but a high occurrence of its macroremains was also recorded at long FRIs (> 250 years). *L. laricina* influx was highest at Lake Ben when the FRIs were short (Fig. 2.5i). At Lake Small, a strong negative correlation ($\rho = -0.4$) was found for *L. laricina* (Fig. 2.5j), and similar to the results for *P. mariana* at this site, this species was most abundant at short FRIs, however a high occurrence of the macroremains was also recorded when FRIs exceeded 250 years. Herbaceous species were found to be significantly correlated with FRI at Lake Ben (Fig. 2.5k) but not at Lake Small (Fig. 2.5l). Herbaceous species at Lake Ben showed a moderate negative correlation; the herbaceous taxa macroremains were most abundant when FRIs were short. At Lake Ben and Lake Small no significant correlations were detected for *B. papyrifera* (Fig. 2.5a,b) and *P. banksiana* (Fig. 2.5c,d), indicating their relative persistence at both sites irrespective of FRI. We note that *P. strobus* (Fig.5e,f), a relatively minor component of the vegetation at each site displays maximum influx when FRIs were near the complete Holocene average at each lake (Lake Small 229 ± 26 years; Lake Ben 186 ± 23 years).



Figure 2.5 The relationship between macroremains influx and fire return interval (FRI) for Lake Ben and Lake Small. Spearman's ρ and *P*-values were calculated for dominant woody species and a grouping of herbaceous plants.

Discussion

Fire history

The general pattern of fire activities at the millennial-scale is counter-intuitive. The climate of the North American boreal forest changed significantly between 6000 – 3000 cal. yr BP with conditions becoming cooler and wetter in comparison to the early Holocene (Kerwin et al. 2004; Viau et al. 2006b: Viau and Gajewski 2009). The establishment of cooler and moister environmental conditions in the late Holocene should not have been favourable for fire ignition, yet the expression of this climactic shift on fire frequency varies between sites, with FRIs at Lake Ben (Fig. 2.3a) remaining similar to those of the early Holocene and FRIs at Lake Small (Fig. 2.3b) decreasing significantly. Data from both sites are different from the increased FRIs observed in coniferous boreal forests during the late Holocene (Ali et al. 2009a), likely due to differences in fire regime seasonality between boreal coniferous and mixedwood forests. In boreal mixedwood forests, severe fires occur mainly in the spring (April – June) (Johnson et al. 1999) prior to the growth deciduous foliage (mainly *B. papyrifera*, and *P. tremuloides*). Since decreased fire frequency in the coniferous boreal is associated with a shortened fire season mainly in summer-fall (June - October) under orbital forcing changes (Helv et al. 2010), the influence of climatic cooling on fire frequency may be limited at our study sites which are predominantly mixedwood stands. The impact of Neoglacial cooling on fire frequency was apparently limited at Lake Ben.

The difference in local fire frequency between the landscapes surrounding the two lakes can be a result of multiple mechanisms. The Lake Ben site is situated in a landscape with more xeric conditions when compared to Lake Small, as such the area was more prone to fire ignition and propagation as evidenced by the shorter average FRI 10,000 – 0 cal. yr BP. The shorter

average FRI at the Lake Ben site in the early Holocene as compared to the Lake Small site might also be explained by the presence of waterbodies in the local landscapes (Cyr *et al.* 2005; Senici *et al.* 2010). There exist abundant lakes that encircle Lake Small (Fig. 2.1c); these lakes may have acted as natural firebreaks, inhibiting fire spread and preventing some regional fires from reaching sizes or dispersal distances that would be recorded in Lake Small's charcoal paleorecord. At the Lake Ben site, surrounding waterbodies are few (Fig. 2.1b) allowing local fires to spread relatively unhindered.

The similar FRIs among sites in the late Holocene could due to be increased waterlogging at the Lake Small site (Lynch *et al.* 2004b), induced by a slower rate of decomposition in organic matter due to increased moisture and a cooler climate. Increased mesic conditions at the Lake Small landscape may have favoured the development of *P. mariana* stands as suggested eastward in other sites (Carcaillet *et al.* 2010; Genries *et al.* 2012); the development of *P. mariana* have been shown to increase boreal landscape flammability (Hallett and Walker 2000; Lynch *et al.* 2002). In contrast, the Lake Ben site is surrounded by moderate hillslopes on welldrained sandy soils; as such it is possible that increased moisture had little effect on soil conditions and *P. mariana* abundance resulting in the stable FRIs at Lake Ben throughout the Holocene.

Different FRIs and independence in fire occurrence were detected in the early Holocene, showing that during this time period forest fires at each site were primarily controlled by local non-climatic factors (*Scenario 4*). However, this statement must be considered with caution because during this period the two sites display significant differences in time-resolution per sample (Fig. 2.2), even if this impairment was minimized by numerical treatments performed for fire detections. In the late Holocene, FRIs were similar but independence in fire events were

recorded, stressing a combined effect of climate and local factors on forest fire activity. During the last 4500 years, the long term probability of fire is similar at both sites but the effect of climate on fire was overridden by local controls at one or both sites (*Scenario 2*). This period displayed more important climatic controls of fire history over time at the regional scale, even if local factors continued to play a key role in local fire ignition and propagation.

Local tree abundance

Since FRI is used as a proxy for fire frequency, our interpretations are based on whether monotonic relationships between species abundance and fire frequency exist. Although we cannot clearly specify the cause and effect (a limitation of Spearman's ρ) in the observed significant monotonic relationships, the suggested mechanisms by which they might occur are based on differences in the local conditions at each site and the life-history traits and fire adaptations of each species.

At both sites *L. laricina* is most abundant at short-to-moderate FRIs, consistent with the species status as an early invader that is well adapted for post-fire regeneration (Rowe and Scotter 1973; Johnston 1990). The early development of *L. laricina* at Lake Small at the beginning of the Holocene, i.e. *c.* 10,000 cal. yr BP under low fire frequency, could be related to high soil moistures with low drainage processes after the disappearance of the glacial lake Agassiz. *L. laricina* abundance in the vegetation record (Fig. 2.3, SI2) and significant correlations with FRI might also be partially explained by edge effects. At the Lake Ben site increased moisture by lakeshore provides mesic conditions. Since *L. laricina* grows most commonly on wet to moist organic soils such as peat and woody peat (Johnston 1990), these local microclimate conditions may have allowed the species to exist at a location where it also maximally contributed macroremains to the sediment. Similarly, *L. laricina* is shade-intolerant,

at the Lake Small site the increased abundance of *L. laricina* macroremains when FRIs are > 250 years might be explained by increased light availability at the lakeshore permitting species regeneration despite a closed forest canopy.

At the Lake Ben site, *P. mariana* is the only species to have a significant positive correlation with FRI, with peak abundance occurring when FRIs are between 145 - 195 years. At the Lake Small site, *P. mariana* displays a weak negative correlation with FRI. At this site *P. mariana* abundance is greatest when FRIs are short. The difference in the direction of the correlation between sites for *P. mariana* is likely due to local soil characteristics. The Lake Ben site is more xeric, as such it is probable that post-fire regeneration favoured the growth of colonizers that are better adapted for dry soils with rapid drainage (e.g. *P. banksiana*), with *P. mariana* succeeding them over time. The increase in *P. mariana* abundance at the Lake Ben site with FRI is consistent with the known ecology of the species when fire frequency is low in xeric environments due to their high tolerance to shade and ability to regenerate by seeding or vegetative layering in the understory (Taylor and Chen 2011). By contrast, at the Lake Small site, post-fire regeneration in mesic conditions would favour those species that are best adapted to moist organic soil conditions like *P. mariana* and *L. laricina*, with populations declining as FRI increases.

B. papyrifera, *P. banksiana*, and *P. strobus*, the most fire-adapted species of boreal forests displayed no significant correlation with FRI, contrasting our expectation that influx of these species would be highest at high fire frequencies (short to intermediate FRIs). These results suggest that factors other than fire may have controlled the local abundance of these species or that these species maintained their presence at each site irrespective of fire. That *P. banksiana* was able to maintain a near continuous presence at both sites is counter-intuitive because the

species is shade-intolerant and is near reliant on fire for regeneration (Rudolph and Laidly 1990). However, a recent study of boreal forest successional pathways show that *P. banksiana* can maintain dominance on xeric and mesic sites well into 300 years post-fire in the boreal forest (Taylor and Chen 2011). Similarly, *B. papyrifera* is well-adapted to post-fire regeneration (Brassard and Chen 2006), however fire itself is not essential for its successful reproduction and it is likely that the species was able to recruit new individuals in the gaps created by less severe fires and other disturbances (insect defoliation and windthrow) (Bergeron 2000; Chen and Taylor 2012). The presence of *P. strobus* independent of FRI is attributable to the fact that it is longlived, able to regenerate under its own canopy, and able to survive moderate-severity fires due to its thick bark at mature stages (Wendel and Smith 1990).

Herbaceous species were found to be more abundant at Lake Ben when FRI was short. The correlation is consistent with the understanding of post-fire regeneration in the boreal region where herbaceous plants are quick to establish themselves when light is abundant and competitors are absent (Hart and Chen 2006), declining over time as competitive woody species and mosses establish themselves. The absence of a similar trend in Lake Small is likely due to the relative dearth of herbaceous macroremains preserved at this site (SI2).

Conclusions

In central boreal forests local landscape factors such as soil moisture could overcome the influence of climate on fire frequency. Fire-adapted species like *Pinus banksiana*, *Pinus strobus* and *Betula papyrifera* can maintain their presence on xeric and mesic sites irrespective of local fire frequency. The abundance of pioneer *Larix laricina* generally increases with fire frequency, but the effect of fire frequency on the abundance of slow-growing shade-tolerant *Picea mariana* is landscape-dependent. These results underline the complexity of fire-climate-vegetation

relationships in the boreal forest and emphasise that climate alone cannot explain changes in boreal forest composition over millennia. This corresponds to a key issue in improving the forecasting of fire-induced vegetation dynamics in response to ongoing climate change. The continuous challenge is to differentiate between the influences of climatic and non-climatic factors on fire frequency, and consequences on vegetation composition. Further paleoecological investigations are required to better understand the ecological processes involved in fire-climatevegetation interactions in the boreal forest.

Acknowledgements

We thank L.B., E.D. and R.J. for their assistance in data collection. We also thank the two anonymous referees for their insightful comments. Financial support was provided to H.C. by the Natural Sciences and Engineering Research Council of Canada (NSERC DG283336-09), Ontario Early Researcher Award, and Northern Ontario Heritage Foundation Corporation, Y.B. by NSERC and to A.A. by the CNRS (BOREOFIRE, PALEO2, INSU). The first author acknowledges NSERC, Forest Ecosystem Science Co-operative Inc, and Resolute Forest Products for an Industrial Postgraduate Scholarship.

CHAPTER THREE The effects of forest fuel connectivity and drainage on spatiotemporal dynamics of Holocene fire regimes in the central boreal forest of North America.

Abstract

Understanding fire regime dynamics is central to predicting forest structure and compositional dynamics of fire driven boreal forests. Spatial and temporal variations in fire frequency in central Canadian boreal forest over the last 10k years were examined to evaluate the influence of bottom-up controls on the regional fire regime. We performed macroscopic-charcoal analysis of sediment cores from six lakes and a GIS analysis of landscape features to investigate the effects of regional non-climatic factors relevant to fire frequency. Our results revealed longer fire intervals (FI) were associated with increasing time before present and near sites with high nonfuel cover. Also there were significant interactions between time and regional environmental factors like drainage and organic terrain. Poorly connected regional landscapes had significantly higher FIs than highly connected landscapes, particularly between 10k – 4k cal a BP, suggesting that natural fuel break coverage may have limited fire spread. Increased fire activity over time among poorly connected sites may be related to climatic changes facilitating increased forest connectivity via the expansion of organic terrain and associated Picea mariana. Results clarify some fire regime controls and imply that climate changes will interact with environmental factors to affect future boreal fire regimes.

Introduction

Fire is a widespread and key disturbance in many terrestrial ecosystems (Bond and Keeley 2005; Bowman *et al.* 2009). In boreal forests, naturally occurring fire is a primary driver of ecosystem dynamics (Weber and Flannigan 1997; Ryan 2002; Clemmensen *et al.* 2013). For several decades, forest fires have been increasing in frequency (Kelly *et al.* 2013), and size and severity (Kasischke and Turetsky 2006) in response to ongoing climate changes. The influence of fire intervals (FI) and the potential impacts of altered fire regimes on ecosystem structure and function have driven investigations into the causes and consequences of future changes in fire frequency for ecosystems and resource management (Littell *et al.* 2009; Westerling *et al.* 2011; Girardin *et al.* 2013b).

Variations in natural fire regimes are generated by both the stochastic nature of fire and complex interactions among environmental factors such as weather, fuels, ignition pattern and topography (Van Wagner 1987; Bowman *et al.* 2009; Parisien *et al.* 2011; Ali *et al.* 2012). In the North American boreal forest, fire-environment relationships differ among regions, indicating that factors controlling fire regimes vary throughout the biome (Flannigan *et al.* 2005; Fauria and Johnson 2008; Parisien *et al.* 2011); moreover, the relative influence of these factors varies, because top-down (e.g., climate) and bottom-up (e.g., connectivity) controls on fire ignition and spread change, depending on the temporal and spatial scales examined (Gavin *et al.* 2003; Parisien *et al.* 2011).

Climate is the principal control of fire (Johnson 1992; Payette 1992) and influences fire at a broad scale (Fauria and Johnson 2008) by creating a latitudinal gradient in the severity of fire weather conditions, influencing the length of the fire season and proportion of flammable vegetation. Following the end of the Earth's latest ice age, the planet entered an interglacial

period of time known as the Holocene, the warmest portion of which in boreal North America, the Holocene Climatic Optimum (HCO), prevailed from about 9k to 5k a BP. Afterward, climate cooled and precipitation increased as the region experienced a Neoglacial period (c. 4.5k – 2.5k a BP). Several paleofire studies (Ali *et al.* 2009a; Kelly *et al.* 2013; Senici *et al.* 2013) from the North American boreal forest indicate significant fire frequency changes concurrent to Neoglacial cooling, the direction and magnitude of such changes are variable between boreal regions. Fire frequency decreased in some eastern coniferous boreal forests (Ali *et al.* 2009a; Hely *et al.* 2010) due to a reduced fire season length, and increased in some eastern mixedwood (Carcaillet *et al.* 2010) and western boreal forests (Lynch *et al.* 2004b; Kelly *et al.* 2013). The variable fire frequency responses across the boreal region suggest interactions with bottom-up controls that may influence future fire regime variation to climatic changes.

Environmental factors operating at regional-to-local scales like forest connectivity, surficial deposits and drainage (Cyr *et al.* 2007; Mansuy *et al.* 2010) are increasingly recognized in paleofire research as important drivers of temporal and spatial variation in millennial-scale wildfire activity (Ali *et al.* 2009a; Genries *et al.* 2012; Barrett *et al.* 2013; Senici *et al.* 2013). However, local environmental factors are numerous and variable and interact with climate making their effects on fire regimes difficult to partition. For instance, fire size is influenced by forest connectivity through a function of fuel loads, fuel-bed bulk density and fuel moisture (Miller and Urban 1999, 2000). Dendrochronological approaches to understanding modern fire regime spatial variation confirm that lakes, watercourses and wetlands can disrupt fuel continuity and thereby inhibit fire spread and final size (Larsen 1997; Hellberg *et al.* 2004; Cyr *et al.* 2005; Senici *et al.* 2010). Similarly, surficial deposits and drainage have altered fire cycles in the eastern Canadian boreal forest (Bergeron *et al.* 2004; Mansuy *et al.* 2010) by moderating the

distribution and growth of vegetation and consequently fuel arrangement, distribution and moisture. The relative influences of local environmental controls on North American boreal fire regimes may change over timescales longer than existing observational records $(20^{th} - 21^{st}$ centuries), thereby obfuscating the long-term relationships between drivers and fire regimes. In the North American boreal forest where detailed fire records prior to the early 1970s are incomplete, a paleoecological approach to fire history reconstruction can reveal changes in fire activity occurring at centennial-millennial timescales (Gavin *et al.* 2003) and clarify some long-term fire-environment relationships. We intend to use paleofire reconstructions to explore the influence of environmental factors specifically landscape connectivity, organic terrain and drainage type on regional fire regimes. Understanding the roles of climatic and non-climatic factors in influencing fire frequency is a basic issue for forecasting how fire regimes might change in response to global warming.

We examined spatial and temporal variations in fire interval for a mixedwood boreal forest in central Canada (Northwestern Ontario) over the last 10,000 years. This region is well suited for exploring the effects of connectivity and surficial geology because of the highly variable distribution of water bodies, surficial deposits and drainage at the landscape scale and relatively homogenous elevation, slope and aspect. Fire histories were reconstructed using high-resolution macroscopic charcoal records obtained from six lakes separated by distances of less than 80 km. Spatial analysis of landscape features was performed to examine local environmental controls on regional FIs. Given the relative proximity among sites, we hypothesize that if climate is the main factor controlling regional fire frequency comparable fire histories should exist among all sites. Because forest fuel connectivity is a strong predictor of regional fire frequency (Miller and Urban 2000; Peters *et al.* 2004), our main hypothesis predicts

that poorly connected landscapes as inferred through high modern non-forest cover in the form of open-water and wetlands will have longer FIs throughout the Holocene. We expect that dry surficial deposit drainage (SDD) could promote conditions favourable to drying of soils, fuels and vegetation, thereby resulting in shorter FIs where dry SDD comprise large portions of the regional landscapes. Finally, we expect the influence of local environmental controls on FIs to decrease in response to shifting landscape cover following deglaciation and ongoing climatic changes.

Materials and methods

Study area

We conducted this study at six lakes, Avril (AVR, 49°22'7" N; 89°25'6" W), Ben (BEN, 49°21'25" N; 89°46'10" W), Beaver (BVR, 49°32'2" N; 90°24'17" W), Dom (DOM, 49°26'22" N; 89°37'55" W), DuBerger (DUB, 49°25'6" N; 90°28'33" W) and Small (SML, 49°34'52" N; 90°23'08" W; 478), in the boreal forest of Northwestern Ontario, Canada (Fig. 3.1). The regional climate is humid continental, with short and warm summers and long and cold winters. The lakes (Table 3.1) are located within the Moist Mid-Boreal (MBx) ecoclimatic region (Ecoregions Working Group 1989), characterized by mean summer temperature of 14°C and mean winter temperature of -13°C. Mean annual precipitation ranges between 700-800 mm. The forest is within the boreal mixedwood region (Baldwin *et al.* 2012), characterized by a complex mosaic of forest types varying in structure and in relative proportions of coniferous and broadleaf tree species. Regional forests are dominated by conifers (needle-leaved, cone-bearing trees) of the family Pinaceae, mainly *Picea mariana* (Mill.) B. S. P. and *Larix laricina* (Du Roi) K. Koch on wet organic soils and *Pinus banksiana* Lamb. on sandy and loamy soils. Broad-leaved trees include *Betula papyrifera* Marsh. and *Populus tremuloides* Michx. These trees tend to be

successional, forming pure stands of variable extent following disturbance contingent on predisturbance forest cover and substrate ecophysiological characteristics (Brassard and Chen 2006; Taylor and Chen 2011).



Figure 3.1 Study region. (a) The six study sites (white dots) in the central boreal forest of Northwestern Ontario, Canada. Inset is a map of Ontario highlighting the location of the sites. Blue polygons are open water and symbol polygons are wetlands surrounding the sites, (b) Lake Ben (BEN), (c) Lake Dom (DOM), (d) Lake Avril (AVR), (e) Lake DuBerger (DUB), (f) Lake Beaver (BVR), and (g) Lake Small (SML).

	Study site					
	Avril	Ben	Dom	Beaver	DuBerger	Small
Characteristics	(AVR)	(BEN)	(DOM)	(BVR)	(DUB)	(SML)
Latitude (N)	49°22'7	49°21'25	49°26'22	49°32'2	49°25'6	49°34'52
Longitude (W)	89°25'6	89°46'10	89°37'55	90°24'17	90°28'33	90°23'08
Lake surface (ha)	2.7	1.6	3.1	3.2	0.4	0.5
Water depth (m)	5	5	6	4	4	4
Landscape cover						
Connectivity and drainage						
Modern connectivity	High	High	High	Poor	Poor	Poor
Open-water cover (%)	5.3	9.5	1.5	12.3	16.2	16.4
Wetland cover (%)	2.1	3.5	6.7	9.8	7.8	7
Dry (%)	85.6	60.8	62.1	71.1	76.1	72.2
Wet (%)	7.3	25.9	26.3	6.8	0	4.4
Surficial deposits						
Glaciofluvial landforms (%)	14.7	50.6	39	28.2	16	47.3
Morainal landforms (%)	71.9	19	30.6	42.9	33.2	24.5
Organic terrain (%)	6	13.3	22	6.8	26.8	4.8
Mean (± SD) elevation (m.a.s.l.)	472 ± 19	476 ± 9	469 ± 13	468 ± 6	479 ± 5	473 ± 5
Mean (± SD) aspect (degrees)	142 ± 120	114 ± 124	150 ± 112	118± 121	84 ± 110	110 ± 119
Sediment records						
Core length (cm)	480	660	460	375	415	390
Sedimentation rate	$0.056 \pm$	$0.086 \pm$	$0.051 \pm$	$0.047 \pm$	$0.050 \pm$	$0.063 \pm$
(cm/yr, mean ± SD)	0.036	0.064	0.033	0.046	0.032	0.069
(yr/sample, mean ± SD)	21 ± 7	16 ± 9	23 ± 9	31 ± 14	25 ± 10	26 ± 12
Median signal-to-noise index	5.1	8.7	7.5	4.9	4.9	5.9

Table 3.1 Characteristics of the study lakes, surrounding landscapes within a 5 km radius, and charcoal records in the central boreal forest of Northwestern Ontario, Canada

-

Sites were chosen on two criteria. First, to evaluate the potential effects of various geophysical characteristics on fire activity, preliminary regional topographic mapping and onsite evaluation was required to provide qualitative evidence that an equal number of sites would be located in two distinct regional landscapes. Second, lakes needed to have small surface area (< 5

ha), deep water depths (> 3 m) and lack inflowing or outflowing streams to optimize local fire reconstruction with undisturbed sediment deposits.

Sampling

Sediment sequences were extracted from the deepest point in each lake in the form of 1 m overlapping cores using a modified Livingstone piston corer (Wright *et al.* 1984). A Kajak-Brinkhurst (KB) gravity corer was used to collect recent accumulated material at the water-sediment interface, and was extruded on site at 1 cm intervals. Sediment cores were wrapped in polyurethane and aluminum foil for preservation and transported to the laboratory where they were sliced into disks at contiguous 1 cm intervals.

Dating and age-depth models

Chronologies are based on radiocarbon dating by ¹⁴C Accelerated Mass Spectrometry (AMS) measurements performed on plant macroremains and bulk organic sediment when macroremains were not abundant enough for AMS measurements (Supporting Information (SI): Table S1). The CALIB program (Reimer *et al.* 2004) was used to calibrate radiocarbon ages to calibrated years before present (cal a BP; 1950 CE) using the IntCal09 (Reimer 2009) calibration curve. Age-depth models were created with a weighted cubic smoothing spline derived from 1000 bootstrapped samples from the calibrated age distributions using the program MCAgeDepth (Higuera *et al.* 2009).

Charcoal analysis and fire history reconstruction

Contiguous subsamples (1 cm intervals, 1 cm³ each) were taken from each sediment sequence for charcoal analysis. To help distinguish charcoal from other biological materials, subsamples were deflocculated in hot 10% KOH solution, bleached in 6% sodium hypochlorite (NaClO) solution and then wet-sieved through a 160 μ m mesh. Charcoal fragments larger than 160 μ m were identified, counted and measured for surface area under a × 20 stereo microscope using an attached digital camera connected to WinSeedle (Regent Instruments Incorporated, Quebec). Charcoal concentrations (cumulative area/cm³) were multiplied by the estimated sedimentation rates (cm/yr) estimated from our age-depth models to obtain charcoal accumulation rates (CHAR, mm².cm⁻².yr⁻¹).

Fire events at each lake were identified through peak analysis of its sedimentary charcoal record using CharAnalysis 1.1(Higuera 2009), available online at http://charanalysis.googlepages.com. To account for uneven sampling intervals resulting from variable sediment accumulation rates among sites, prior to decomposition, all charcoal data were interpolated to a temporal resolution of 25 years/sample, corresponding to the approximate median sample resolution (23.5) of the six records (Table 3.1). Each CHAR series was decomposed into "background" (Cback), and "peak" (Cpeak) subpopulations. Cback, is composed of low-frequency variations in the charcoal record and represents changes in charcoal production (regional biomass burning), sedimentation mixing and secondary charcoal transport (Clark et al. 1996); we estimated *Cback* with a locally-weighted regression using a 1000-year moving median applied to the raw charcoal series. Chack were subtracted to obtain a residual series, Cpeak. We assume that Cpeak is composed of two subpopulations, Cnoise, representing variability in sediment mixing, sampling and analytical and naturally occurring noise, and Cfire, representing charcoal input from local fire events (Higuera et al. 2010). Cfire and Cnoise distributions were estimated in a 1000-year moving window using a Gaussian mixture model and at the centre of each window a threshold was defined as the 99th percentile of Cnoise to separate samples into "fire" and "nonfire" events. For each record, we chose the window width that maximized a signal-to-noise index

(SNI > 3) and the goodness-of-fit between the empirical and modelled C_{noise} distributions (KS-test, *P*-value > 0.05) (Higuera *et al.*, 2009). We did not screen peaks based on charcoal counts of each peak as in Higuera *et al* (2008; 2009), because this procedure is specific to charcoal count data only (Ali *et al.* 2009b).

Fire intervals (FIs) were calculated as the number of years between two consecutive fire events. Median FIs occurring during the HCO (9k – 5k cal a BP, hereafter referred to as early-Holocene) and during the Neoglacial and modern period (5k – 0 cal a BP, late-Holocene) between highly and poorly connected sites were calculated. To test whether fires occurred synchronously (\pm 100 years) between highly and poorly connected sites in these time periods, we used the L function, a modified version of Ripley's K-function in the program K1D (Gavin 2010).

Landscape analysis

We analysed landscapes at spatial scales larger than local fire event detection inferred from charcoal peaks ($\sim 1-3$ km radius) to evaluate the potential importance of regional-to-local nonclimatic environmental factors on fire spread and occurrence. We characterized the local environment at 3 – 5 km radiuses around each lake using ESRI's ArcMap version 10.1 (ESRI 2012). We present the 5 km results (Table 3.1). Surficial deposit, open-water and wetland cover were calculated using data from Digital Northern Ontario Engineering Geology Terrain Study (Ontario Geological Survey 2005) and CanVec (Natural Resources Canada 2013). Surficial deposits were classified as glaciofluvial, encompassing outwash, delta and esker, morainal, which included ground moraine and mixed terrain, and organic terrain. In this paper we make a distinction between wetlands and peatlands; wetlands include riparian fen or bog dominated by graminoid non-woody vegetation, marshes and shallow water wetlands (< 2 m deep); peatlands encompass terrain where organic substrate > 40 cm, drainage is poor, and dominant vegetation is *Picea mariana* and *Larix laricina*. Wetland and peatland type and extent were verified on site using the Canadian Wetland Classification System (National Wetlands Working Group 1997). Regional SDD was classified into wet and dry. Total modern non-fuel cover (open water + wetlands) representing a minimum value for regional landscape connectivity was calculated and used to categorize each site as highly or poorly connected. Additionally, total organic terrain (wetlands + peatlands) was calculated, representing maximum landscape cover by organic material and associated coniferous vegetation. Area calculations were converted to percent cover. Elevation, slope and aspect were calculated using a 8 – 23-m resolution digital elevation model (Natural Resources Canada 2007).

Statistical analysis

We used generalized linear mixed effect models (GLMM) in the package *arm* (Gelman *et al.* 2013) in the statistics program R (R Development Core Team2013) to model both fixed-effects and random effects. Since some time is measured on a very different scale than our geophysical predictors, all independent variables were standardized prior to analysis. All models used lake (site) as a random variable to account for variance derived from the hierarchical levels in the data. To examine temporal and connectivity controls on fire frequency we examined three scenarios. In Scenario 1, modern wetland cover is considered an impediment to fire spread, similar to open-water; correspondingly in Model 1, we considered the effects of time and modern non-fuel cover on FIs.

(3-1)

$$FI_{ijk} = \mu + T_i + NF_j + T \times NF_{ij} + \pi_k$$

Where FI_{ijkl} is FI (years); T_i is time (cal a BP), NF_j is non-fuel cover in percentage; π_k is random effect of sampling lake.

Given that wetland abundance and composition likely changed through time and may not have been an impediment to burning we developed additional scenarios. In Model 2 – we examine the effects of modern open-water and wetlands separately, where open-water is a more conservative estimate of landscape connectivity and wetlands are treated as potential fuel. In Model 3 – we examine the effects of modern open-water and total organic terrain, total organic terrain represents a maximum value for regional peat and associated vegetation.

$$FI_{ijkl} = \mu + T_i + OW_j + W_k + T \times OW_{ij} + T \times W_{ik} + \pi_l$$
(3-2)

Where FI_{ijkl} is FI (years); T_i is time (cal a BP), OW_j is modern open-water cover in percentage; W_k is modern wetland cover (%) in Model 2 and total organic terrain (%) in Model 3; π_l is random effect of sampling lake.

We used an information-theoretic approach based on corrected Akaike Information Criterion (AIC_c) (Burnham and Anderson 2002; Stauffer 2008) to select the most parsimonious model. To avoid multicollinearity among predictors (Graham 2003), we investigated the correlation matrix of environmental factors (SI; Table S2). Variables were omitted from full models in favour of our initial model variables when bivariate correlations exceeded $|r_{spearman}| > \pm$ 0.50. In Model 1, we added main effects and interaction terms of peatlands and dry SDD. In Model 2, we added dry SDD and the interaction term with time. No additional variables were added to Model 3. Model goodness of fit was assessed by R², calculated using the methodology presented in Nakagawa & Schielzeth (2013). For mixed-effects models, R² is evaluated as marginal R^2 and conditional R^2 (Vonesh *et al.* 1996) where marginal R^2 represents variance explained by fixed factors, and conditional R^2 represents variance explained by both fixed and random factors.

Fire reconstructions from lake sediments are strongly influenced by user choices during statistical treatment (Blarquez *et al.* 2013; Brossier *et al.* 2014), consequently CHAR records were analysed using 20 and 30 year sample resolutions and Models 1 - 3 were analysed to assess GLMM robustness using FIs from the median (25 year) sample resolution reconstruction.

Results

Dating and age-depth models

The age-depth models represent 10,004 at AVR, 11,700 at BVR, 10,212 at BEN, 10,299 at DOM, 10,108 at DUB and 10,007 at SML years of sedimentation. The models (Fig. 3.2) are comparable in sedimentation rate, with BEN having the highest mean sedimentation rate at 0.0866 ± 0.0534 cm yr⁻¹ and BVR having the lowest at 0.0468 ± 0.0495 cm yr⁻¹ (Table 3.1). Models for BEN, BVR and SML exhibit acceleration in the sedimentation rate between *c*. 5k – 3.5k cal a BP.



Figure 3.2 Age-depth models for sediment cores from the six lakes calculated using a cubic spline. Hollow circles are ¹⁴C ages, and error bars represent estimated 95% confidence intervals based on 1000 bootstrapped samples of the calibrated dates. Ages are expressed in 1k years BP.

Charcoal analysis and fire history reconstruction

Global SNI values for all charcoal records are high (median SNI > 3.0) (Fig. 3.3d,e,f,m,n,o) and show a clear separation between background charcoal and fire event signals indicating that these records are suitable for fire history reconstructions using peak analysis (Kelly *et al.* 2011). There are two instances where SNI < 3.0 exist at BVR between *c*. 10.5k and 10k cal a BP (Fig 3m) and DUB between *c*. 8.4k – 8.2k cal a BP, however, overall median SNIs are high (BVR = 4.7 and DUB = 4.3), and the period of low SNI is short (< 300 years), consequently we felt safe to include these sites in subsequent analyses.

In total, 50 fires were detected at AVR, 46 at BEN, 47 at DOM, 39 at BVR, 39 at DUB and 36 at SML, (Figs. 3g,h,i,p,q,r). FI reconstructions (raw FIs interpolated to annual values and smoothed with LOESS) indicate that fires were most frequent late-Holocene among all sites save DOM. Among highly connected sites, fires recur at relatively consistent intervals throughout the reconstruction; fires occur most frequently between *c*. 10k and 9k cal a BP at AVR, between *c*. 6k and 4k cal a BP at BEN and between *c*. 8k – 6k cal a BP at DOM. Among poorly connected sites, fire occurrence was somewhat irregular following deglaciation but increased with time; fires were most frequent *c*. 4k cal a BP at BVR and DUB, and between *c*. 4k and 2k cal a BP at SML.



Figure 3.3 Sediment-charcoal records over time for AVR (a, d, g), BEN (b, e, h), DOM (c, f, i), BVR (j, m, p), DUB (k, n, q), and SML (l, o, r). (a-c, j-l) Charcoal accumulation rates plotted against time. Vertical gray lines are the interpolated CHAR peaks, the heavy black line represents background values and the light gray dashed line indicates the threshold used to identify potential fires. (d-f, m-o) Signal-to-noise index plotted against time. The dashed gray line indicates the threshold (SNI = 3) used to evaluate the suitability of the charcoal record for peak detection. (g-i, p-r) Smoothed fire return intervals (FRIs) plotted against time. Circles represent fire events and the corresponding fire interval.

Median FIs (mFI) were longer among poorly connected sites throughout the Holocene (Table 3.2); differences were especially pronounced in the early-Holocene where mFI variations between highly and poorly connected sites was > 100 years. Fire occurrence was predominantly independent among highly connected sites (Fig.3.4c, d), with limited temporal synchrony between 1000 - 1100 year window widths in the early-Holocene and 800 - 1000 year window widths in the late-Holocene. Among poorly connected sites fires occurred independently throughout the early-Holocene (Fig.3.4a), but were mostly synchronous throughout multiple window widths during the late-Holocene (Fig.3.4b).

Table 3.2 Median fire intervals (years) in the early (10k - 5k cal a BP), late (5k - 0 cal a BP) and complete (10k - 0 cal a BP) Holocene time periods among highly and poorly connected sites using three fire history reconstructions at different temporal sample resolutions.

	Poorly c	Poorly connected		Highly connected		Complete sequence	
	Early	Late	Early	Late	Poorly	Highly	
FI (years)	300 ± 52	175 ± 45	200 ± 39	150 ± 38	225 ± 32	175 ± 29	



Figure 3.4 Synchrony analysis of fire occurrence between poorly connected (a, b) and highly connected (c, d) sites in the early (a, c) or late (b, d) Holocene time periods. In each plot, fire event synchrony is assessed by comparing the bivariate L function (black line) to a bootstrapped 95% confidence envelope (gray lines). Fire occurrence within a specified time window is considered significantly synchronous if the L function exceeds the upper CI, asynchronous if the L function exceeds the lower CI, or occurring independently if no CIs are exceeded.

Landscape analysis

The spatial distributions of land cover surrounding the six lakes are dissimilar between highly and poorly connected sites (Table 3.1). The percentage of open-water coverage was highest at SML (16.4%) and lowest at DOM (1.5%). Wetland coverage was highest at BVR (9.8%) and lowest at AVR (2.1%). Total organic material was highest at DOM (28.7%) and lowest at AVR (8.1%). Dominant surficial deposits include glaciofluvial in the form of eskers, ice-contact deltas and outwash plains, and ground moraine. At DUB a large portion (26.8%) of the regional landscape is classified as organic terrain (Table 3.1), however this area is a heterogeneous mixture of peat, sand and gravel in glaciofluvial outwash with a subordinate landforms of till and sand in ground moraine with overall dry SDD; accordingly this area was not considered a peatland nor was it pooled in total organic terrain at this site. Elevation, slope and aspect were relatively homogenous among all sites.

Temporal and spatial effects on fire interval

We present results from the median resolution fire reconstruction as there were no significant differences among different temporal resolutions (SI: Table S3). In all three GLMMs longer FIs are associated with increasing time into the past, indicating an overall increase in fire frequency following deglaciation to present (Table 3.3; Fig.3.5a). In Model 1, FIs were longer near sites with higher modern non-fuel cover (Fig.3.5b, c), and FIs were longer near sites with high nonfuel cover and increasing time BP (Fig.3.5b). The main effect of peatlands and dry drainage were not significantly related to FIs, however their interactive terms with time were significant; FIs were longer near sites with high peatland cover and increasing time BP, and FIs were shorter near sites with high dry drainage and increasing time BP. For Model 1 the marginal R^2 was 0.82 and the conditional R^2 was 0.86. In Model 2, longer FIs were found near sites with higher coverages of both open-water and wetlands (Fig.3.5c). The interactive terms between time and these variables were both significant but differ in magnitude, with longer FIs near sites with high open-water and wetlands and increasing time BP. The effect of dry drainage was not significant; the interactive term with time was significant with lower FIs associated with increasing landscape dryness and time BP. For Model 2 the marginal R^2 was 0.84 and the conditional R^2 was 0.86. In Model 3, longer FIs are associated with high open-water coverage; the open-water effect also increased with time BP. Total organic material was not found to significantly influence FIs; the interactive term with time was significant, indicating that longer FIs occurred

in the early-Holocene near sites with high organic landscape cover. For Model 3 the marginal R^2 was 0.70 and the conditional R^2 was 0.85.



Figure 3.5 Scatterplots and fitted fire intervals (FI) of significant fixed effects from generalized linear mixed models. (a) Time, (b) interactive effect of modern connectivity and time, and (c) modern non-fuel cover, open-water cover and wetland cover. For illustrative purposes we present simple linear regression lines. In (b) blue triangles and regression line represents highly connected sites (AVR, BEN and DOM) while red circles and regression line represents poorly connected sites (BVR, DUB and SML).In (c) the black regression line represents modern non-fuel cover, the medium grey line represents open-water cover and the light grey represents wetland cover.

Model	Term	Estimate	SE	p-value
Model 1	(Intercept)	5.450	0.015	< 0.001
	Time	0.075	0.004	< 0.001
	Non-fuel	0.119	0.015	< 0.001
	Peatlands	-0.012	0.015	0.407
	Dry SDD	-0.007	0.016	0.673
	Time × Non-fuel	0.013	0.004	0.001
	Time × Peatlands	0.016	0.004	< 0.001
	Time × Dry SDD	-0.057	0.004	< 0.001
Model 2	(Intercept)	5.450	0.010	< 0.001
	Time	0.075	0.004	< 0.001
	Open-water	0.069	0.012	< 0.001
	Wetlands	0.075	0.012	< 0.001
	Dry SDD	0.008	0.012	0.471
	Time × Open-water	0.014	0.005	0.006
	Time × Wetlands	0.036	0.005	< 0.001
	Time × Dry SDD	-0.051	0.005	< 0.001
Model 3	(Intercept)	5.453	0.028	< 0.001
	Time	0.076	0.004	< 0.001
	Open-water	0.113	0.036	0.002
	Total organic	0.014	0.037	0.705
	Time × Open-water	0.041	0.005	< 0.001
	Time × Total organic	0.074	0.005	< 0.001

Table 3.3 Parameter estimates and standard error of the fixed effects from generalized linear mixed models with fire intervals as dependent variable (Poisson distribution; log-link).

Discussion

Our results suggest regional patterns of fire frequency observed throughout the Holocene in the central boreal forest were controlled by a complex combination of local environmental factors and regional climate. Overall, FIs have decreased over time, indicating that regional fire activity has increased from deglaciation to the present. Fire intervals were significantly shorter in the early-Holocene near sites with high percent cover of dry SDD and longer FIs tend to occur near sites where landscapes have abundant non-fuel cover in the form of open-water lakes and treeless

wetlands, indicating that the long-term influence of climatic changes on central boreal forest fire regimes may differ by drainage and connectivity. Additionally, organic terrain cover in the modern form of treed peatlands with wet drainage was significantly associated with longer FIs with increasing time BP. The temporal changes in FIs associated with these environmental factors support the idea that top-down controls, like climate, strongly influenced regional centennial-millennial scale fire frequency dynamics with bottom-up controls affecting fire ignition and spread thereby generating variable FI distributions.

The observed differences in FIs between poorly and highly connected landscapes may be a consequence of forest fuel continuity on fire behavior, since heat transfer is reduced by fuel discontinuity, fire spread was diminished amongst landscapes with large fuel gaps. Among sites considered poorly connected where regional landscapes are fragmented by large areas of openwater and wetland cover, FIs decrease from deglaciation to present. During the HCO of the early-Holocene when it was significantly warmer than it is today, mFIs were approximately 42% longer (Table 3.2) than those of the last 5k years; even though the regional climate was favorable for burning, fires ignition and spread may have been limited by a lack of burnable materials. Similar mFIs and synchronous fire occurrence among poorly connected landscapes in the Neoglacial and modern period of the late-Holocene indicate that large-scale processes, likely climate and biome transformations were the main factors determining fire frequency, and the effect of natural fuel breaks were diminished. These results contrast those of highly connected forests where mFIs did not change significantly with time nor were fire events significantly synchronous over multiple temporal windows. These observations complicate our expectation of generally increasing fire activity because the increase was driven primarily by poorly connected

sites. The non-uniform fire frequency response suggest that in addition to non-fuel cover other ecological mechanisms and climatic interactions contributed to the observed FI variations.

Similar mFIs in the late-Holocene amongst all sites may be explained by Neoglacial climate cooling; cooler temperatures and increased precipitation directly influence soil hydrology, creating environmental conditions unfavorable for fire ignition and spread, but also conditions favorable for the growth and development of highly flammable coniferous vegetation. Throughout the Neoglacial period *Picea mariana* expanded in eastern mixedwood boreal forests (Carcaillet et al. 2001; Carcaillet et al. 2010; Genries et al. 2012), and in boreal forest of Alaska (Lynch et al. 2002; Higuera et al. 2009; Kelly et al. 2013). In central boreal forest where the fire regime is characterized by few very large fires occurring under specific weather conditions, regional landscapes could have become highly flammable during intermittent dry periods resulting in very large and severe fires able to overcome natural fuel breaks which may account for the significantly decreased FIs and synchronous fire event occurrence among poorly connected sites in this time period. One plausible explanation for why FIs did not significantly change in the early-Holocene among highly connected sites, is that the high-severity fires also promoted increased broadleaved growth (Chen et al. 2009; Kelly et al. 2013). Deciduous pioneers like Betula papyrifera and Populus tremuloides, may have increased post-fire recruitment in the late-Holocene on dry SDD where increased soil moistures may have be unfavorable for Pinus banksiana (Rudolph and Laidly 1990), thereby producing a negative feedback to increased regional burning (Beck et al. 2011; Johnstone et al. 2011; Girardin et al. 2013a). These interpretations are partially supported by local plant macrofossil data (Senici et al. 2013), where decreasing local *Pinus banksiana* and increasing local *Betula papyrifera* and *Picea*
mariana abundance were found after *c*. 4.5k cal a-BP at BEN (highly connected), and *Picea mariana* abundance was continuous c.4.5k cal a BP at SML.

Shorter FIs in the early-Holocene tended to occur where dry surficial deposit drainage was high. Regional dry SDD highlights the effects of moisture on fuels and as a result, the potential to ignite and propagate fire. Although, it is likely dry SDD was responsible for shorter FIs due to ecological effects on soil moisture because of the length of time these trends persist, it is possible that dry SDD cover may indicate differences in the time required for regional landscapes to become suitable for forest vegetation and subsequent fire occurrence due to different temporal and spatial patterns of the retreating Laurentide Ice Sheet among sites. The earliest recorded post-glacial fires and shortest first fire intervals occur at DUB and AVR where dry SDD is the highest and Latitude are among the lowest indicating these sites may have had a head start in regional vegetation development. Alternately, shorter FIs in the early-Holocene associated with high dry SDD may be a reflection of the post-glacial landscape, where a greater proportion of terrestrial area influenced the biomass available for burning and subsequently fire event detection as inferred through charcoal peaks. Although surficial deposit and drainage compositions can be a main controlling factor of fire regimes when climate seems less conducive to fire (i.e. the Neoglacial period) (Mansuy et al. 2010), we found no such changes in fire activity amongst our sites in the late-Holocene. This may be because differences in SDD only become significant under certain climatic conditions, like when the level of moisture is limiting to plant growth on specific substrates (Brubaker 1975). It is also possible that SDD combinations captured in our landscape analysis of our study area were not diverse enough to generate regional fire frequency variations.

In comparison with lakes, wetlands are highly dynamic fuel breaks since hydrology and vegetation varies over time scales much shorter than that of a paleoecological reconstruction, thus complicating our understanding of their potential for stopping fires (Hellberg et al. 2004). We sought to mitigate this uncertainty by categorizing organic terrain into wetlands; comprised of non-woody, riparian, marsh and shallow water, and peatlands encompassing other wooded types (i.e. swamp, bogs and fens). While it is clear that wetland cover contributed in reducing fire spread, peatlands as a whole may not have acted as absolute fuel breaks throughout the Holocene as only with increasing time BP are regional peatlands associated with longer FIs. This is in agreement with studies that indicate organic deposits with poor drainage tend to burn less (Mansuy et al. 2010), however that this trend was observed only with increasing time BP indicates several possible scenarios involving interactions with climate and the post-glacial landscape conditions in the early-Holocene. In the first scenario, the pattern and timing of peatland succession and development may explain the observed FI variation among sites where wetlands and organic terrain comprise large portions of the regional landscape. Many peatland successional pathways in northwestern Ontario originate from lakeshores, shallow water depressions and ponds (Harris et al. 1996; Rydin and Jeglum 2006); there is an average 4,000 year (Gorham et al. 2007) lag time between deglaciation and peak peat development due to the time it takes for land to become amenable to peat formation (Halsey et al. 2000). Consequently, modern terrestrial cover considered peatland may have acted as non-fuel following deglaciation, thereby decreasing regional forest connectivity resulting in longer FIs in the past but not the present when regional peatlands are a viable source of forest fuels. Alternatively, in scenario two, rising temperatures following the end of the Neoglacial climate cooling period (c. 2.5k cal a. BP) (Kerwin et al. 2004; Viau et al. 2006a; Viau and Gajewski 2009), may have increased fire

activity in peatlands as they have large organic matter stocks in vegetation and surface soil layers that may have become more susceptible to burning during dry weather periods (Turetsky *et al.* 2004). Decreased FIs at these sites may be ecologically significant as over the past 10k years boreal peatlands have functioned as net sinks for carbon (Thornley and Cannell 2004; Kashian *et al.* 2006; Lecomte *et al.* 2006), and increased fire activity could release large amounts of atmospheric CO_2 .

Conclusion

This study demonstrates complex spatial and temporal variability in fire regimes over the past 10k years. Our results clearly show the extent of the variability of the fire intervals between two distinct landscapes and emphasize that the influence of long-term climate on boreal fire regimes at regional-to-local scales differs with forest connectivity. These results also highlight the importance of considering the effects of local landscape features when interpreting Holocene-scale paleofire records in the boreal forest. These results clarify some patterns and controls of central boreal fire regime spatial variation, and imply that climate changes will interact with land cover to affect boreal fire dynamics in the future with far-reaching implications for forest ecology, carbon sequestration and wildfire management

Acknowledgements

We thank L.B, E.D. and R.J. for their assistance in data collection. Financial support was provided to Han Chen by the Natural Sciences and Engineering Research Council of Canada (NSERC DG283336-09), Ontario Early Researcher Award, and Northern Ontario Heritage Foundation Corporation, Yves Bergeron by NSERC and to Adam Ali by the CNRS (BOREOFIRE, PALEO2, INSU). The lead author acknowledges NSERC, Forest Ecosystem

Science Co-operative Inc., and Resolute Forest Products for an Industrial Postgraduate

Scholarship.

CHAPTER FOUR Fire-vegetation interactions in mixedwood central boreal shield forests of Northwestern Ontario, Canada

Abstract

Vegetation assemblages and associated disturbance regimes are spatially heterogeneous throughout the boreal forest due to the complex terrain, varying climate, and strong environmental gradients. Given this complexity, multiple sites describing postglacial vegetation and fire histories are needed to adequately understand forest development and ecosystem responses to varying climate and disturbance regimes. We analysed charcoal and pollen from sediments obtained from two kettle lakes in the central Canadian boreal forest in order to reconstruct fire return intervals and vegetation dynamics over the last 10,000 years. Overall, fire frequency has increased following deglaciation to present. The pollen-inferred vegetation exhibits different trajectories of boreal forest dynamics after afforestation. At both sites, Picea populations were most extensive shortly after deglaciation, with vegetation later shifting towards mixed woodlands with more extensive Betula papyrifera and Pinus banksiana populations. Mean fire intervals appeared to have no significant impacts on pollen communities, in terms of diversity or trajectories. Species-specific responses to median fire interval (mFI) were more varied, likely due to species specific fire adaptations, site specific local environmental conditions, climatic changes and competition. Although wildfire is the dominant natural disturbance in boreal ecosystems, this study does not support the hypothesis that changes in fire frequency is the main controlling factor of the long-term dynamics of boreal forest vegetation at regional spatial scale.

Introduction

In boreal forest ecosystems naturally occurring wildfire is a prominent disturbance factor driving vegetation dynamics over decadal to centennial scales (Burton et al. 2008; Bowman et al. 2009; de Groot *et al.* 2013a). Fire regimes, which describe long-term patterns of fire activity, vary significantly throughout the North American boreal forest and these variations play a central role in determining patterns of ecosystem response and landscape legacies left by fire (Turner 2010). Frequent forest fires have pronounced impacts on species composition, tree age, size structures and nutrient cycling; nonlinear responses to changes in fire frequency may cause rapid changes in plant community diversity and forest structure consequently providing different fuel compositions that will feed new fire regimes (Paine et al. 1998; Johnstone and Chapin 2006; Girardin et al. 2013a). In recent decades, boreal ecosystems have experienced marked changes in response to ongoing climatic changes, including increased burning (Gillett et al. 2004; Kasischke and Turetsky 2006; Kelly et al. 2013). Continued climate warming will increase the incidence of severe fire weather conditions (hot and dry) that favour larger and more severe fires (de Groot et al. 2013b; Flannigan et al. 2013). Furthermore, fire occurrence is also likely to increase as average temperatures rise and summer precipitation declines with far-reaching ecological and socioeconomic consequences (Marlon et al. 2009; Wotton et al. 2010).

The boreal forest of North America has low diversity and is comprised mainly of *Abies*, *Betula, Larix, Picea, Pinus* and *Populus*. Many boreal species have strategies to cope with fire and recolonize burns, either by seed or vegetative means. *Pinus banksiana* and *Picea mariana* use serotinous or semi-serotinous cones to protect and disperse seeds post-fire (Greene *et al.* 1999). *Populus tremuloides*, and *Betula papyrifera* regenerate vegetatively through root suckers or stump sprouts (Perala 1990; Greene *et al.* 1999), or seed in from outside the burn (Greene and

Johnson 1999). Fire sensitive species like *Abies balsamea* and *Larix laricina* can only seed into a burn from edges or unburned residual patches within the burn perimeter (Greene and Johnson 1999). Because fire is one of the most important abiotic disturbances in boreal forest ecosystems, it is critical to examine historical rates of change and the mechanisms interacting at various spatial and temporal scales to understand how future climate and vegetation changes could affect fire regimes and vice versa (Turner 2010; Hessl 2011). Retrospective studies using lake-sediment charcoal as a proxy for past fire regimes provide valuable long-term records, but are scarce in the central boreal shield region of Canada. When paired with high temporal-resolution and contiguous paleobotanical reconstructions, charcoal records can be used to infer the ecological impacts of changing fire regimes over millennia (Whitlock and Anderson 2003; Power *et al.* 2008; Power *et al.* 2010). Long-term records of vegetation and fire regime variability are necessary to inform land management policy and decision making (Gavin *et al.* 2007)

This chapter uses a Holocene macroscopic charcoal record from the central Canadian shield ecoregion in Canada that has been previously published (Senici *et al.* 2013), with a new pollen record. We present a 10,000-year record of vegetation and fire activity in the central boreal shield forest of North America. We examine the local area fire history through high resolution macroscopic charcoal analysis, and the regional vegetation history through pollen records established from lacustrine sediment cores from two separate lakes. We then integrate vegetation data with the charcoal record to investigate Holocene fire-vegetation interactions. Having previously identified differences in fire activity among the study sites (Senici *et al.* 2013), we expect significant changes in regional vegetation dynamics temporally concurrent to changes in local median fire interval (mFI).

Methods

Study Area

Lake surface area and morphometry strongly influence the relevant source-area shape for the collection of pollen (Sugita 1993) and macroscopic charcoal (Higuera *et al.* 2007). To provide the best historical records, small lakes with simple bathymetries, narrow littoral zones, and no inflow or outflowing streams were targeted for sampling (Fig. 4.1, Table 4.1) (Whitlock and Millspaugh 1996). The two lakes, Ben (BEN, 49°21'25" N; 89°46'10" W; 480 m above sea level [a.s.l]) and Small (SML, 49°34'52" N; 90°23'08" W; 478 m a.s.l) are located within the Central Canadian Shield boreal forests of northwestern Ontario, Canada. Climate is typically moist and cold with a mean summer temperature of 14°C and mean winter temperature of -13°C (Ecoregions Working Group 1989); annual precipitation ranging from 654 to 879 mm, and mean summer rainfall from 231 to 298 mm (Crins *et al.* 2009). Recent climate models indicate that this area is cooler than immediately adjacent areas of similar latitude and hence, exhibits a stronger boreal nature with lower mean annual temperatures (-1.7 to 2.1°C) and a shorter mean growing season length (161 to 182 days) (Mackey *et al.* 1996; Colombo *et al.* 2007).



Figure 4.1 Map of the study area.

	Ben	Small
Latitude (N)	49°21'25.02"	49°34'52.19"
Longitude (W)	89°46'10.36"	90°23'08.34"
Local woody vegetation	Pinus banksiana, Picea mariana, Betula papyrifera,	Picea mariana, Abies balsamea, Pinus banksiana
Lake surface (ha)	1.6	0.5
Water depth (m)	5	4

Table 4.1 Study lake characteristics. Local vegetation belongs to the three most dominant species listed in descending order of abundance at each site.

Regional modern vegetation

Dominant vegetation includes coniferous species like Picea mariana (Mill.) B.S.P. (black spruce), Picea glauca (Moench) Voss. (white spruce), Abies balsamea (L.) Mill. (balsam fir), Larix laricina (Du Roi) K. Koch. (tamarack) and Pinus banksiana Lamb. (jack pine), and deciduous species like Populus tremuloides Michx. (trembling aspen) and Betula papyrifera Marsh. (white bitch). Other extant species with small contributions to regional vegetation include Pinus resinosa Ait. (red pine), Pinus strobus L. (white pine), and Thuja occidentalis L. (Eastern white cedar). The most frequent landcover type in the region is mixed forest (23.5%), followed by coniferous forest (23%), water (17.1%), sparse forest (15.1%), deciduous forest (9.0%), and cutovers (5.7%) (Crins et al. 2009). Vegetation in the mixedwood portion is typically mixed closed-canopy forest, characterised by coniferous P. mariana, P. glauca and A. balsamea with deciduous P. tremuloides and B. papyrifera. In the coniferous portions, P. mariana and L. *laricina* predominate in swamps and peatlands in low-lying areas, while large areas of pure *P*. banksiana and mixed P. banksiana – P. mariana stands dominate on sandy materials throughout the region. Other trees such as *P. strobus* and *P. resinosa* occur in pockets near warm microclimates or sites protected from lethal fires (Bergeron et al. 2004) as the study region is at the modern northern range limit of these species.

Lake sediments

Sediment cores were taken atop frozen lake surfaces from the centre of each lake using a modified Livingstone-type square-rod piston corer $(100 \times 5 \text{ cm})$ (Wright *et al.* 1984) in the spring of 2010. Sediment cores were preserved in polyurethane and aluminum foil and transported to the laboratory. Cores were sliced into disks at contiguous 1 cm intervals. The most recently accumulated materials (soft, fine grained sediments) at the water-sediment interface were collected using a Kajak-Brinkhurst (KB) gravity corer (Glew 1989). These materials were extruded on site at 1 cm intervals.

Chronologies

Datings were obtained from terrestrial plant macroremains and charcoal or bulk organic sediments when macroremains were not abundant enough for ¹⁴C Accelerated Mass Spectrometry (AMS) measurements. Radiocarbon ages were calibrated with CALIB 6.1.1 (Stuiver and Reimer 2011) using the Intcal09 calibration curve (Reimer 2009). We used a weighted cubic smoothing spline derived from 1000 bootstrapped samples from the calibrated age distributions using the program MCAgeDepth 0.1 (Higuera *et al.* 2009).

Charcoal analysis and fire reconstructions

To reconstruct fire history, we performed charcoal peak analysis on each of our records. Contiguous subsamples of 1 cm³ were obtained from each sediment core at 1 cm intervals for charcoal analysis. Samples were deflocculated in a 10% KOH or 6% NaClO (bleach) to increase the visibility of charcoal particles and wet-sieved through a 160 μ m mesh. Macroscopic charcoal fragments were identified, counted and measured for surface area under a × 20 stereo microscope using an attached digital camera connected to WinSeedle (Regent Instruments Inc.2007). Empirical and modelling studies indicate that peaks of charcoal particles > 160 μ m are indicative

of fire occurrence within a distance of $\sim 1.2 - 3.0$ km from lakeshore (Higuera *et al.* 2007; Higuera *et al.* 2010).

Prior to decomposition into Charcoal Accumulation Rates (CHAR) each record was interpolated to equivalent time steps and volumes along the sedimentary sequences. We used a temporal resolution of 20 years/sample, the approximate median sample resolution of the two records (Table 4.1). Sedimentary macro-charcoal records (Craw) were broken down into low frequency variation (C_{background}) and peak components (C_{peak}) using a locally defined threshold to reconstruct and identify local fire events. We estimate C_{background} with a locally weighted regression (lowess) using a 1000 year window width. We then subtracted C_{background} from C_{raw} to obtain the residual high-frequency C_{peak}; C_{peak} is composed of two subpopulations C_{noise}, representing variability in sediment mixing, sampling and analytical and natural noise, and Cfire representing the occurrence of one or more local fire events (Gavin et al. 2006; Higuera et al. 2010). The C_{fire} and C_{noise} distributions were estimated in a 1000-year moving window using a Gaussian mixture model with a threshold defined as the 99th percentile of the modeled noise distribution. Residual CHAR values greater than the threshold was considered a C_{peak} and a potential fire event. We use the term fire event to describe events from C_{fire} that could represent one or more local fires occurring near our sites. The signal-to-noise index (SNI) was used to evaluate the effectiveness of the discrimination (Kelly et al. 2011). The goodness-of-fit (GOF) metric is used to assess the degree of contrast between C_{noise} and C_{fire} by comparing C_{noise} with the one derived from the Gaussian mixture model. Cfire was used to reconstruct local fire-return intervals (FRI) and fire frequency (FF) trends.

Pollen analysis and vegetation reconstruction

Subsamples of 1 cm³ were taken at 4 cm intervals from BEN and SML sediment sequences to be chemically treated following the standard procedures described by Faegri and Anderson (1989) in preparation for pollen identification. A small amount (0.5 mm) of Eucalyptus pollen solution (222, 367 ± 6403) was added at the beginning of each treatment as an exotic marker to estimate pollen concentrations (grains cm⁻³). Samples were soaked with hot 10% KOH and sieved through 700 µm mesh to deflocculate the sediments and remove humic acids and particles (plant macroremains) larger than known pollen or spores. Carbonate, silicate, and a fraction of the organic materials (cellulose) were eliminated with hot 10% HCl, 40% HF, acetic acid (CH3COOH), and acetolysis using acetic anhydride (CH3CO)2O and 97% sulphuric acid H2SO4. Samples were rinsed with water and 30% glycerine solution. Contrast dye was added to each sample to facilitate identification. Samples were mixed with glycerine and mounted on glass slides. Pollen was identified under 4000-1000x magnification to a minimum terrestrial pollen count of 300 pollen grains. Pollen treatment and identification was performed at Université Laval (UL) from September 2012 to January 2013. For identification we used the reference collection housed at UL and several dichotomous keys (Richard 1970; McAndrews et al. 1973; Moore et al. 1991).

Pollen data are expressed as a percentage of total pollen grains from terrestrial plants excluding pollen from obligate aquatics and spores. The genus *Picea* includes pollen from *P*. *mariana* and *P. glauca*. *Populus* pollen was identified as the *P.* cf. *tremuloides* type, which may include several species (mainly *Populus balsamifera* L. and *Populus grandidentata* Michx). *P. strobus* pollen grains can be differentiated from other pines; as a haploxylon subgenera its pollen grains characteristically possess distal verrucae whereas *P. banksiana* and *P. resinosa* are diploxylon and lack distal verrucae. Pollen grains from *P. banksiana* and *P. resinosa* cannot be reliably distinguished and were grouped together as *P. cf. banksiana*. To avoid the influence of an ecologically non-relevant pollen taxon, in subsequent statistical analyses the counts of nonidentified *Pinus* pollen (e.g. single sacci, broken pollen grains) were redistributed into the two identified *Pinus* classes (*P. cf. banksiana* and *P. strobus*) based on their relative abundance in each sample. Pollen zones were established through constrained incremental sums of squares cluster analysis (CONISS) of the relative abundance of pollen taxa (Grimm 1987) with the number of statistically significant zones obtained by comparison with the broken-stick model (Bennett 1996) implemented in the statistical computing program R 3.1.0 (R Development Core Team 2013) using the 'rioja' package (Juggins 2014).

Rate of change and diversity and richness

The rate of change (ROC) is a dissimilarity measurement, defined here as the amount of ecological change per unit time, determined from our pollen records. In these analyses, ROC was the dissimilarity based on a Euclidian distance between pollen assemblages from two adjacent pollen spectra divided by the time interval between them. ROC was used to detect any rapid vegetation changes like those associated with changes in fire frequency (Shuman *et al.* 2005; Urrego *et al.* 2009). Two estimations of ROC were calculated, the first with the complete terrestrial pollen dataset from each site and the second with only terrestrial taxa with a mean frequency greater than 1%.

Two biodiversity indices were used to estimate variations in the pollen-inferred vegetation composition: Shannon index (H') and Simpson index (D). These indices help compare the structure of those pollen assemblages in which the record of each species is comparable through time, and are calculated here using the complete terrestrial taxa dataset from each site.

H' is calculated by determining the proportional contribution of each taxon to the total number of pollen in the complete spectrum.

$$H' = -\sum_{i=1}^{S} p_i \ln p_i$$
(4-1)

Where p_i is the pollen frequency for the *i*th taxon. Simpson's diversity index (D) is calculated as:

(4-2)
$$D = \frac{1}{\sum_{i=1}^{S} p_i^2}$$

While Simpson's index squares the relative abundance, the weight of rare species is reduced relatively more than that of more abundant species, the Shannon's index includes the log-normal transformation of species, thus the importance of abundant species is reduced slightly relative to rare species. Both indices were used together to assess the potential roles of both abundant and rare taxa in the diversity scores at each lake. Temporal variability in pollen diversity was assessed by the Kruskal-Wallis test (KW-test) on mean pollen diversity scores per millennia.

To assess changes in pollen richness we performed a rarefaction analysis.

$$E(T_n) = \sum_{i=1}^{T} 1 - \left[\frac{\binom{N - N_i}{n}}{\frac{N}{n}} \right]$$
(4-3)

Where $E(T_n)$ is the expected number of taxa in a random sample of *n* individuals from a larger population of *N* individuals containing *T* taxa (Heck *et al.* 1975). We performed rarefaction

analysis on the complete set of terrestrial pollen. Shannon's and Simpsons's diversity indices and the rarefaction analysis were calculated using the 'vegan' package (Oksanen *et al.* 2010) in R (R Core Development Team 2013).

Fire and vegetation interactions

To describe the relationships among the pollen assemblages the pollen data from Lake Ben and Small were ordinated by Principal Component Analysis (PCA) using the 'vegan' package (Oksanen *et al.* 2010). In ordination, pollen percentages were calculated using only pollen types from each site. Taxa < 1% in at least one sample were excluded (Flakne 2003). To decipher the changes in the main plant populations, pollen percentages were standardised to allow a comparison of the magnitude of changes independent of the pollen productivity differences among taxa and the taphonomical properties of each taxon (Fuller 1998). Each taxon is displayed individually at a scale that is the same for all, allowing an estimate to be made of the magnitude of pollen population changes within and among pollen taxa (Carcaillet *et al.* 2010). The rescaled pollen percentages were based on a min-max transformation:

$$P_i' = \frac{P_i - P_{min}}{P_{max} - P_{min}}$$
(4-4)

Where P'_i is the standardised value of the *i*th sample P_i , and P_{min} and P_{max} are the minimum and maximum of the P'_i .

To assess the relationship between individual pollen taxa and fire, Spearman's rank correlation coefficient (Spearman's ρ) was calculated for mFI and pollen percentages of the dominant tree and shrub taxa at Lake Ben and Lake Small throughout the complete reconstruction. Charcoal data were interpolated to the age/depths per sample used in pollen

analysis using the package 'rioja' (Juggins 2014). mFIs were also calculated for each statistically significant pollen zone as determined by CONISS for each site.

Results

Chronologies

The age-depth models (Fig. 4.2a, b) represent approximately 10,375 and 9975 years of sedimentation at the Lake Ben and Lake Small sites, respectively. Sites are comparable in sedimentation rate with Lake Ben having a higher mean sedimentation rate at 0.086 ± 0.06 cm yr⁻¹ to Lake Small's 0.063 ± 0.05 cm yr⁻¹. Each sequence exhibits acceleration in sedimentation rate between *c*. 5000 - 3500 cal. yr BP, with the maximum deposition occurring at 4,310 cal. yr BP (0.029 cm yr⁻¹) in Lake Ben and 3785 cal. yr BP (0.033 cm yr⁻¹) in Lake Small. Due to the increased sedimentation rate, our fire and vegetation records for the time period *c*. 5,000 – 3,500 cal. yr BP have high temporal resolutions.



Figure 4.2 Age-depth models for (a) Lake Ben and (b) Lake Small

Fire reconstruction

The trends in the new reconstructions of median fire intervals (Fig. 4.3c, d) are not significantly different from previous reconstructions using a shorter temporal resolution and smoothing window (Chapter 2) (Senici *et al.* 2013). Forty-six fires were detected at Lake Ben throughout the Holocene, with fire intervals ranging from 50 to 750 years (Fig. 4.3c). The highest fire frequencies occurred between 2,000 cal. yr BP and the present and the lowest fire frequencies occurred 6,000 – 5,000 cal. yr BP. Global mFI for the reconstruction was 150 ± 24 years (mean \pm SE). Thirty-six fires were identified at Lake Small between 9,800 cal. yr BP and the present, with fire intervals ranging from 50 to 850 years (Fig. 4.3d). The highest fire frequencies occurred between 3,000 and 1,500 cal. yr BP and the lowest fire frequencies occurred 6,500 – 4,500 cal. yr BP. Overall mFI at this site was approximately 213 ± 24 years.



Figure 4.3 Sedimentary charcoal and pollen records at Lake Ben and Lake Small plotted against time. (a, b) Charcoal accumulation rates; vertical gray bars are CHAR peaks, the heavy black line represents CHAR background. (c, d) Smoothed fire return intervals (FRIs); circles are identified fire events and corresponding fire interval. (e, f) Pollen influx. (g, h) Rate of change

(ROC). Three indices of vegetation change: (i, j) Shannon's index (*H*); (k, l) Simpson's index (*D*); and (m, n) rarefaction analysis ($E(T_n)$); the histograms show the mean values of *D*, *H* and ($E(T_n)$) calculated per millennia; the letters refer to differences based on the Kruskal-Wallis test (different letters mean significant differences). The solid grey vertical lines indicate statistically significant pollen zones derived from CONISS at Lake Ben and at Lake Small.

Pollen influx and diversity

Total pollen influx was low at both sites following deglaciation to c. 9,000 cal. yr BP at Lake Ben (Fig. 4.3e) and c. 8,500 cal. yr BP at Lake Small (Fig. 4.3f). Overall pollen influx was higher at Lake Ben throughout the complete sequence. The ROC is based on pollen count data for all identified terrestrial pollen taxa. The first elevated ROC at Lake Ben and Small corresponds to the post-glacial afforestation phase and can be dated at 9,750 and 8,220 years BP respectively (Fig. 4.3g, h). Lake Ben exhibits fluctuations between 7,000 and 4,000 cal. yr BP likely resulting from vegetation transformations following afforestation; after 4,000 cal. yr BP ROC remains low. At Lake Small, the ROC is very low throughout most of the sequence save two peaks, the first c. 8,250 cal. yr BP and the second c. 4,000 cal. yr BP. This suggests that while very little vegetation transformations have occurred there were short time periods where vegetation transformed rapidly into a new state after which vegetation remained stable. However the periods of high ROC are at least partially due to the shorter intervals (high sample resolution) between pollen samples, which in turn reflects the high sediment accumulation rates in both chronologies. The diversity indices (Fig. 4.3i-l) and the rarefaction analysis (Fig. 4.3m, n) for both sites have remained relatively stable following the afforestation phase at each site.

Vegetation dynamics

Tree pollen abundance among our sites was greater than 60% throughout the entire profile (Fig. 4.4a, b), indicating that trees were a ubiquitous presence in the surrounding landscape following deglaciation. The pollen record of BEN was divided into five zones using CONISS (Fig. 4.4a). In zone B1 (c. 10,390 - 9,750 cal. yr BP), woodland cover values were high (80%) following deglaciation with pollen assemblages dominated by coniferous *Picea*. Other tree species like *L. laricina* and *T. occidentalis* are at their peak values (2 - 5%) in this zone. *Quercus* spp. and *Ulmus* spp. pollen are also at peak abundances (5%). Pollen from deciduous trees and shrub taxa was scarce, but herbaceous pollen was high, particularly among the Asteraceae (artemisia and ambrosia).



Figure 4.4 Simplified diagrams of pollen percentages at (a) Lake Ben and (b) Lake Small. The taxa include only those reaching mean pollen percentages $\geq 1\%$ in at least one sample. Dashed red lines separate the pollen zones.

The onset of zone B2 (c. 9,750 – 8,750 cal. yr BP) was marked by a rapid rise in *P*. *banksiana* accompanied with a concomitant decline in *Picea* and herbaceous taxa. The highest abundance of *P. tremuloides* (> 2%) and *P. strobus* (> 5%) occurs midway in this zone. *Abies* remained mostly low or absent throughout. *Quercus* spp. and *Ulmus* spp. pollen abundance remain relatively high (2 - 5%) but decline towards zone B3. Herbaceous Asteraceae and Chenopodiceae remained abundant (> 1%) throughout, however Cyperaceae was mostly absent as was Ericaceae. Except for a sharp increase in *A. viridis* near the zone's endpoint, shrub taxa was largely absent.

In zone B3 (c. 8,750 – 6,350 cal. yr BP), *B. papyrifera* pollen increases sharply from 5% to > 20% as does *A. viridis*. *P. banksiana* abundance remains high (18 – 20%) and is generally stable in this zone. While *Picea* pollen is abundant, it is least abundant (25%) in this zone. The continuous presence of *P. strobus* pollen indicates its apparent persistence in the regional landscape. The presence of *A. balsamea* is intermittent and scarce (< 1%). Abundance of *P. tremuloides* decreases and is mainly absent throughout this zone. Abundance of *L. laricina* decreases in this zone while *T. occidentalis* is mostly absent. *Quercus* spp. and *Ulmus* spp. pollen continue to decline in this zone but remain a relatively continuous presence. Asteraceae, Chenopodiaceae and Ericaceae are abundant throughout (>2%), however overall herbaceous pollen declines with a concomitant increase in shrub pollen.

Zone B4 (c. 6,350 - 2,500 cal. yr BP), is marked by increased *Picea* (30 - 40%) and decreased herbaceous taxa. *Pinus* cf. *banksiana* pollen abundance steadily declines, while *P. strobus* remains stable (2 - 5%). *T. occidentalis* reappears *c.* 5,000 cal. yr BP after a 3,500 year absence. *L. laricina* persists with discontinuous low (< 1%) abundances. Mainly discontinuous *P. tremuloides* abundances are recorded after 6,350 cal. yr BP. This zone marks the first

appearance of shrub *A*. *incana* (subsp. *rugosa*) *c*. 5,500 cal. yr BP. The increased abundance of *A*. *balsamea* and *L*. *laricina* pollen, indicate more diverse mixed forests. *Quercus* spp. and *Ulmus* spp. pollen continues to decline and become discontinuous. *A*. *viridis* remains the dominant shrub (10 - 20%), however abundance is lower compared to prior zone B3. Salix and *A*. *spicatum* are intermittently present (<2%). Herbaceous plant taxa are least abundant and predominantly discontinuous in this zone, with Chenopodiaceae becoming rare after 4,000 cal. yr BP.

In zone B5 (2,500 – 0 cal. yr BP), *A. balsamea* and *B. papyrifera* pollen increase to their highest levels of abundance (2% and 20% respectively) in the BEN record. *Picea* abundance remains similar to zone B4. *P. banksiana* and *P. strobus* decline to levels similar to the immediate post-glacial zone B1. *L. laricina* remains a constant low-level presence. Overall shrub pollen abundance increases, due mainly to an increase in *A. incana*. Overall understory herbaceous taxa remain low.

At SML between 9,800 and 8,100 cal. yr BP (zone S1, Fig. 4.4b), woodland were dominated by coniferous vegetation including *Picea* (40%), *P. strobus* (7%) *A. balsamea* (5%), *T. occidentalis* (2 - 5%) and *L. laricina* (2 - 5%). *Pinus* cf. *banksiana* percentages were at their lowest levels (< 3%). Deciduous species abundance is low, with *B. papyrifera* abundance at the lowest recorded values at this site (10 - 20%) and *Populus* cf. *tremuloides* absent until the onset of zone S2. *Quercus* spp. and *Ulmus* spp. pollen are abundant (5%) at the start of zone S1 but decline over time (< 1%). Overall, shrub pollen is scarce; *A. viridis* abundance increases sharply midway through the zone (*c.* 8,750 cal. yr BP), and *Salix* reaches maximum recorded abundance (3%). This zone is characterized by high herbaceous pollen frequencies, in particular Cyperaceae, Artemisia and Ambrosia were abundant (> 5%). There was overall low pollen accumulation rates for all species.

Around 8,100 cal. yr BP (onset of Zone S2), a sharp rise in *B. papyrifera* and *A. viridis* abundance is associated with lower relative abundances of coniferous *Picea*, *A. balsamea*, and *L. laricina*. While abundant, *Picea* percentages were at their lowest values (30%) in this zone. *P. cf. banksiana* pollen progressively increases while *P. strobus* abundance remains relatively constant (5%). *L. laricina* and *T. occidentalis* abundances were low throughout, never exceeding 1% and disappearing from the fossil record entirely for > 1,000 years towards the end of the zone. The presence of *P. tremuloides, Quercus* spp., and *Ulmus* spp. pollen indicates a more diverse and mixed forest composition with higher abundances of deciduous vegetation and shrubs. Cyperaceae, Artemisia and Ambrosia decline from their high zone S1 levels with a concomitant increase in shrub pollen.

In Zone S3 (c. 4,250 cal. yr BP to present), P. cf. banksiana percentages are at their highest values, the increase being primarily at the expense of A. viridis and herbaceous plants. *Picea* abundance remains similar to Zone S2. Abundance of L. laricina increases. Pollen abundance of A. incana increased along with an accompanying decrease in herb pollen (mainly Chenopodiaceae and Ericaceae). A. viridis pollen is at lower levels throughout the zone but increased sharply in the last 300 years. A. incana increases to peak abundances for this species in this zone. All herbaceous taxa, save Artemisia and Ambrosia display low abundance with discontinuous appearances.

Fire and vegetation interactions

The first two axes of each PCA explain a significant amount of variation, accounting for 38% and 50% of the total variance of the Lake Ben and Small datasets (Fig. 4.5a-d, Table 4.2). respectively. Only the first axes of each PCA account for large amounts of variation for Lake Ben (22.4%) and Lake Small (36.2%) (Fig. 4.5a, b, Table 4.2). The first axis, shows a strong contrast between the initial vegetation assemblages following the retreat of the Laurentide Ice Sheet and proglacial waters in the late Holocene between sites (Fig. 4.5a, b). The gradual transition observed at Lake Ben (Fig. 4.5a), suggests a long-term and gradual process dominated the vegetation dynamics at this site; while the abrupt change at Lake Small followed by relative stability suggest rapid afforestation in a short time span that occurred later than afforestation at Lake Ben, which was followed by relatively stable states to present. The second PCA axis accounts for $\sim 14 - 15\%$ (Table 4.2) of the variation at both sites; at Lake Ben following afforestation phase, the axis is relatively stable to present. At Lake Small this axis is more variable with change points roughly corresponding to CONISS identified pollen zone change points (c. 8,100 and 4,250 cal. yr BP). The eigenvalues for the third and greater PCA axes are less than the mean eigenvalues for each PCA (Table 4.2), suggesting these axes do not account for a significant amount of variability (Legendre and Legendre 1998).

	Lake	Lake Ben		Lake Small		
Axes	%	Eigenvalue	%	Eigenvalue		
1	22.4	0.11	36.2	0.24		
2	15.3	0.07	13.8	0.09		
3	10.9	0.05	10.7	0.07		
4	9.2	0.04	7.7	0.05		
5	7.4	0.04	6.2	0.04		
6	6.9	0.03	5.3	0.03		

Table 4.2 Variance (%) explained and eigenvalues of the two PCAs.



Figure 4.5 Axis scores of the principal component analyses (PCA) based on pollen data plotted against time for Lake Ben (a, c) and Lake Small (b, d). Ordination biplot of pollen taxa on PCA axes 1 and 2 at (e) Lake Ben, and (f) Lake Small. Red circles represent site scores for pollen zone 1, blue inverse triangles for zone 2, and orange squares for zone 3. In (a) yellow diamonds represent zone B4 and upright grey triangles zone B5.

The main long-term vegetation trajectories in relation to fire interval at Lake Ben (Fig. 4.5e) are characterized by *Picea*, *Quercus* spp., *Ulmus* spp, and *T. occidentalis* (zone B1) in the afforestation phase and are most dominant in the absence of fire (Table 4.3). As fire begins to recur regularly (mFI 190 years; Table 4.3) we find increasing *P. strobus* and *P. banksiana* in zone B2. Following the afforestation phase, mixed forest assemblages dominate in zone B3 where fire frequency is at a record low; in this zone increased *B. papyrifera* and shrubs *A. viridis*

and *A. incana* are significant contributors to local pollen assemblages. Zone 4 has the most diverse mixed forest while mFIs decrease from zone B3. Zone B5 has the highest fire frequency (mFI 119 years) and is predominantly mixed forest with dominated by *Picea*, and *B. papyrifera*, *A. balsamea*, and shrubby vegetation. The main long-term vegetation trajectories at Lake Small, as summarized by the first axis (Fig 4.5f), are characterized by a shift in coniferous dominated assemblages of *Picea*, *L. laricina*, and *A. balsamea* (zone S1) where fire frequency is low, to mixed assemblages with increasing abundances of *B. papyrifera*, *A. viridis*, and other deciduous shrub cover (zone S2), to mixed assemblages dominated by *P. cf. banksiana*, *P. tremuloides*, and *A. incana* in zone S3 where fire frequency is at the highest recorded levels. The second PCA axis at Lake Small is mainly related to changes in *P. cf. banksiana*, *P. tremuloides*, and other shrub vegetation. Kruskal-Wallis tests were used to assess differences in mFIs between adjacent pollen zones; there were no significant differences in mFIs at each site as indicated by the overlapping confidence intervals.

Table 4.3 Median fire return intervals (mFIs) and number of identified fire events (N_{fe}) for each site for pollen zones identified by CONISS.

· · · · · · · · · · · · · · · · · · ·	Lake Ben		Lake Small	
Pollen Zone	mFI	N_{fe}	mFI	N_{fe}
1	-	1	290 (221 - 359)	4
2	190 (135 - 255)	5	269 (200 - 342)	13
3	318 (150 - 489)	7	197 (137 - 284)	19
4	199 (142 - 256)	19		
5	119 (90 - 148)	13		

Spearman's correlations between species and mFI reveal different patterns at Lake Ben and Lake Small. At Ben, among woody taxa, *A. balsamea, B. papyrifera*, and *P. tremuloides* abundances are significantly negatively correlated with mFI as is shrub *A. incana* (Table 4.4). Significant positive correlations were found among *P* cf. *banksiana* and *P. strobus*. Other species

Table 4.4 Correlation analyses (Spearman's rho) between the mFI and pollen percentages	s for
Lake Ben and Lake Small.	

	La	ke Ben	Lak	e Small
Taxon	rho	p-value	rho	p-value
Abies balsamea	-0.36	< 0.001	-0.21	0.060
Picea sp.	-0.15	0.059	0.02	0.889
Pinus cf. banksiana	0.38	< 0.001	-0.42	< 0.001
Pinus strobus	0.23	0.003	0.21	0.063
Larix larixina Thuia	-0.13	0.083	-0.12	< 0.001
occidentalis/Juniperus	-0.04	0.619	0.11	0.313
Betula papyrifera	-0.17	0.029	-0.06	0.624
Populus cf. tremuloides	-0.27	< 0.001	-0.28	0.013
Quercus spp.	-0.08	0.296	0.50	< 0.001
Ulmus spp.	0.02	0.803	0.53	< 0.001
Alnus viridis	-0.15	0.056	0.24	0.032
Alnus incana	-0.33	< 0.001	-0.57	< 0.001
Salix spp.	-0.03	0.730	0.41	0.000
Acer spicatum	-0.05	0.537	-0.15	0.180

Discussion

Holocene fire dynamics

Reconstructed local fire events show that the trend in local fire frequency changed over time (Fig. 4.3c, d) with an overall increase from deglaciation to the present. Our results suggest that while individual species were affected by fires, overall regional compositions were only somewhat affected. The following discussion examines the long-term consequences of these

processes on central boreal vegetation, i.e., species-specific reactions to mFI and the community assemblages through time

Early post-glacial vegetation history

Based on the estimated ages and sedimentation rates of the lower most ¹⁴C at both sites, it appears that gyttia formation started rapidly following the retreat of the Laurentide Ice Sheet. At Lake Ben, the forested landscape potentially burned as early as c. 10,265 cal. yr BP and at Lake Small the first fire event occurred c. 9,690 cal. yr BP. The rapid afforestation by *Picea* at Lake Ben agrees with other studies in central North American boreal forest (Ritchie and Yarranton 1978; Genries et al. 2012) showing that moisture-tolerant trees were present near the front edge of the retreating ice sheet. However, the low pollen influx (< 10,000 grains cm⁻² yr⁻¹) and low C_{background} suggest that the *Picea* and Herb dominated forests of the early Holocene was not as dense as the mid- and late-Holocene forests (Fig. 3a, b, e, f; Fig 4a, b). Cbackground, an indicator of biomass burning in the local terrestrial landscape (Marlon et al. 2006) was lowest during this afforestation period, and the relatively high abundance of herbaceous species indicates a more open environment and thus biomass was probably too low to support large fires that would be preserved in sedimentary charcoal records (Tinner et al. 1996). Following deglaciation, local fire activity was negligible in zone B1 with the onset of zone B2 marking significant increases in fire activity (Fig. 3c). At Lake Small low pollen influx (< 10,000 grains $\text{cm}^{-2} \text{ yr}^{-1}$) persists until c. 8,500 cal. yr BP, coinciding with the first peak in ROC (c. 8,220) which roughly corresponds to the end of pollen zone S1. This seems to indicate that the forest landscape at Lake Small was sparse for much longer when compared to Lake Ben. Low average pollen influx at Lake Small may also be a reflection of lower terrestrial cover and thus vegetation in the surrounding

landscape, as Lake Small is encircled by abundant open-water lakes and wetlands. This delayed afforestation may account for the longer regional mFIs in the early Holocene.

Transition and long-term vegetation dynamics

The beginning of zone B2 corresponds to a densification of the forest cover, as indicated by a sharp increase in overall pollen influx (Fig. 4.3e) and pollen percentages of *P*. cf. *banksiana* specifically (Fig. 4.4b). Vegetation dynamics at both lakes have followed similar trajectories after *c*. 8,750 cal yr BP, which is consistent with vegetation dynamics in nearby northeastern Ontario (Genries *et al.* 2012) and other sites farther east (Liu 1990; Carcaillet *et al.* 2010). However there was one notable difference between sites, where at Lake Ben, *P. cf. banksiana* abundance gradually declines following afforestation, the opposite is true at Lake Small where abundance gradually increases to the present. Overall, the main long-term vegetation trajectories identify transitions from *Picea* dominated forests to mixed forests, where *P. banksiana*, *B. papyrifera*, *P. tremuloides*, and *A. viridis* became predominant. Although overall fire frequency fluctuated throughout the Holocene (Fig 4.3c, d) and between pollen zones (Table 4.3), vegetation has changed slightly, much less than might be expected on the basis of observed local fire history, which may indicate that regional compositions are resilient to changes in fire interval (Carcaillet *et al.* 2010).

Fire-climate-vegetation interactions

Because *Abies balsamea* is a late successional tree and the least fire-resistant conifer in the North American boreal forest (Frank 1990), it is counterintuitive that *A. balsamea* pollen abundance was highest when fire was most frequent at Lake Ben (Fig. 4a, zone B5) and marginally significant (P= 0.06) at Lake Small. Despite its relatively low pollen abundance (Fig. 4a, b), *A. balsamea* was likely locally present because it has low pollen productivity rates and short

dispersal distances (1 - 2 km) (Richard 1976; MacInnis 2012); as a result, the elevated abundances were likely inputs from communities near lakeshores or on lake islands where the species could avoid lethal regional fires and continue to contribute to the fossil pollen record (Bergeron *et al.* 2004).

The *Picea* taxon was not found to be significantly correlated with mFI at either site. While *P. mariana* is often a postfire pioneer on both uplands and peatlands and fire typically results in the immediate regeneration of the species provided an extant local seed source, both P. glauca and P. mariana are shade-tolerant and can be abundant in late-successional stands (Taylor and Chen 2011). The immediate post-glacial afforestation by this species and ubiquitous presence thereafter (20 - 40%, Fig. 4a, b), attests to the resilience of the taxon to fire frequency variations. The regional successional role of this taxon is somewhat ambiguous, as its life-history characteristics can make it suited as both a pioneer and late successional species (Payette 1993; Taylor and Chen 2011); this also highlights a limitation in using a single taxon to represent multiple species (Birks and Line 1992; Blarquez et al. 2014). Several studies show that increased regional burning in the Holocene, particularly during the Neoglacial period (c. 4,500 - 2,500 cal. yr BP) can be linked to an increase in the growth of flammable conifer species like *Picea* (Carcaillet et al. 2001; Lynch et al. 2002; Higuera et al. 2009); no significant increases in Picea abundance were detected at our sites. However, shorter mFIs in the Neoglacial period were associated with increased abundance of P. cf. banksiana at Lake Small in zone S3 (4,250 - 0 cal. yr BP). It is possible that increased fire frequency favored the growth and development of mixed P. banksiana stands (Fig. 5f) with likely positive feedbacks to the local fire regime.

At Lake Ben, high abundance of *P*. cf. *banksiana* is associated with longer mFIs, this is noteworthy particularly in zones B4 and B5 where mFIs are shortest and *P*. cf. *banksiana* is least

abundant (6,350 to 0 cal. yr BP). This time period corresponds to the end of the Holocene Climatic Optimum (HCO; *c*. 6,000 yr BP) and continues through the Neoglacial period; since the PCA (Fig. 5e) indicates that the decline in *P*. cf. *banksiana* abundance is associated mainly with increased *B. papyrifera* and *Picea* abundances. Where *B. papyrifera* and *Picea* can grow on a wide variety of soil moistures, optimal growth of *P. banksiana* is limited to very or moderately dry soil moisture regimes (Rudolph and Laidly 1990); as a result the decline in *P. banksiana* may be due to increased regeneration competition post-fire from moisture-tolerant pioneers under cooler and moister regional climatic conditions.

At Lake Ben, *P. strobus* pollen abundance was found to be higher when mFIs were longer; at Lake Small this correlation was only marginally significant (P= 0.06). However, because *P. strobus* pollen is abundantly produced and aerially transported over long distances, there is some doubt as to its true local abundance, particularly due the species continuous low level abundance in both vegetation records, relative insensitivity to changes in local fire frequency at Lake Small, and that the study area is at the northern limit of the modern natural range of the species.

At Lake Small, *L. laricina* is most abundant when mFIs are short (zone S3, Fig. 4b). These results are consistent with local macroremains presented in Senici *et al* (2013), and may indicate that increased fire frequency favoured *L. laricina* in the regional mesic landscape. At both sites, *P. tremuloides* was most abundant when mFIs were short. While *P.* cf. *tremuloides* is the most abundant deciduous species in the central boreal forest, *Populus* pollen is typically underrepresented in palynological reconstructions because it is more susceptible to corrosion than are most other pollen types (Campbell 1999). Abundant *Populus* pollen in our reconstructions may indicate that either the *Populus* pollen preserved very well in our sedimentary deposits, or that the taxon may have been more abundant in the regional landscape than our pollen stratigraphies suggest.

Prolific growth of clonal shrubs can be a major obstacle in conifer regeneration, due to the speed in which they regenerate by sprouting from basal or underground stems following fire (Mallik *et al.* 1997; Titus and Bishop 2014). In the PCA from Lake Small (Fig. 4.5f), *A. viridis* is in direct opposition to multiple conifers including *Picea*, *P. strobus*, *A. balsamea*, and *L. laricina* while at Lake Ben the shrub is most frequent at the expense to *P. strobus*, *Quercus* spp., *Ulmus* spp. and Herbaceous plants. The correlation analysis indicates higher abundances at Lake Small when mFIs are long which occurs primarily in the mid-Holocene within zone S2 (Fig. 4.4b); no significant correlations were found at Lake Ben. However, *A. viridis* remains one of the most dominant taxa at both sites following afforestation indicating the species is very adaptable and successful regardless of length of mFIs. *A. incana* is significantly negatively correlated to mFI at both site, indicating high local abundance when mFIs are short.

Conclusions

Delayed afforestation at Lake Small may account for the longer mFIs detected at this site in the early Holocene. Overall vegetation composition trajectories move from *Picea* dominated coniferous stands to mixed compositions including *Pinus banksiana*, *Betula papyrifera*, and *Alnus viridis*. While overall communities appeared insensitive to mFI variations, individual taxons exhibit varied responses between sites indicating potential interactions with local environmental factors, interspecific competition and climate. Additional high-resolution analyses of vegetation variability may reveal more information regarding fire-climate-vegetation interactions and would be beneficial to land management and conservation planning in the region.

Acknowledgements

We thank Loic Bircker, Etienne DuBerger, and Raynald Juliens for their assistance in data collection. Élisabeth Robert and Martin Lavoie at for their help and advice with regard to pollen analysis. Financial support was provided to Han Chen by the Natural Sciences and Engineering Research Council of Canada (NSERC DG283336-09), Ontario Early Researcher Award, and Northern Ontario Heritage Foundation Corporation, Yves Bergeron by NSERC and to Adam Ali by the CNRS (BOREOFIRE, PALEO2, INSU). The lead author acknowledges NSERC, Forest Ecosystem Science Co-operative Inc., and Resolute Forest Products for an Industrial Postgraduate Scholarship.

CHAPTER FIVE Future Work and Conclusion

Future work in the region should focus the development of additional long-term paleoecological records, and the incorporation of the data into broad-scale syntheses. The development of additional charcoal and pollen records in the central boreal shield region would contribute to a more complete understanding of the variability of climate-vegetation-fire interactions across North America. When compared to eastern and western boreal forests there is a relative dearth of paleofire and vegetation records in the vast central region. New records should be developed to provide greater coverage across the various topographic and vegetation gradients in the region. To this end, increased paleoecological reconstructions in the region should focus towards northern and north-western latitudes, where the projected impacts of climate warming are likely to be the greatest (Field et al. 2007). Additionally, new sedimentary records could be analysed for multiple proxies using various paleoecological techniques not considered in this dissertation, including but not limited to physical sedimentology, isotope geochemistry, carbon-nitrogen ratios, planktonic assemblages, insects (chironomids, coleopterans) and diatoms, all of which could provide insight into past environmental states and terrestrial-aquatic interactions allowing examinations of how the local ecosystems respond to a wide array of environmental changes. This research could generate useful data and knowledge about the variability of disturbance regimes and interactions between various types of disturbance.

Future work on the unpublished chapters presented in this dissertation should also focus on 2 main areas: 1) the incorporation of pollen data into chapter 3; and 2) the use of long-term paleoclimate data and improved rigor of the main statistical analyses in chapter 4. The third chapter could be significantly improved by the addition of pollen data from chapter four, as it would help to clarify the discussion section and some of the proposed climate-vegetation

interactions that may have influenced the fire regimes at poorly and highly connected sites. Currently, the fourth chapter is largely an exploratory data analysis approach to analyzing the multiple pollen data sets and represents the author's first attempt at understanding pollen records and performing various pollen analyses. The work could be improved by using more rigorous statistical analyses such as a permutational Multivariate Analysis of Variance (formerly 'nonparametric' MANOVA) on the pollen data sets using paleoclimatic (temperature and precipitation) and fire regime (CHAR, C_{background}, and FRI) variables to examine fire-climate-vegetation interactions more explicitly and partition the variance throughout the Holocene.

General Conclusion

The development of multiple Holocene records of forest fire has allowed for the analysis of the relative importance of various spatial controls of fire as well as how some of those controls have varied over time. The results from this study highlight the importance of considering localenvironmental factors when interpreting paleoecological records of fire and vegetation. By quantifying how fire regimes have responded to past environmental change, the information in this dissertation can be used to better construct models of how forest vegetation and fire regimes may respond to ongoing global climate change. The most important ecological insights provided by this dissertation is also one of the most obvious, that change is typical, fluctuations and variability in fire activity and ecosystem response are normal and expected. Only by understanding the natural variability of wildfire, how and where it originates and its consequences to communities and ecosystems can we truly begin to develop research programs necessary to understand and protect the boreal forest in the future.
Literature Cited

- (2007) 'WinSeedle.' (Regent Instruments Incorporated: Quebec)
- Ali, AA, Asselin, H, Larouche, AC, Bergeron, Y, Carcaillet, C, Richard, PJH (2008) Changes in fire regime explain the Holocene rise and fall of Abies balsamea in the coniferous forests of western Quebec, Canada. *Holocene* 18, 693-703.
- Ali, AA, Blarquez, O, Girardin, MP, Hely, C, Tinquaut, F, El Guellab, A, Valsecchi, V, Terrier, A, Bremond, L, Genries, A, Gauthier, S, Bergeron, Y (2012) Control of the multimillennial wildfire size in boreal North America by spring climatic conditions. *Proceedings of the National Academy of Sciences of the United States of America* 109, 20966-20970.
- Ali, AA, Carcaillet, C, Bergeron, Y (2009a) Long-term fire frequency variability in the eastern Canadian boreal forest: the influences of climate vs. local factors. *Global Change Biology* 15, 1230-1241.
- Ali, AA, Higuera, PE, Bergeron, Y, Carcaillet, C (2009b) Comparing fire-history interpretations based on area, number and estimated volume of macroscopic charcoal in lake sediments. *Quaternary Research* **72**, 462-468.
- Baldwin, KA, MacKenzie, WH, Pfalz, J, Meades, WJ, Meidinger, DV, Robitaille, A, Saucier, J-P, Uhlig, P (2012) Level 4 map, version 1, Canadian component of the Circumboreal Vegetation Map (CBVM). Canadian regional team of the Circumboreal Vegetation Map Project. Natural Resources Canada, Canadian Forest Service, Great Lakes Forestry Centre, Sault Ste. Marie, ON. Unpublished map.
- Barrett, CM, Kelly, R, Higuera, PE, Hu, FS (2013) Climatic and land cover influences on the spatiotemporal dynamics of Holocene boreal fire regimes. *Ecology* **94**, 389-402.
- Beck, PSA, Goetz, SJ, Mack, MC, Alexander, HD, Jin, YF, Randerson, JT, Loranty, MM (2011) The impacts and implications of an intensifying fire regime on Alaskan boreal forest composition and albedo. *Global Change Biology* 17, 2853-2866.
- Bennett, KD (1996) Determination of the number of zones in a biostratigraphical sequence. *New Phytologist* **132**, 155-170.
- Bergeron, Y (2000) Species and stand dynamics in the mixed woods of Quebec's southern boreal forest. *Ecology* **81**, 1500-1516.
- Bergeron, Y, Gauthier, S, Flannigan, M, Kafka, V (2004) Fire regimes at the transition between mixedwood and coniferous boreal forest in Northwestern Quebec. *Ecology* 85, 1916-1932.
- Bessie, WC, Johnson, EA (1995) The Relative Importance of Fuels and Weather on Fire Behavior in Sub-Alpine Forests. *Ecology* **76**, 747-762.
- Birks, HH, Birks, HJB (2000) Future uses of pollen analysis must include plant macrofossils. Journal of Biogeography 27, 31-35.
- Birks, HH, John, B, 2008. Paleoecology. Encyclopedia of ecology. Elsevier, Amsterdam. 2623-2634.
- Birks, HJB, Line, JM (1992) The use of Rarefaction Analysis for Estimating Palynological Richness from Quaternary Pollen-Analytical Data. *The Holocene* **2**, 1-10.
- Blarquez, O, Bremond, L, Carcaillet, C (2010) Holocene fires and a herb-dominated understorey track wetter climates in subalpine forests. *Journal of Ecology* **98**, 1358-1368.
- Blarquez, O, Carcaillet, C (2010) Fire, fuel composition and resilience threshold in subalpine ecosystem. *Plos One* **5**, e12480.

- Blarquez, O, Carcaillet, C, Frejaville, T, Bergeron, Y (2014) Disentangling the trajectories of alpha, beta and gamma plant diversity of North American boreal ecoregions since 15,500 years. *Frontiers in Ecology and Evolution* **2**,
- Blarquez, O, Girardin, MP, Leys, B, Ali, AA, Aleman, JC, Bergeron, Y, Carcaillet, C (2013) Paleofire reconstruction based on an ensemble-member strategy applied to sedimentary charcoal. *Geophysical Research Letters* **40**, 2667-2672.
- Bonan, GB, Shugart, HH (1989) Environmental-Factors and Ecological Processes in Boreal Forests. *Annual Review of Ecology and Systematics* **20**, 1-28.
- Bond-Lamberty, B, Peckham, SD, Ahl, DE, Gower, ST (2007) Fire as the dominant driver of central Canadian boreal forest carbon balance. *Nature* **450**, 89.
- Bond, WJ, Keeley, JE (2005) Fire as a global 'herbivore': the ecology and evolution of flammable ecosystems. *Trends in Ecology & Evolution* **20**, 387-394.
- Bouchard, M, Pothier, D, Gauthier, S (2008) Fire return intervals and tree species succession in the North Shore region of eastern Quebec. *Canadian Journal of Forest Research* **38**, 1621-1633.
- Bowman, DMJS, Balch, JK, Artaxo, P, Bond, WJ, Carlson, JM, Cochrane, MA, D'Antonio, CM, DeFries, RS, Doyle, JC, Harrison, SP, Johnston, FH, Keeley, JE, Krawchuk, MA, Kull, CA, Marston, JB, Moritz, MA, Prentice, IC, Roos, CI, Scott, AC, Swetnam, TW, van der Werf, GR, Pyne, SJ (2009) Fire in the Earth System. *Science* **324**, 481-484.
- Brassard, BW, Chen, HYH (2006) Stand structural dynamics of North American boreal forests. *Critical Reviews in Plant Sciences* **25**, 115-137.
- Brassard, BW, Chen, HYH, Wang, JR, Duinker, PN (2008) Effects of time since stand-replacing fire and overstory composition on live-tree structural diversity in the boreal forest of central Canada. *Canadian Journal of Forest Research* **38**, 52-62.
- Briles, CE, Whitlock, C, Bartlein, PJ, Higuera, P (2008) Regional and local controls on postglacial vegetation and fire in the Siskiyou Mountains, northern California, USA. *Palaeogeography Palaeoclimatology Palaeoecology* **265**, 159-169.
- Brossier, B, Oris, F, Finsinger, W, Asselin, H, Bergeron, Y, Ali, AA (2014) Using tree-ring records to calibrate peak detection in fire reconstructions based on sedimentary charcoal records. *The Holocene* 24, 635-645.
- Brubaker, LB (1975) Postglacial forest patterns associated with till and outwash in North-central upper Michigan. *Quaternary Research* **5**, 499-527.
- Brubaker, LB, Higuera, PE, Rupp, TS, Olson, MA, Anderson, PM, Hu, FS (2009) Linking sediment-charcoal records and ecological modeling to understand causes of fire-regime change in boreal forests. *Ecology* **90**, 1788-1801.
- Burnham, KP, Anderson, DR (2002) 'Model selection and multi-model inference: a practical information-theoretic approach.' (2nd edn. Springer Berlag, New York.:
- Burton, PJ, Parisien, MA, Hicke, JA, Hall, RJ, Freeburn, JT (2008) Large fires as agents of ecological diversity in the North American boreal forest. *International Journal of Wildland Fire* **17**, 754-767.
- Campbell, ID (1999) Quaternary pollen taphonomy: examples of differential redeposition and differential preservation. *Palaeogeography, Palaeoclimatology, Palaeoecology* **149**, 245-256.
- Cappers, RTJ, Bekker, RM, Jans, JEA (2006) Digitale zadenatlas van Nederland. Barkhuis Publishing). [Accessed

- Carcaillet, C, Bergeron, Y, Richard, PJH, Frechette, B, Gauthier, S, Prairie, YT (2001) Change of fire frequency in the eastern Canadian boreal forests during the Holocene: does vegetation composition or climate trigger the fire regime? *Journal of Ecology* **89**, 930-946.
- Carcaillet, C, Richard, PJH, Bergeron, Y, Frechette, B, Ali, AA (2010) Resilience of the boreal forest in response to Holocene fire-frequency changes assessed by pollen diversity and population dynamics. *International Journal of Wildland Fire* **19**, 1026-1039.
- Chen, HYH, Popadiouk, RV (2002) Dynamics of North American boreal mixedwoods. Environmental Reviews 10, 137-166.
- Chen, HYH, Taylor, AR (2012) A test of ecological succession hypotheses using 55-year timeseries data for 361 boreal forest stands. *Global Ecology and Biogeography* **21**, 441-454.
- Chen, HYH, Vasiliauskas, S, Kayahara, GJ, Ilisson, T (2009) Wildfire promotes broadleaves and species mixture in boreal forest. *Forest Ecology and Management* **257**, 343-350.
- Clark, JS (1988) Particle Motion and the Theory of Charcoal Analysis Source Area, Transport, Deposition, and Sampling. *Quaternary Research* **30**, 67-80.
- Clark, JS (1989) Ecological Disturbance As A Renewal Process Theory and Application to Fire History. *Oikos* 56, 17-30.
- Clark, JS, Royall, PD, Chumbley, C (1996) The role of fire during climate change in an eastern deciduous forest at Devil's Bathtub, New York. *Ecology* 77, 2148-2166.
- Clements, FE (1916) 'Plant succession; an analysis of the development of vegetation.' (Carnegie Institution of Washington: Washington)
- Clemmensen, KE, Bahr, A, Ovaskainen, O, Dahlberg, A, Ekblad, A, Wallander, H, Stenlid, J, Finlay, RD, Wardle, DA, Lindahl, BD (2013) Roots and Associated Fungi Drive Long-Term Carbon Sequestration in Boreal Forest. *Science* **339**, 1615-1618.
- Colombo, SJ, McKenney, DW, Lawrence, KM, Gray, PA (2007) Climate Change Projections for Ontario: Practical Information for Policymakers and Planners. Ontario Ministry of Natural Resources, Applied Research and Development Branch. Climate Change Research Report., Sault Ste. Marie, ON.
- Connell, JH, Slatyer, RO (1977) Mechanisms of succession in natural communities and their role in community stability and organization. *American Naturalist* **111**, 1119 1144.
- Crins, WJ, Gray, PA, Uhlig, PWC, Wester, MC (2009) The Ecosystems of Ontario, Part 1: Ecozones and Ecoregions. Ontario Ministry of Natural Resources, Inventory, Monitoring and Assessment Section., Peterborough, Ontario.
- Cumming, SG (2001) Forest type and wildfire in the alberta boreal mixedwood: What do fires burn? *Ecological Applications* **11**, 97-110.
- Cyr, D, Bergeron, Y, Gauthier, S, Larouche, A (2005) Are the old-growth forests of the Clay Belt part of a fire-regulated mosaic? *Canadian Journal of Forest Research* **35**, 65-73.
- Cyr, D, Gauthier, S, Bergeron, Y (2007) Scale-dependent determinants of heterogeneity in fire frequency in a coniferous boreal forest of eastern Canada. *Landscape Ecology* 22, 1325-1339.
- Cyr, D, Gauthier, S, Bergeron, Y, Carcaillet, C (2009) Forest management is driving the eastern North American boreal forest outside its natural range of variability. *Frontiers in Ecology and the Environment* 7, 519-524.
- de Groot, WJ, Cantin, AS, Flannigan, MD, Soja, AJ, Gowman, LM, Newbery, A (2013a) A comparison of Canadian and Russian boreal forest fire regimes. *Forest Ecology and Management* **294**, 23-34.

- de Groot, WJ, Flannigan, MD, Cantin, AS (2013b) Climate change impacts on future boreal fire regimes. *Forest Ecology and Management* **294**, 35-44.
- Doss, H (1989) On estimating the dependence between two point processes. *Annals of Statistics* 17, 749-763.
- ESRI (2012) 'ArcGIS 10.1.' (Environmental Systems Research Institute: Redlands, CA)
- Faegri, K, Iversen, J (1989) 'Textbook of Pollen Analysis, fourth ed.' (Wiley: Chichester, UK)
- Fauria, MM, Johnson, EA (2006) Large-scale climatic patterns control large lightning fire occurrence in Canada and Alaska forest regions. *Journal of Geophysical Research-Biogeosciences* 111, 17.
- Fauria, MM, Johnson, EA (2008) Climate and wildfires in the North American boreal forest. Philosophical Transactions of the Royal Society B-Biological Sciences **363**, 2317-2329.
- Field, CB, Mortsch, LD, Brklacich, M, Forbes, DL, Kovacs, P, Patz, JA, Running, SW, Scott, MJ (2007) Climate Change 2007: Impacts, Adapatation and Vulnerability. Contribution of Working Group II to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change. (Eds ML Parry, OF Canziani, JP Palutikof, PJ van der Linden, CE Hanson.) pp. 617-652. (Cambridge University Press: Cambridge, UK)
- Flakne, R (2003) The Holocene vegetation history of Isle Royale National Park, Michigan, U.S.A. *Canadian Journal of Forest Research* **33**, 1144-1166.
- Flannigan, M, Cantin, AS, de Groot, WJ, Wotton, M, Newbery, A, Gowman, LM (2013) Global wildland fire season severity in the 21st century. *Forest Ecology and Management* **294**, 54-61.
- Flannigan, M, Stocks, B, Turetsky, M, Wotton, M (2009) Impacts of climate change on fire activity and fire management in the circumboreal forest. *Global Change Biology* 15, 549-560.
- Flannigan, MD, Logan, KA, Amiro, BD, Skinner, WR, Stocks, BJ (2005) Future area burned in Canada. *Climatic Change* 72, 1-16.
- Flannigan, MD, Van Wagner, CE (1991) Climate Change and Wildfire in Canada. *Canadian* Journal of Forest Research 21, 66-72.
- Frank, RM (1990) Abies balslamea (L.) Mill. Balsam Fir. In 'Silvics of North America. Vol 1. Conifers.' pp. 26-35. (U.S.D.A. Forest Service: Washington, D.C.)
- Fuller, JL (1998) Ecological impact of the mid-Holocene Hemlock decline in Southern Ontario, Canada. *Ecology* **79**, 2337-2351.
- Gavin, DG (2010) 'K1D: Multivariate Ripley's K-function for one dimensional data.' (University of Oregon, Department of Geography.:
- Gavin, DG, Brubaker, LB, Lertzman, KP (2003) An 1800-year record of the spatial and temporal distribution of fire from the west coast of Vancouver Island, Canada. *Canadian Journal of Forest Research* **33**, 573-586.
- Gavin, DG, Hallett, DJ, Hu, FS, Lertzman, KP, Prichard, SJ, Brown, KJ, Lynch, JA, Bartlein, P, Peterson, DL (2007) Forest fire and climate change in western North America: insights from sediment charcoal records. *Frontiers in Ecology and the Environment* 5, 499-506.
- Gavin, DG, Hu, FS, Lertzman, K, Corbett, P (2006) Weak climatic control of stand-scale fire history during the late Holocene. *Ecology* **87**, 1722-1732.
- Gedalof, Z (2011) Climate and spatial patterns of wildfire. In 'The landscape ecology of fire.' (Eds D McKenzie, C Miller, D Falk.) (Springer: Dordrecht)
- Gelman, A, Su, Y-S, Yajima, M, Hill, J, Granzia Pittau, M, Kerman, J, Zheng, T, Dorie, V (2013) 'arm: Data Analysis Using Regression and Multilevel/Hierarchical Models.'

- Genries, A, Finsinger, W, Asnong, H, Bergeron, Y, Carcaillet, C, Garneau, M, Hely, C, Ali, AA (2012) Local versus regional processes: can soil characteristics overcome climate and fire regimes by modifying vegetation trajectories? *Journal of Quaternary Science* 27, 745-756.
- Gillett, NP, Weaver, AJ, Zwiers, FW, Flannigan, MD (2004) Detecting the effect of climate change on Canadian forest fires. *Geophysical Research Letters* **31**,
- Girardin, MP, Ali, AA, Carcaillet, C, Blarquez, O, Hely, C, Terrier, A, Genries, A, Bergeron, Y (2013a) Vegetation limits the impact of a warm climate on boreal wildfires. *New Phytologist* **199**, 1001-1011.
- Girardin, MP, Ali, AA, Carcaillet, C, Gauthier, S, Hely, C, Le Goff, H, Terrier, A, Bergeron, Y (2013b) Fire in managed forests of eastern Canada: Risks and options. *Forest Ecology* and Management **294**, 238-249.
- Girardin, MP, Bergeron, Y, Tardif, JC, Gauthier, S, Flannigan, MD, Mudelsee, M (2006) A 229year dendroclimatic-inferred record of forest fire activity for the Boreal Shield of Canada. *International Journal of Wildland Fire* **15**, 375-388.

Girardin, MP, Wotton, BM (2009) Summer Moisture and Wildfire Risks across Canada. Journal of Applied Meteorology and Climatology 48, 517-533.

- Gleason, HA, 1926. The individualistic concept of the plant association. The Bulletin of the Torrey Botanical Club.
- Glew, JR (1989) A new trigger mechanism for sediment samplers. *Journal of Paleolimnology* **2**, 241-243.
- Gorham, E, Lehman, C, Dyke, A, Janssens, J, Dyke, L (2007) Temporal and spatial aspects of peatland initiation following deglaciation in North America. *Quaternary Science Reviews* 26, 300-311.
- Graham, M (2003) Confronting multicollinearity in ecological multiple regression. *Ecology* 84, 2809-2915.
- Greene, DF, Johnson, EA (1999) Modelling recruitment of Populus tremuloides, Pinus banksiana, and Picea mariana following fire in the mixedwood boreal forest. *Canadian Journal of Forest Research* **29**, 462-473.
- Greene, DF, Zasada, JC, Sirois, L, Kneeshaw, D, Morin, H, Charron, I, Simard, MJ (1999) A review of the regeneration dynamics of North American boreal forest tree species. *Canadian Journal of Forest Research* **29**, 824-839.
- Grimm, EC (1987) Coniss a Fortran-77 Program for Stratigraphically Constrained Cluster-Analysis by the Method of Incremental Sum of Squares. *Computers & Geosciences* 13, 13-35.
- Group, EW (1989) Ecoclimatic regions of Canada, first approximation. Ecological Land Classification Series, No.23. ed. Sustainable Development Branch, Canadian Wildlife Service, Conservation and Protection, Environment Canada, Ottawa, Ont.
- Group, NWW (1997) The Canadian Wetland Classification System, 2nd edition. University of Waterloo, Waterloo, Ontario.
- Hallett, D, Walker, R (2000) Paleoecology and its application to fire and vegetation management in Kootenay National Park, British Columbia. *Journal of Paleolimnology* **24**, 401-414.
- Halsey, LA, Vitt, DH, Gignac, LD (2000) Sphagnum-Dominated Peatlands in North America since the Last Glacial Maximum: Their Occurrence and Extent. *The Bryologist* **103**, 334-352.

- Harden, JW, Trumbore, SE, Stocks, BJ, Hirsch, A, Gower, ST, O'Neill, KP, Kasischke, ES (2000) The role of fire in the boreal carbon budget. *Global Change Biology* 6, 174-184.
- Harris, AG, McMurray, SC, Uhlig, PWC, Jeglum, JK, Foster, RF, Racey, GD (1996) 'Field guide to the wetland ecosystem classification for northwestern Ontario.' (Ontario Ministry of Natural Resources.: Ontario)
- Hart, SA, Chen, HYH (2006) Understory vegetation dynamics of North American boreal forests. *Critical Reviews in Plant Sciences* **25**, 381-397.
- Heck, KL, van Belle, G, Simberloff, D (1975) Explicit Calculation of the Rarefaction Diversity Measurement and the Determination of Sufficient Sample Size. *Ecology* **56**, 1459-1461.
- Heinselman, ML (1981) Fire and succession in the conifer forests of northern North America. In 'Forest Succession: Concepts and Applications.' pp. 374-405. (Springer-Verlag: New York)
- Hellberg, E, Niklasson, M, Granstrom, A (2004) Influence of landscape structure on patterns of forest fires in boreal forest landscapes in Sweden. *Canadian Journal of Forest Research* 34, 332-338.
- Hely, C, Bergeron, Y, Flannigan, MD (2000) Effects of stand composition on fire hazard in mixed-wood Canadian boreal forest. *Journal of Vegetation Science* **11**, 813-824.
- Hely, C, Girardin, MP, Ali, AA, Carcaillet, C, Brewer, S, Bergeron, Y (2010) Eastern boreal North American wildfire risk of the past 7000 years: A model-data comparison. *Geophysical Research Letters* **37**,
- Hessl, AE (2011) Pathways for climate change effects on fire: Models, data, and uncertainties. *Progress in Physical Geography* **35**, 393-407.
- Heyerdahl, EK, McKenzie, D, Daniels, LD, Hessl, AE, Littell, JS, Mantua, NJ (2008) Climate drivers of regionally synchronous fires in the inland Northwest (1651-1900). *International Journal of Wildland Fire* 17, 40-49.
- Higuera, P (2009) 'CharAnalysis: diagnostic and analytical tools for sediment-charcoal analysis.'
- Higuera, PE, Brubaker, LB, Anderson, PM, Brown, TA, Kennedy, AT, Hu, FS (2008) Frequent Fires in Ancient Shrub Tundra: Implications of Paleorecords for Arctic Environmental Change. *Plos One* **3**,
- Higuera, PE, Brubaker, LB, Anderson, PM, Hu, FS, Brown, TA (2009) Vegetation mediated the impacts of postglacial climate change on fire regimes in the south-central Brooks Range, Alaska. *Ecological Monographs* **79**, 201-219.
- Higuera, PE, Gavin, DG, Bartlein, PJ, Hallett, DJ (2010) Peak detection in sediment-charcoal records: impacts of alternative data analysis methods on fire-history interpretations. *International Journal of Wildland Fire* **19**, 996-1014.
- Higuera, PE, Peters, ME, Brubaker, LB, Gavin, DG (2007) Understanding the origin and analysis of sediment-charcoal records with a simulation model. *Quaternary Science Reviews* **26**, 1790-1809.
- Johnson, EA (1992) 'Fire and vegetation dynamics studies from the North American boreal forest.' (Cambridge University Press: Cambridge, UK)
- Johnson, EA, Gutsell, SL (1994) Fire Frequency Models, Methods and Interpretations. In 'Advances in Ecological Research, Vol 25.' pp. 239-287.
- Johnson, EA, Miyanishi, K, O'Brien, N (1999) Long-term reconstruction of the fire season in the mixedwood boreal forest of Western Canada. *Canadian Journal of Botany-Revue Canadienne De Botanique* 77, 1185-1188.

- Johnson, EA, Miyanishi, K, Weir, JMH (1998) Wildfires in the western Canadian boreal forest: Landscape patterns and ecosystem management. *Journal of Vegetation Science* 9, 603-610.
- Johnson, EA, Van Wagner, CE (1985) The Theory and Use of 2 Fire History Models. *Canadian* Journal of Forest Research 15, 214-220.
- Johnston, WF (1990) Larix laricina (Du Roi) K. Koch Tamarack. In 'Silvics of North America. Vol 1. Conifers.' (U.S.D.A. Forest Service: 26-35: Washington, D.C.)
- Johnstone, JF, Chapin, FS (2006) Fire interval effects on successional trajectory in boreal forests of northwest Canada. *Ecosystems* 9, 268-277.
- Johnstone, JF, Chapin, FS, Hollingsworth, TN, Mack, MC, Romanovsky, V, Turetsky, M (2010) Fire, climate change, and forest resilience in interior Alaska. *Canadian Journal of Forest Research* **40**, 1302-1312.
- Johnstone, JF, Rupp, TS, Olson, M, Verbyla, D (2011) Modeling impacts of fire severity on successional trajectories and future fire behavior in Alaskan boreal forests. *Landscape Ecology* **26**, 487-500.
- Juggins, S (2014) 'rioja: Analysis of Quaternary Science Data, R Package version (0.8-7). (<u>http://cran.r-project.org/package=rioja</u>).'
- Kashian, DM, Romme, WH, Tinker, DB, Turner, MG, Ryan, MG (2006) Carbon Storage on Landscapes with Stand-replacing Fires. *Bioscience* 56, 598-606.
- Kasischke, ES, Turetsky, MR (2006) Recent changes in the fire regime across the North American boreal region - Spatial and temporal patterns of burning across Canada and Alaska. *Geophysical Research Letters* **33**,
- Keeley, JE (2009) Fire intensity, fire severity and burn severity: a brief review and suggested usage. *International Journal of Wildland Fire* **18**, 116-126.
- Kelly, R, Chipman, ML, Higuera, PE, Stefanova, I, Brubaker, LB, Hu, FS (2013) Recent burning of boreal forests exceeds fire regime limits of the past 10,000 years. *Proceedings of the National Academy of Sciences* **110**, 13055-13060.
- Kelly, RF, Higuera, PE, Barrett, CM, Hu, FS (2011) A signal-to-noise index to quantify the potential for peak detection in sediment-charcoal records. *Quaternary Research* 75, 11-17.
- Kerby, J, Fuhlendorf, S, Engle, D (2007) Landscape heterogeneity and fire behavior: scaledependent feedback between fire and grazing processes. *Landscape Ecology* 22, 507-516.
- Kerwin, MW, Overpeck, JT, Webb, RS, Anderson, KH (2004) Pollen-based summer temperature reconstructions for the eastern Canadian boreal forest, subarctic, and Arctic. *Quaternary Science Reviews* 23, 1901-1924.
- Krawchuk, MA, Cumming, SG, Flannigan, MD, Wein, RW (2006) Biotic and abiotic regulation of lightning fire initiation in the mixedwood boreal forest. *Ecology* **87**, 458-468.
- Krawchuk, MA, Moritz, MA, Parisien, MA, Van Dorn, J, Hayhoe, K (2009) Global Pyrogeography: the Current and Future Distribution of Wildfire. *Plos One* **4**,
- Larsen, CPS (1997) Spatial and temporal variations in boreal forest fire frequency in northern Alberta. *Journal of Biogeography* 24, 663-673.
- Last, WM, Smol, JP (Eds) (2001) 'Tracking environmental change using lake sediments. Volume 1: Basin analysis, coring, and chronological techniques.' (Kluwer Academic Publishers.: Dordrecht.)

- Lecomte, N, Simard, M, Fenton, N, Bergeron, Y (2006) Fire severity and long-term ecosystem biomass dynamics in coniferous boreal forests of eastern Canada. *Ecosystems* 9, 1215-1230.
- Legendre, P, Legendre, L (1998) 'Numerical Ecology.' (Elsevier Science:
- Littell, JS, McKenzie, D, Peterson, DL, Westerling, AL (2009) Climate and wildfire area burned in western U. S. ecoprovinces, 1916-2003. *Ecological Applications* **19**, 1003-1021.
- Liu, KB (1990) Holocene Paleoecology of the Boreal Forest and Great-Lakes-St-Lawrence Forest in Northern Ontario. *Ecological Monographs* **60**, 179-212.
- Long, CJ, Whitlock, C, Bartlein, PJ, Millspaugh, SH (1998) A 9000-year fire history from the Oregon Coast Range, based on a high-resolution charcoal study. *Canadian Journal of Forest Research* 28, 774-787.
- Lynch, JA, Clark, JS, Bigelow, NH, Edwards, ME, Finney, BP (2002) Geographic and temporal variations in fire history in boreal ecosystems of Alaska. *Journal of Geophysical Research-Atmospheres* 108,
- Lynch, JA, Clark, JS, Stocks, BJ (2004a) Charcoal production, dispersal, and deposition from the Fort Providence experimental fire: interpreting fire regimes from charcoal records in boreal forests. *Canadian Journal of Forest Research* **34**, 1642-1656.
- Lynch, JA, Hollis, JL, Hu, FS (2004b) Climatic and landscape controls of the boreal forest fire regime: Holocene records from Alaska. *Journal of Ecology* **92**, 477-489.
- MacInnis, GM (2012) Measuring and Modelling the Dispersal of Pollen and Spores by Wind. Concordia University.
- Mackey, BG, McKenney, DW, Yang, YQ, McMahon, JP, Hutchinson, MF (1996) Site regions revisited: A climatic analysis of Hills' site regions for the province of Ontario using a parametric method. *Canadian Journal of Forest Research-Revue Canadienne de Recherche Forestiere* **26**, 333-354.
- Malanson, GP (Ed. L Trabaud (1987) 'Diversity, stability and resilience: effects of fire regime.' (SPB Academic Publishing by:
- Mallik, AU, Bell, FW, Gong, Y (1997) Regeneration behavior of competing plants after clear cutting: implications for vegetation management. *Forest Ecology and Management* **95**, 1-10.
- Mansuy, N, Gauthier, S, Robitaille, A, Bergeron, Y (2010) The effects of surficial depositdrainage combinations on spatial variations of fire cycles in the boreal forest of eastern Canada. *International Journal of Wildland Fire* **19**, 1083-1098.
- Marlon, J, Bartlein, PJ, Whitlock, C (2006) Fire-fuel-climate linkages in the northwestern USA during the Holocene. *Holocene* 16, 1059-1071.
- Marlon, JR, Bartlein, PJ, Carcaillet, C, Gavin, DG, Harrison, SP, Higuera, PE, Joos, F, Power, MJ, Prentice, IC (2008) Climate and human influences on global biomass burning over the past two millennia. *Nature Geoscience* **1**, 697-702.
- Marlon, JR, Bartlein, PJ, Walsh, MK, Harrison, SP, Brown, KJ, Edwards, ME, Higuera, PE, Power, MJ, Anderson, RS, Briles, C, Brunelle, A, Carcaillet, C, Daniels, M, Hu, FS, Lavoie, M, Long, C, Minckley, T, Richard, PJH, Scott, AC, Shafer, DS, Tinner, W, Umbanhowar, CE, Whitlock, C (2009) Wildfire responses to abrupt climate change in North America. Proceedings of the National Academy of Sciences of the United States of America 106, 2519-2524.

MathWorks (2005) 'Matlab software.' (MathWorks: Natick, Massachusetts, USA)

- McAndrews, JH, Berti, AA, Norris, G (1973) 'Key to the Quaternary pollen and spores of the Great Lakes region.' (Royal Ontario Museum: Toronto)
- Miller, C, Urban, DL (1999) Forest pattern, fire, and climatic change in the Sierra Nevada. *Ecosystems* **2**, 76-87.
- Miller, C, Urban, DL (2000) Connectivity of forest fuels and surface fire regimes. *Landscape Ecology* **15**, 145-154.
- Millspaugh, SH, Whitlock, C, Bartlein, PJ (2000) Variations in fire frequency and climate over the past 17 000 yr in central Yellowstone National Park. *Geology* 28, 211-214.
- Moore, P, Webb, JA, Collinson, ME (1991) 'Pollen Analysis.' (Blackwell Scientific Publications: Oxford, England)
- Nakagawa, S, Schielzeth, H (2013) A general and simple method for obtaining R2 from generalized linear mixed-effects models. *Methods in Ecology and Evolution* **4**, 133-142.
- Nash, CH, Johnson, EA (1996) Synoptic climatology of lightning-caused forest fires in subalpine and boreal forests. *Canadian Journal of Forest Research* **26**, 1859-1874.
- Natural Resources Canada, EaSSCfti, 2007. Canadian Digital Elevation Data. Sherbrooke, Quebec, Canada.
- Natural Resources Canada, EaSSCfti, 2013. CanVec. Sherbrooke, Quebec, Canada.
- Oksanen, J, Blanchet, FG, Kindt, R, Legendre, P, O'Hara, G, Simpson, GL, Solymos, P, Stevens, MHH, Wagner, H (2010) 'vegan: Community Ecology Package.'
- Ontario Geological Survey, MoNDaM, and Northeast Science and Information Section, Ministry of Natural Resources (MNDM), 2005. Digital Northern Ontario Engineering Geology Terrain Study (NOEGTS). Miscellaneous Releast.
- Oris, F, Asselin, H, Ali, AA, Finsinger, W, Bergeron, Y (2013) Effect of increased fire activity on global warming in the boreal forest. *Environmental Reviews* 1-14.
- Paine, RT, Tegner, MJ, Johnson, EA (1998) Compounded perturbations yield ecological surprises. *Ecosystems* 1, 535-545.
- Parisien, MA, Parks, SA, Krawchuk, MA, Flannigan, MD, Bowman, LM, Moritz, MA (2011) Scale-dependent controls on the area burned in the boreal forest of Canada, 1980-2005. *Ecological Applications* 21, 789-805.
- Payette, S (1992) Fire as a controlling process in the North American boreal forest. In 'A Systems Analysis of the Global Boreal Forest.' (Eds HH Shugart, R Leemans, GB Bonan.) pp. 144-169. (Cambridge University Press: Cambridge)
- Payette, S (1993) The range limit of boreal tree species in Québec-Labrador: an ecological and palaeoecological interpretation. *Review of Palaeobotany and Palynology* **79**, 7-30.
- Pechony, O, Shindell, DT (2010) Driving forces of global wildfires over the past millennium and the forthcoming century. *Proceedings of the National Academy of Sciences of the United States of America* **107**, 19167-19170.
- Perala, DA (1990) Populus tremuloides Michx Quaking aspen. In 'Silvics of North America. Vol 2. Hardwoods.' pp. 555-569. (U.S.D.A. Forest Service: Washington D.C.)
- Peters, DPC, Pielke, RA, Bestelmeyer, BT, Allen, CD, Munson-McGee, S, Havstad, KM (2004) Cross-scale interactions, nonlinearities, and forecasting catastrophic events. *Proceedings* of the National Academy of Sciences of the United States of America **101**, 15130-15135.
- Peters, ME, Higuera, PE (2007) Quantifying the source area of macroscopic charcoal with a particle dispersal model. *Quaternary Research* 67, 304-310.
- Pickett, STA, Collins, SL, Armesto, JJ (1987) A hierarchical consideration of causes and mechanisms of succession. *Vegetatio* **69**, 109 114.

- Power, MJ, Marlon, J, Ortiz, N, Bartlein, PJ, Harrison, SP, Mayle, FE, Ballouche, A, Bradshaw, RHW, Carcaillet, C, Cordova, C, Mooney, S, Moreno, PI, Prentice, IC, Thonicke, K, Tinner, W, Whitlock, C, Zhang, Y, Zhao, Y, Ali, AA, Anderson, RS, Beer, R, Behling, H, Briles, C, Brown, KJ, Brunelle, A, Bush, M, Camill, P, Chu, GQ, Clark, J, Colombaroli, D, Connor, S, Daniau, AL, Daniels, M, Dodson, J, Doughty, E, Edwards, ME, Finsinger, W, Foster, D, Frechette, J, Gaillard, MJ, Gavin, DG, Gobet, E, Haberle, S, Hallett, DJ, Higuera, P, Hope, G, Horn, S, Inoue, J, Kaltenrieder, P, Kennedy, L, Kong, ZC, Larsen, C, Long, CJ, Lynch, J, Lynch, EA, McGlone, M, Meeks, S, Mensing, S, Meyer, G, Minckley, T, Mohr, J, Nelson, DM, New, J, Newnham, R, Noti, R, Oswald, W, Pierce, J, Richard, PJH, Rowe, C, Goni, MFS, Shuman, BN, Takahara, H, Toney, J, Turney, C, Urrego-Sanchez, DH, Umbanhowar, C, Vandergoes, M, Vanniere, B, Vescovi, E, Walsh, M, Wang, X, Williams, N, Wilmshurst, J, Zhang, JH (2008) Changes in fire regimes since the Last Glacial Maximum: an assessment based on a global synthesis and analysis of charcoal data. *Climate Dynamics* 30, 887-907.
- Power, MJ, Marlon, JR, Bartlein, PJ, Harrison, SP (2010) Fire history and the Global Charcoal Database: A new tool for hypothesis testing and data exploration. *Palaeogeography Palaeoclimatology Palaeoecology* **291**, 52-59.
- Reimer, PB, MGL; Bard, E; Bayliss, A; Beck, JW; Blackwell, PG; Bronk Ramsey, C; Buck, CE; Burr, GS; Edwards, RL; Friedrich, M; Grootes, PM; Guilderson, TP; Hajdas, I; Heaton, TJ; Hogg, AG; Hughen, KA; Kaiser, KF; Kromer, B; McCormac, FG; Manning, SW; Reimer, RW; Richards, DA; Southon, JR; Talamo, S; Turney, CSM; van der Plicht, J; Weyhenmeyer, CE (2009) IntCal09 and Marine09 Radiocarbon Age Calibration Curves, 0-50,000 Years cal BP. *Radiocarbon* 50, 1111-1150.
- Reimer, PJ, Baillie, MGL, Bard, E (2004) INTCAL04 terrestrial radiocarbon age calibration, 0-26 cal kyr BP. *Radiocarbon* 46, 1029-1058.
- Richard, PJH (1970) L'Atlas pollinique des arbres et de quelques arbustes indigènes du Québec. Le Naturaliste Canadien 97, 1-34, 97-161, 241-306.
- Richard, PJH (1976) Relations entre la végétation actuelle et le spectre pollinique au Québec. *Naturaliste Canadien* **103**, 53-66.
- Ripley, BD (1977) Modeling Spatial Patterns. Journal of the Royal Statistical Society Series B-Methodological **39**, 172-212.
- Ritchie, JC, Yarranton, GA (1978) Late-Quaternary history of boreal forests of Central canada, based on standard pollen stratigraphy and Principal Components Analysis. *Journal of Ecology* **66**, 199-212.
- Romme, WH (1982) Fire and Landscape Diversity in Subalpine Forests of Yellowstone-National-Park. *Ecological Monographs* 52, 199-221.
- Rowe, JS (1972) Forest Regions of Canada. Canadian Forestry Service, Ottawa.
- Rowe, JS, Scotter, GW (1973) Fire in the Boreal Forest. Quat. Res. 3, 444-464.
- Rudolph, TD, Laidly, PR (1990) *Pinus banksiana* Lamb. Jack Pine. In 'Silvics of North America. Vol 1. Conifers.' pp. 280-293. (U.S.D.A. Forest Service: Washington D.C.)
- Ryan, KC (2002) Dynamic interactions between forest structure and fire behavior in boreal ecosystems. *Silva Fennica* **36**, 13-39.
- Rydin, H, Jeglum, H (2006) 'The Biology of Peatlands.' (Oxford University Press: United States)
- Senici, D, Chen, HYH, Bergeron, Y, Cyr, D (2010) Spatiotemporal Variations of Fire Frequency in Central Boreal Forest. *Ecosystems* 13, 1227-1238.

- Senici, D, Lucas, A, Chen, HYH, Bergeron, Y, Larouche, A, Brossier, B, Blarquez, O, Ali, AA (2013) Multi-millennial fire frequency and tree abundance differ between xeric and mesic boreal forests in central Canada. *Journal of Ecology* 101, 356-367.
- Shuman, B, Bartlein, PJ, Webb, T (2005) The magnitudes of millennial- and orbital-scale climatic change in eastern North America during the Late Quaternary. *Quaternary Science Reviews* **24**, 2194-2206.
- Skinner, WR, Flannigan, MD, Stocks, BJ, Martell, DL, Wotton, BW, Todd, JB, Mason, JA, Logan, KA, Bosch, EM (2002) A 500-hPa synoptic wildland fire climatology for large Canadian forest fires 1959-1996. *Theoretical and Applied Climatology* 71, 157-169.
- Skinner, WR, Stocks, BJ, Martell, DL, Bonsal, B, Shabbar, A (1999) The association between circulation anomalies in the mid-troposphere and the area burned by wildland fire in Canada. *Theoretical and Applied Climatology* **63**, 89-105.
- Stauffer, H (2008) 'Contemporary Bayesian and frequentist statistical research methods for natural resource scientists.' (Wiley and Sons: Hoboken, New Jersey)
- Stocks, BJ, Mason, JA, Todd, JB, Bosch, EM, Wotton, BM, Amiro, BD, Flannigan, MD, Hirsch, KG, Logan, KA, Martell, DL, Skinner, WR (2002) Large forest fires in Canada, 1959-1997. Journal of Geophysical Research-Atmospheres 108,
- Stocks, BJ, Wotton, BM, Flannigan, M, Fosberg, MA, Cahoon, DR, Goldammer, JG (2001) Boreal Forest Fire Regimes and Climate Change. In 'Remote Sensing and Climate Modeling: Synergies and Limitations.' pp. 233-246. (Springer Netherlands:
- Stuiver, M, Reimer, PJ (1993) Extended C-14 Data-Base and Revised Calib 3.0 C-14 Age Calibration Program. *Radiocarbon* 35, 215-230.
- Stuiver, M, Reimer, PJ (2011) 'CALIB 6.1.1.'
- Sugita, S (1993) A model of pollen source area for an entire lake surface. *Quaternary Research* **39**, 239-244.
- Taylor, AR, Chen, HYH (2011) Multiple successional pathways of boreal forest stands in central Canada. *Ecography* **34**, 208-219.
- Taylor, AR, Chen, HYH, VanDamme, L (2009) A Review of Forest Succession Models and Their Suitability for Forest Management Planning. *Forest Science* **55**, 23-36.
- Team, RDC (2013) 'R: A Language and Environment for Statistical Computing.' Vienna, Austria)
- Telford, RJ, Heegaard, E, Birks, HJB (2004) The intercept is a poor estimate of a calibrated radiocarbon age. *Holocene* 14, 296-298.
- Thornley, JH, Cannell, MG (2004) Long-term effects of fire frequency on carbon storage and productivity of boreal forests: a modeling study. *Tree Physiol* **24**, 765-73.
- Tilman, D (1982) 'Resource competition and community structure.' (Princeton Univ. Press: Princeton, NJ)
- Tinner, W, Ammann, B, Germann, P (1996) Treeline fluctuations recorded for 12,500 years by soil profiles, pollen, and plant macrofossils in the Central Swiss Alps. *Arctic and Alpine Research* 28, 131-147.
- Titus, JH, Bishop, JG (2014) Propagule limitation and competition with nitrogen fixers limit conifer colonization during primary succession. *Journal of Vegetation Science* **25**, 990-1003.
- Turetsky, MR, Amiro, BD, Bosch, E, Bhatti, JS (2004) Historical burn area in western Canadian peatlands and its relationship to fire weather indices. *Global Biogeochemical Cycles* 18,

- Turner, MG (2010) Disturbance and landscape dynamics in a changing world. *Ecology* **91**, 2833-2849.
- Turner, MG, Romme, WH (1994) Landscape dynamics in crown fire ecosystems. Landscape Ecology 9, 59-77.
- Urrego, D, Bush, M, Silman, M, Correa-Metrio, A, Ledru, M-P, Mayle, F, Paduano, G, Valencia, B (2009) Millennial-Scale Ecological Changes in Tropical South America Since the Last Glacial Maximum. In 'Past Climate Variability in South America and Surrounding Regions.' (Eds F Vimeux, F Sylvestre, M Khodri.) Vol. 14 pp. 283-300. (Springer Netherlands:
- Van Wagner, CE (1987) 'The development and structure of the Canadian forest fire weather index system.' (Canadian Forest Service, For Tech Rep 35: Ottawa, Ontario)
- Viau, AE, Gajewski, K (2009) Reconstructing Millennial-Scale, Regional Paleoclimates of Boreal Canada during the Holocene. *Journal of Climate* 22, 316-330.
- Viau, AE, Gajewski, K, Sawada, MC, Fines, P (2006a) Millennial-scale temperature variations in North America during the Holocene. *Journal of Geophysical Research-Atmospheres* 111,
- Viau, AE, Gajewski, K, Sawada, MC, Fines, P (2006b) Millennial-scale temperature variations in North America during the Holocene. *Journal of Geophysical Research-Atmospheres* 111, D09102.
- Vonesh, EF, Chinchilli, VP, Pu, KW (1996) Goodness-of-fit in generalized nonlinear mixedeffects models. *Biometrics* **52**, 572-587.
- Wallenius, TH, Kuuluvainen, T, Vanha-Majamaa, I (2004) Fire history in relation to site type and vegetation in Vienansalo wilderness in eastern Fennoscandia, Russia. *Canadian Journal of Forest Research* 34, 1400-1409.
- Weber, MG, Flannigan, M (1997) Canadian boreal forst ecosystem structure and function in a changing clime: impact on fire regimes. *Environmental Review* **5**, 145-166.
- Weir, JMH, Johnson, EA, Miyanishi, K (2000) Fire frequency and the spatial age mosaic of the mixed-wood boreal forest in western Canada. *Ecological Applications* **10**, 1162-1177.
- Wendel, GW, Smith, HC (1990) Silvics of North America: 1. Conifers. In 'Pinus strobus L. -Eastern White Pine.' (Eds RM Burns, BH Honkala.) pp. 476-488. (U. S. Department of Agriculture, Forest Service: Washington, DC, USA)
- Westerling, AL, Turner, MG, Smithwick, EAH, Romme, WH, Ryan, MG (2011) Continued warming could transform Greater Yellowstone fire regimes by mid-21st century. *Proceedings of the National Academy of Sciences* **108**, 13165-13170.
- Whelan, RJ (1995) 'The Ecology of fire.' (Cambridge University Press: New York)
- Whitlock, C, Anderson, RS (2003) Fire History Reconstructions Based on Sediment Records from Lakes and Wetlands. In 'Fire and Climatic Change in Temperate Ecosystems of the Western Americas.' (Eds TT Veblen, WL Baker, G Montenegro, TW Swetnam.) pp. 3-31. (Springer: New York)
- Whitlock, C, Larsen, C (2001) Charcoal as a fire proxy. In 'Tracking environmental change using lake sediments. Vol. 3. Terrestrial, algal, and siliceous indicators.' (Eds JP Smol, HJ Birks, WM Last.) pp. 75-97. (Kluwer Academic Publishers: Dordrecht)
- Whitlock, C, Millspaugh, SH (1996) Testing the assumptions of fire history studies: An examination of modern charcoal accumulation in Yellowstone National Park, USA. *Holocene* 6, 7-15.

- Wotton, BM, Martell, DL (2005) A lightning fire occurrence model for Ontario. *Canadian* Journal of Forest Research **35**, 1389-1401.
- Wotton, BM, Nock, CA, Flannigan, MD (2010) Forest fire occurrence and climate change in Canada. *International Journal of Wildland Fire* **19**, 253-271.
- Wright, HE, Mann, DH, Glaser, PH (1984) Piston corers for peat and lake sediments. *Ecology* **65**, 657-659.

Appendices

SI 2.1: Figure S1

Lake Ben - CHAR raw series



Lake Small - CHAR raw series



SI 2.2: Figure S2





Lake Small - Macroremains raw counts



SI 3.1: Table S1

Table S1: Accelerator mass spectrometry 14C dating of the study lakes (mean \pm 1 SEM)

Site and depth	¹⁴ C year BP	Calibration range	Materials dated	Lak Ord
(cm)		(cal. years BP; 2o)	Materials dated	Lab. Code
AVR	<u> </u>			
415-420	615 ± 30	601 (549-656)	Plant macroremains	Poz-48149
450-455	905 ± 30	833 (744-911)	Plant macroremains	Poz-48150
520-525	1815 ± 30	1757 (1640-1825)	Plant macroremains	Poz-48151
595-600	2800 ± 35	2902 (2803-2994)	Plant macroremains	Poz-48152
670-675	4400 ± 35	4953 (4851 - 5253)	Plant macroremains	Poz-48169
757-758	6500 ± 40	7418 (7323-7486)	Bulk sediment	Poz-48147
859-860	8430 ± 50	9461 (9317-9524)	Bulk sediment	Poz-48148
874-880	8860 ± 50	9989 (9734-10161)	Plant macroremains	Poz-48155
BEN				
30-35	320 ± 30	387 (309-473)	Plant macroremains	Poz-39543
58-60	605 ± 30	602 (545-653)	Plant macroremains	Poz-39542
120-122	1670 ± 30	1573 (1512-1685)	Plant macroremains	Beta-29360
184-185	2720 ± 30	2814 (2764-2874)	Plant macroremains	Beta-29360
259-260	3780 ± 40	4155 (4008-4302)	Plant macroremains	Beta-29360
359-360	3930 ± 40	4364 (4247-4504)	Plant macroremains	Beta-29359
459-460	4940 ± 40	5663 (5603-5831)	Plant macroremains	Beta-29359
606-610	7310 ± 40	8107 (8025-8190)	Plant macroremains and charcoal	Poz-39541
631-635	8300 ± 50	9316 (9134-9440)	Plant macroremains and charcoal	Beta-29359
651 - 661	8990 ± 50	10174 (9934-10232)	Plant macroremains	Poz-39891
DOM				
576-581	180 ± 30	179 (2-287)	Plant macroremains	Poz-48166
616-621	1255 ± 30	1210 (1088-1271)	Plant macroremains	Poz-48167
686-691	2065 ± 30	2033 (1950-2118)	Plant macroremains	Poz-48168
756-761	3155 ± 30	3382 (3305-3443)	Plant macroremains	Poz-48169
826-831	4630 ± 35	5407 (5302-5462)	Plant macroremains	Poz-48170
896-899	5880 ± 40	6702 (6585-6788)	Plant macroremains	Poz-48171
963-971	7590 ± 50	8396 (8314-8508)	Plant macroremains	Poz-48172
998-1007	8860 ± 50	9989 (9734-10161)	Plant macroremains	Poz-48173
BVR		· · · · · ·		
306-307	575 ± 30	602 (534-644)	Bulk sediment	Poz-48156
335-336	1090 ± 35	977 (821-1131)	Bulk sediment	Poz-48300
387-388	2710 ± 35	2809 (2759-2876)	Bulk sediment	Poz-48158
440-441	3300 ± 35	3524 (3451-3623)	Bulk sediment	Poz-48159
500-501	3870 + 35	4306 (4166-4409)	Bulk sediment	Poz-48161

560-561	6490 ± 50	7391 (7295-7493)	Bulk sediment	Poz-48162
629-630	8480 ± 50	9495 (9416-9536)	Bulk sediment	Poz-48163
647-648	9850 ± 50	11251 (11200-11384)	Bulk sediment	Poz-48165
DUB				
410-420	395 ± 30	467 (331-506)	Plant macroremains	Poz-48175
455-460	1640 ± 30	1538 (1424-1618)	Plant macroremains	Poz-48176
525-530	3085 ± 35	3306 (3217-3375)	Plant macroremains	Poz-48177
595-600	5120 ± 40	5843 (5753-5947)	Plant macroremains	Poz-48178
675-674	6860 ± 40	7689 (7617-7786)	Bulk sediment	Poz-48179
739-740	7830 ± 50	8613 (8485-8865)	Bulk sediment	Poz-48180
810-820	8820 ± 50	9874 (9681-10141)	Plant macroremains	Poz-48181
830-840	8960 ± 50	10085 (9919-10221)	Plant macroremains	Poz-48182
SML				
47-50	1190 ± 30	1115 (1014-1221)	Plant macroremains and charcoal	Beta-293608
76-80	2120 ± 30	2093 (2006-2285)	Plant macroremains and charcoal	Beta-293607
145-150	3490 ± 40	3764 (3652-3865)	Bulk sediment	Beta-293606
216-220	3750 ± 40	4110 (3989-4233)	Plant macroremains and charcoal	Beta-293605
287-288	6410 ± 40	7349 (7349-7419)	Plant macroremains	Beta-293609
356-360	7860 ± 40	8644 (8563-8917)	Plant macroremains	Beta-293604
380-390	8850 ± 40	9972 (9972-10150)	Plant macroremains	Beta-293603

115

SI 3.2: Table S2

	Non-fuel	Glaciofluvial	Morainal	Peatlands	Open-water	Wetlands	Dry drainage	Total organic
Non-fuel	1	-0.01	-0.38	0.01	0.94	0.71	0.07	-0.50
Glaciofluvial	-0.01	1	-0.79	-0.13	-0.01	0.00	-0.82	0.68
Morainal	-0.38	-0.79	1	-0.37	-0.31	-0.36	0.84	-0.53
Peatlands	0.01	-0.13	-0.37	1	-0.10	0.24	-0.34	0.28
Open-water	0.94	-0.01	-0.31	-0.10	1	0.44	0.19	-0.61
Wetlands	0.71	0.00	-0.36	0.24	0.44	1	-0.22	-0.06
Dry drainage	0.07	-0.82	0.84	-0.34	0.19	-0.22	1	-0.89
Total organic	-0.50	0.68	-0.53	0.28	-0.61	-0.06	-0.89	1

Table S2: Spearman correlation matrix of environmental variables.

SI 3.3: Table S3

Table S3: Final model results using 20 and 30 year temporal resolution fire reconstructions.

Model		20-ye	ar resolu	ution	30-year resolution		
	Term	Estimate	SE	p-value	Estimate	SE	p-value
Model 1	(Intercept)	5.334	0.012	< 0.001	5.555	0.008	< 0.001
	Time	0.060	0.004	< 0.001	0.068	0.004	< 0.001
	Non-fuel	0.091	0.012	< 0.001	0.087	0.008	< 0.001
	Peatlands	-0.017	0.012	0.162	-0.055	0.008	< 0.001
	Dry drainage	0.016	0.013	0.227	0.017	0.008	0.042
	Time × Non-fuel	0.040	0.004	< 0.001	0.073	0.004	< 0.001
	Time × Peatlands	-0.052	0.004	< 0.001	0.011	0.004	0.015
	Time × Dry drainage	-0.065	0.004	< 0.001	-0.076	0.004	< 0.001
Model 2	(Intercept)	5.332	0.011	< 0.001	5.557	0.022	< 0.001
	Time	0.058	0.004	< 0.001	0.067	0.004	< 0.001
	Open-water	0.049	0.013	< 0.001	0.061	0.027	0.023
	Wetlands	0.058	0.013	< 0.001	0.039	0.027	0.142
	Dry drainage	0.029	0.012	0.018	0.037	0.025	0.142
	Time × Open-water	0.011	0.005	0.032	0.047	0.005	< 0.001
	Time × Wetlands	0.033	0.005	< 0.001	0.038	0.005	< 0.001
	Time × Dry drainage	-0.040	0.005	< 0.001	-0.075	0.004	< 0.001
Model 3	(Intercept)	5.335	0.023	< 0.001	5.558	0.027	< 0.001
	Time	0.060	0.004	< 0.001	0.066	0.004	< 0.001
	Open-water	0.072	0.030	0.016	0.068	0.034	0.047
	Total organic	-0.017	0.030	0.577	-0.030	0.035	0.381
	Time × Open-water	0.059	0.005	< 0.001	0.119	0.005	< 0.001
	Time × Total organic	0.063	0.005	< 0.001	0.105	0.005	< 0.001